UNITED STATES BANKRUPTCY COURT DISTRICT OF MINNESOTA

In re:	Jointly Administered under Case No. 08-46617
Polaroid Corporation, et al.,	Court Files No.'s:
Debtors.	08-46617 (GFK)
(includes:	
Polaroid Holding Company;	08-46621 (GFK)
Polaroid Consumer Electronics, LLC;	08-46620 (GFK)
Polaroid Capital, LLC;	08-46623 (GFK)
Polaroid Latin America I Corporation;	08-46624 (GFK)
Polaroid Asia Pacific LLC;	08-46625 (GFK)
Polaroid International Holding LLC;	08-46626 (GFK)
Polaroid New Bedford Real Estate, LLC;	08-46627 (GFK)
Polaroid Norwood Real Estate, LLC;	08-46628 (GFK)
Polaroid Waltham Real Estate, LLC)	08-46629 (GFK)
	Chapter 11 Cases
	Judge Gregory F. Kishel

REVISED NOTICE OF PREVAILING BIDDER

Attached for filing please find the Asset Purchase Agreement between the Debtors and PLR Holdings, LLC, dated April 16, 2009.

DATED: April 17, 2009 LINDQUIST & VENNUM P.L.L.P.

By: /e/ George H. Singer
James A. Lodoen (173605)
George H. Singer (0262043)
4200 IDS Center
80 South Eighth Street
Minneapolis, MN 55402-2274
(612) 371-3211
(612) 371-3207 (facsimile)
www.lindquist.com

ATTORNEYS FOR THE DEBTORS

ASSET PURCHASE AGREEMENT

by and among

POLAROID HOLDING COMPANY, POLAROID CORPORATION,
POLAROID CONSUMER ELECTRONICS, LLC, POLAROID CAPITAL, LLC,
POLAROID LATIN AMERICA I CORPORATION, POLAROID ASIA PACIFIC, LLC,
POLAROID INTERNATIONAL HOLDING, LLC,
POLAROID NEW BEDFORD REAL ESTATE, LLC, POLAROID NORWOOD REAL ESTATE, LLC,
POLAROID WALTHAM REAL ESTATE, LLC

and

PLR ACQUISITION, LLC

Dated as of April 16, 2009

TABLE OF CONTENTS

		Page
ARTICLE 1 PURCH	ASE AND SALE OF THE ACQUIRED ASSETS	2
SECTION 1.1	Transfer of Acquired Assets	
SECTION 1.2	Excluded Assets	
SECTION 1.3	Assumption of Liabilities	
SECTION 1.4	Excluded Liabilities	6
SECTION 1.5	Identification of Additional and Excluded Contracts;	
	Additional Acquired Equipment	7
SECTION 1.6	Assumption/Rejection of Certain Contracts	
SECTION 1.7	[Intentionally Omitted]	
SECTION 1.8	Prorations	8
ARTICLE 2 PURCH	ASE PRICE	8
SECTION 2.1	Purchase Price	
SECTION 2.2	Deposit	
ARTICLE 3 CLOSIN	G AND DELIVERIES	9
SECTION 3.1	Closing	9
SECTION 3.2	Seller's Deliveries	
SECTION 3.3	Buyer's Deliveries	10
ARTICLE 4 REPRES	SENTATIONS AND WARRANTIES	10
SECTION 4.1	Representations and Warranties of Sellers	10
SECTION 4.2	Representations and Warranties of Buyer	
SECTION 4.3	Warranties Are Exclusive	15
ARTICLE 5 COVEN	ANTS AND OTHER AGREEMENTS	15
SECTION 5.1	Pre-Closing Covenants of Sellers	15
SECTION 5.2	Pre-Closing Covenants of Buyer	16
SECTION 5.3	Other Covenants of Sellers and Buyer	17
SECTION 5.4	Ownership and Use of Polaroid Name	18
SECTION 5.5	Bankruptcy Actions	
SECTION 5.6	Employees	20
SECTION 5.7	Employee Benefit Plans and Pre-Closing Employment Liabilities	21
SECTION 5.8	Other Actions	
SECTION 5.9	[Intentionally Omitted]	
SECTION 5.10	Use of Names.	
SECTION 5.11	Certain Affiliate Arrangements.	
ARTICLE 6 TAXES		22
SECTION 6.1	Taxes Related to Purchase of Acquired Assets	
SECTION 6.2	Cooperation on Tax Matters	
SECTION 6.3	Allocation of Purchase Price	

ARTICLE 7 CONDITI	ONS PRECEDENT TO PERFORMANCE BY PARTIES	24
SECTION 7.1	Conditions Precedent to Performance by Sellers	24
SECTION 7.2	Conditions Precedent to the Performance by Buyer	
ARTICLE 8 TERMINA	ATION	26
SECTION 8.1	Conditions of Termination	26
SECTION 8.2	Effect of Termination	28
ARTICLE 9 SURVIVA	AL	28
SECTION 9.1	Survival	28
SECTION 9.2	Specific Performance	28
SECTION 9.3	Covenant Not to Sue	28
ARTICLE 10 MISCEL	LANEOUS	29
SECTION 10.1	Joint Drafting	29
SECTION 10.2	Further Assurances	
SECTION 10.3	Successors and Assigns	
SECTION 10.4	Governing Law; Jurisdiction	
SECTION 10.5	Expenses	
SECTION 10.6	Severability	
SECTION 10.7	Notices	
SECTION 10.8	Amendments; Waivers	31
SECTION 10.9	Public Announcements	31
SECTION 10.10	Entire Agreement	32
SECTION 10.11	No Third Party Beneficiaries	
SECTION 10.12	Headings	
SECTION 10.13	Counterparts	32
SECTION 10.14	Construction	
SECTION 10.15	Tax Disclosure	32
SECTION 10.16	Sellers' Representative	33
ARTICLE 11 DEFINIT	ΓΙΟΝS	33

SCHEDULES

Schedule 1.1(a)	Real Property
Schedule 1.1(b)	Owned Machinery and Equipment Locations
Schedule 1.1(c)	Acquired Contracts
Schedule 1.1(d)	Inventory and Inventory Locations
Schedule 1.1(e)	Supplies Locations
Schedule 1.1(f)	Acquired Intellectual Property
Schedule 1.1(n)	Letters of Credit
Schedule 1.1(o)	Intellectual Property Causes of Action
Schedule 1.2(c)	Excluded Contracts
Schedule 1.2(e)	Miscellaneous Excluded Assets
Schedule 1.2(q)	Artwork, Archival Documents, Images and Scientific Notes
Schedule 1.3(f)	Other Liabilities
Schedule 1.6(b)	Cure Schedule
Schedule 1.6(c)	Executory Contracts
Schedule 4.1(a)	Jurisdictions of Organization
Schedule 4.1(d)	Seller Reports
Schedule 4.1(f)	Subsidiaries
Schedule 4.1(g)	Sellers Consents and Approvals
Schedule 4.1(h)	Litigation
Schedule 4.1(j)	Intellectual Property Matters
Schedule 4.1(1)	Permit Violations
Schedule 4.1(m)	Environmental Matters
Schedule 4.1(r)	Impaired Inventory
Schedule 4.1(s)	Contracts
Schedule 5.1(e)	Subject Internet Domain Names
Schedule 5.6(b)	Offered Employees
Schedule 11(a)	Acquired Subsidiaries
Schedule 11(b)	Definitively Acquired Contracts
Schedule 11(c)	Definitively Excluded Contracts

EXHIBITS

Exhibit A	Limited Liability Company Agreement
Exhibit B	Bankruptcy Bidding Procedures Order
Exhibit C	Bankruptcy Sale Order

Exhibit D Specified Entities

ASSET PURCHASE AGREEMENT

THIS ASSET PURCHASE AGREEMENT (this "Agreement"), dated as of April 16, 2009 (the "Execution Date"), is made by and among POLAROID HOLDING COMPANY, a Delaware corporation ("PHC"), POLAROID CORPORATION, a Delaware corporation ("PC"), POLAROID CONSUMER ELECTRONICS, LLC, a Delaware limited liability company ("PCE"), POLAROID CAPITAL, LLC, a Delaware limited liability company ("PCAP"), POLAROID LATIN AMERICA I CORPORATION, a Delaware corporation ("PLA"), POLAROID ASIA PACIFIC, LLC, a Delaware limited liability company ("PAP"), POLAROID INTERNATIONAL HOLDING, LLC, a Delaware limited liability company ("PINT"), POLAROID NEW BEDFORD REAL ESTATE, LLC, a Delaware limited liability company ("PNB"), POLAROID NORWOOD REAL ESTATE, LLC, a Delaware limited liability company ("PNOR"), and Polaroid Waltham Real Estate, LLC, a Delaware limited liability company ("PNOR"), and Polaroid Waltham Real Estate, LLC, a Delaware limited liability company ("PWALT" and, together with PHC, PC, PCE, PCAP, PLA, PAP, PINT, PNB and PNOR, the "Sellers" and each, individually, a "Seller"), and PLR ACQUISITION, LLC, a Delaware limited liability company ("Buyer"). Capitalized terms used in this Agreement are defined or cross-referenced in Article 11.

RECITALS

WHEREAS, Sellers are debtors in possession under Title 11 of the United States Code, 11 U.S.C. § 101 et seq. (the "<u>Bankruptcy Code</u>"), and filed voluntary petitions for relief under Chapter 11 of the Bankruptcy Code on December 18, 2008, in the United States Bankruptcy Court for the District of Minnesota (the "<u>Bankruptcy Cases</u>");

WHEREAS, the Bankruptcy Cases are currently pending and being jointly administered under Bankruptcy Case No. 08-46617;

WHEREAS, PC is a direct wholly-owned subsidiary of PHC, and each of PCE, PCAP, PLA, PAP, PINT, PNB PNOR and PWALT is a direct wholly-owned subsidiary of PC and, in turn, an indirect wholly-owned subsidiary of PHC;

WHEREAS, there are other entities which are not Sellers and which are direct or indirect subsidiaries of Sellers;

WHEREAS, Sellers, together with their consolidated subsidiaries, design, develop and market instant and digital imaging products and consumer electronics products (the "Business");

WHEREAS, Buyer desires to purchase the Acquired Assets and assume the Assumed Liabilities from Sellers, and Sellers desire to sell, convey, assign and transfer to Buyer the Acquired Assets together with the Assumed Liabilities, all in the manner and subject to the terms and conditions set forth in this Agreement and in accordance with Sections 105, 363 and 365 and other applicable provisions of the Bankruptcy Code; and

WHEREAS, the Acquired Assets and Assumed Liabilities shall be purchased and assumed by Buyer pursuant to the Bankruptcy Sale Order approving such sale, free and clear of all Liens (other than Permitted Liens) and Claims, pursuant to Sections 105, 363 and 365 of the Bankruptcy Code, which order will include the authorization for the assumption by Sellers and assignment to Buyer of the Acquired Contracts and the liabilities thereunder in accordance with Section 365 of the Bankruptcy Code, all in the manner and subject to the terms and conditions set forth in this Agreement and the Bankruptcy Sale Order and in accordance with other applicable provisions of the Bankruptcy Code and the Federal Rules of Bankruptcy Procedure (the "Bankruptcy Rules").

STATEMENT OF AGREEMENT

NOW, THEREFORE, in consideration of the foregoing and their respective representations, warranties, covenants and agreements herein contained, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Sellers and Buyer hereby agree as follows:

ARTICLE 1

PURCHASE AND SALE OF THE ACQUIRED ASSETS

SECTION 1.1 Transfer of Acquired Assets.

At the Closing, and upon the terms and conditions herein set forth, Sellers shall sell to Buyer or its designee, and Buyer or its designee shall acquire from Sellers, all right, title and interest of Sellers in, to and under the Acquired Assets, free and clear of all Liens (other than Permitted Liens) and all Claims. As used herein, the term "Acquired Assets" shall mean all of the properties, assets and rights of Sellers of whatever kind and nature, excluding the Excluded Assets, that are described below:

- (a) the owned real property of Sellers listed on <u>Schedule 1.1(a)</u> (the owned real property of Sellers listed on <u>Schedule 1.1(a)</u> being the "<u>Real Property</u>"), together with all appurtenant, subsurface and mineral rights, licenses, rights-of-way, privileges and easements belonging to, appertaining to or benefiting the Real Property in any way and all Improvements erected thereon;
- (b) all (i) owned equipment, machinery, furniture, fixtures and improvements, tooling and spare parts of Sellers specifically identified by <u>Schedule 1.1(b)</u> (the "<u>Owned Machinery and Equipment</u>") (the locations at which any such Owned Equipment and Machinery are kept are set forth on <u>Schedule 1.1(b)</u>), and (ii) rights of Sellers to the warranties and licenses received from manufacturers and sellers of the Owned Machinery and Equipment;
- (c) those Contracts listed on <u>Schedule 1.1(c)</u> (collectively, the "<u>Acquired Contracts</u>") and all deposits made under any Acquired Contract;
- (d) all (i) Inventory of Sellers, including, without limitation, all (A) Inventory at the locations listed on <u>Schedule 1.1(d)</u> (the "<u>Inventory Locations</u>"), (B) Inventory held by third parties on a consignment basis, (C) Inventory held by third-party processors, and (D) Inventory located on any Real Property, and (ii) rights of Sellers to the warranties received from suppliers with respect to such Inventory;
- (e) all Supplies of Sellers (the locations at which any such Supplies are kept are set forth on <u>Schedule 1.1(e)</u>) (the "<u>Supplies Locations</u>"), including, without limitation, the Supplies located on any Real Property;
- (f) all Intellectual Property owned by Sellers or licensed to Sellers pursuant to an Acquired Contract or otherwise used by Sellers in connection with the operation of the Business, as historically operated by Sellers (collectively, the "<u>Acquired Intellectual Property</u>"), including, without limitation, all rights to the name "Polaroid" (and all rights to any other trade names, trademarks and service marks owned by Sellers; and the Intellectual Property filings listed on <u>Schedule 1.1(f)</u>);
- (g) computer hardware, excluding for all purposes the Excluded Computer-Related Assets;

- (h) to the extent assignable, all permits, authorizations and licenses (collectively, the "<u>Permits</u>") issued to Sellers by any Government Authority and all pending applications therefor;
- (i) copies or originals of all books, files, documents and records owned by or in the control of Sellers and relating to the Acquired Assets (in whatever format they exist, whether in hard copy or electronic format), including, without limitation, customer lists, historical customer files, accounting records, test results, product specifications, plans, data, studies, drawings, diagrams, training manuals, engineering data, safety and environmental reports and documents, maintenance schedules and operating and production records, inventory records, business plans, credit records of customers, and marketing materials;
- (j) all goodwill, payment intangibles and general intangible assets and rights of Sellers;
- (k) any chattel paper owned or held by Sellers for which Sellers are not the account debtor;
- (l) all books, files and records owned by Sellers that relate to Offered Employees who accept employment with Buyer on the Closing Date, including, without limitation, books, files and records that are related to medical history, medical insurance or other medical matters and to workers' compensation and to the evaluation, appraisal or performance of such employees, but only (in each case) to the extent that the sale, assignment and transfer thereof is permitted by applicable law;
- (m) all (i) outstanding shares of capital stock or equity or other ownership interest held by Sellers in the Acquired Subsidiaries (collectively, the "Acquired Equity Interests") and any related or associated investment property and (ii) to the extent in possession of and maintained by Seller, corporate seals, minute books, charter documents, stock transfer records, record books, original Tax and financial records and such other files, books and records relating to each of the Acquired Subsidiaries (excluding those files, books and records relating to any of the Excluded Assets or to the organization, existence or capitalization of Sellers and the Acquired Subsidiaries);
- (n) to the extent transferable and arising under the Acquired Contracts, all of Sellers' interest in any letters of credit issued by any Person at the request or for the benefit of Sellers, including, without limitation, the letters of credit described on <u>Schedule 1.1(n)</u>;
- (o) all of Sellers' Causes of Action to enforce rights in respect of any Acquired Intellectual Property, excluding those listed on <u>Schedule 1.1(o)</u> or any Avoidance Actions (all of Sellers' Causes of Action to enforce rights in respect of any Acquired Intellectual Property, excluding those listed on <u>Schedule 1.1(o)</u> and any Avoidance Actions, being the "<u>Acquired IP Causes of Action</u>");
- (p) all license fees, royalties, commissions and like payments due and owing under the Acquired Contracts accruing after the Closing Date or otherwise attributable or relating to the period of time after the Closing Date;
- (q) all accounts and notes receivable of Sellers, other than the Excluded Receivables (the "Accounts Receivable"); and
- (r) the art, artifacts, archival documents, archival images, memorabilia and other items known as the "Corporate Archives," including without limitation the historical physical artifacts and product samples located in Waltham, Massachusetts and Readville, Massachusetts and other various locations.

For the avoidance of doubt, notwithstanding anything to the contrary in this Agreement, the Acquired Assets shall include all right, title and interest of Sellers in, to and under: (a) any and all Contracts between Sellers or any of them, on the one hand, and Zink Imaging, Inc. (whether in its own name or as successor-in-interest by merger to Zink Imaging, LLC) and/or any of its Affiliates (Zink Imaging, Inc. and its Affiliates, collectively, the "Zink Entities"), on the other hand, including (i) that certain Asset Purchase Agreement dated January 9, 2006, as amended, (ii) that certain Intellectual Property License Agreement dated January 9, 2006, as amended, and (iii) that certain Agreement dated August 29, 2008 granting, among other things, to Sellers certain exclusive rights, (b) any and all equity interests in any Zink Entity owned by any Seller, and (c) any and all Claims and Causes of Action of any Seller against any Zink Entity (other than to the extent such Claims and Causes of Action could be asserted by any Seller as a defense, counterclaim or a right of recoupment or setoff in response to any Cause of Action or Claim brought, commenced, filed or asserted by or on behalf of any Zink Entity against any Seller). Buyer acknowledges that the Acquired Assets shall not include any right, title or interest of PGW or any of its Affiliates (other than Sellers) in equity interests in any Zink Entity owned by PGW or any of its Affiliates (other than Sellers).

SECTION 1.2 Excluded Assets.

Notwithstanding anything to the contrary in this Agreement, Sellers shall retain all right, title and interest to, in and under the properties, rights, interests and assets of Sellers set forth below (all such properties, rights, interests and assets not being acquired by Buyer being herein referred to as the "Excluded Assets"):

- (a) any asset of a Seller that otherwise would constitute an Acquired Asset but for the fact that it is conveyed, leased or otherwise disposed of, in the ordinary course of such Seller's business and consistent with the terms of this Agreement, during the time from the Execution Date until the Closing Date;
- (b) all of Sellers' prepaid expenses, marketable securities and Cash and all of Sellers' right, title and interest in and to all deposit or similar accounts in which Sellers deposit cash or any other assets:
- (c) all Contracts that are not Acquired Contracts, including, without limitation, those Contracts listed on <u>Schedule 1.2(c)</u> (the "<u>Excluded Contracts</u>");
- (d) all Employee Benefit Plans currently or previously sponsored or maintained by Sellers or any of Sellers' ERISA Affiliates (together with Sellers, the "Seller Controlled Group") or their respective predecessors or with respect to which the Seller Controlled Group or their respective predecessors has made or is required to make payments, transfers or contributions in respect of any present or former employees, directors, officers, shareholders, consultants or independent contractors of Sellers or any of Sellers' ERISA Affiliates or their respective predecessors (collectively, the "Seller Benefit Plans"), and all insurance policies, fiduciary liability policies, benefit administration contracts, actuarial contracts, trusts, escrows, surety bonds, letters of credit and other contracts primarily relating to any Seller Benefit Plan;
 - (e) all of the assets set forth on Schedule 1.2(e);
 - (f) all rights to Claims for refunds,
 - (g) overpayment or rebates of Taxes;

- (h) all of Sellers' Causes of Action and any and all proceeds thereof, other than the Acquired IP Causes of Action;
- (i) all of Sellers' rights under any insurance policy or contract of insurance or indemnity (or similar agreement) under which a Seller is an insured, named as an additional insured or is otherwise a beneficiary, and all proceeds realized in connection therewith, including, but not limited to, any and all insurance recovery related to amounts that are owing to Sellers by Circuit City Stores, Inc. or its Affiliates;
 - (j) all amounts due to Sellers from any Affiliate of Sellers;
- (k) all outstanding shares of capital stock or equity or other ownership interest held by a Seller in any other Seller;
- (l) all outstanding shares of capital stock or equity or other ownership interest held by Sellers in any entity that is not an Acquired Subsidiary;
- (m) all Tax records and information ("Tax Records") and all corporate books and records, board minutes, organizational documents of Sellers; <u>provided</u>, <u>however</u>, that copies of corporate books and records, board minutes and organizational documents shall be provided;
- (n) all real property of Sellers, whether owned or leased, that is not listed on Schedule 1.1(a);
 - (o) the Excluded Receivables;
- (p) all license fees, royalties, commissions and like payments due and owing under the Acquired Contracts and accruing on or before the Closing Date or otherwise attributable or relating to the period of time on or before the Closing Date;
- (q) all artwork, archival documents, and archival images owned by Sellers as listed on Schedule 1.2(q);
- (r) any and all information not relating to the Business that is stored on any Seller's computer systems, data network or servers;
- (s) all instant film Inventory held in company code 120 and 122 (the "<u>Subject Inventory</u>"), including rights of Sellers to the warranties received from suppliers with respect to such Inventory;
- (t) all owned equipment, machinery, furniture, fixtures and improvements, tooling and spare parts of Sellers, except as specifically set forth on <u>Schedule 1.1(b)</u>;
 - (u) the Excluded Computer-Related Assets;
- (v) all of Sellers' Causes of Action, other than those acquired by Buyer pursuant to Section 1.1(o); and
 - (w) all Avoidance Actions.
 - SECTION 1.3 Assumption of Liabilities.

At the Closing, Buyer shall assume, and thereafter pay, perform and discharge when due, only the following liabilities (the "Assumed Liabilities"):

- (a) all obligations arising under all Acquired Contracts either (i) arising on or after the Closing Date; or (ii) any cure amount not greater than 110.0% of the total amount set forth in the cure amounts schedule attached as Schedule 1.6(b) hereof;
- (b) all Transaction Taxes that may be imposed by reason of the sale, transfer, assignment and delivery of the Acquired Assets;
- (c) all warranty and return obligations, including without limitation, all liabilities and obligations to repair or replace, or to refund the sales price (or any other related expenses) for Inventory sold by Buyer after the Closing Date;
 - (d) all liabilities in respect of the Acquired Assets arising after the Closing;
- (e) all liabilities relating to amounts required to be paid by Buyer under this Agreement; and
 - (f) the other liabilities set forth on Schedule 1.3(f).

SECTION 1.4 Excluded Liabilities.

Buyer is assuming only the Assumed Liabilities and is not assuming any other liability or obligation of Sellers of whatever nature, whether presently in existence or arising hereafter. All such other liabilities and obligations shall be retained by, and remain liabilities and obligations of, Sellers (all such liabilities are, collectively, the "Excluded Liabilities"). The Excluded Liabilities include, without limitation, the following liabilities and obligations:

- (a) all liabilities and obligations of Sellers relating to Excluded Assets;
- (b) all liabilities and obligations of Sellers or the Seller Controlled Group to all former employees of Sellers (and their respective spouses and dependents);
- (c) all liabilities and obligations of Sellers for: (i) payments made to or fees and expenses accrued with respect to professionals retained or employed by Sellers' Chapter 11 estates or any official committee of creditors appointed in the Sellers' Bankruptcy Cases, (ii) reclamation Claims and (iii) any prepetition, priority or other tax Claims;
- (d) all liabilities in connection with and with respect to the Worker Adjustment and Retraining Notification Act of 1988 or similar state law or any other applicable law and, to the extent required by applicable law, all liabilities and obligations of Sellers or the Seller Controlled Group to all current (as of the Execution Date) employees of Sellers (and their respective spouses and dependents);
- (e) Claims related to products sold and/or produced by or on behalf of Sellers prior to the Closing;
- (f) all liabilities of Sellers relating to the presence or migration of Hazardous Materials on, in or under the Real Property prior to the Closing or the compliance with applicable Environmental Laws prior to the Closing; and

- (g) all warranty and return obligations, including without limitation, all liabilities and obligations to repair or replace, or to refund the sales price (or any other related expenses) for Inventory sold prior to the Closing Date.
- SECTION 1.5 Identification of Additional and Excluded Contracts; Additional Acquired Equipment.
- (a) Buyer shall have the right to designate any Contracts related to the Acquired Assets which do not appear on Schedule 1.1(c) or Schedule 1.2(c) to be assumed and assigned by Sellers to Buyer, and upon designating such Contracts, Schedule 1.1(c) shall be revised. Upon such designation, Sellers shall file a motion seeking the assumption and assignment of such Contracts to Buyer with a proposed cure amount reflecting an amount agreed upon by Sellers and Buyer. In the event an objection to the proposed assumption and assignment is timely received, Buyer shall have the option to remove any such Contracts from revised Schedule 1.1(c). Notwithstanding anything contained herein to the contrary, under no circumstances shall Buyer be permitted to designate any (i) Definitively Excluded Contract as an Acquired Contract following the date hereof or any (ii) Definitively Acquired Contract as an Excluded Contract following the date hereof.
- Following the Closing, Sellers shall provide Buyer notice prior to rejecting any Contract that constitutes an Excluded Contract (other than any Definitively Excluded Contract, with respect to which (i) no such notice shall be required and (ii) Sellers shall not be subject to any restriction hereunder relating to the rejection thereof) as of the Closing (a "Rejection Notice"). Following its receipt of a Rejection Notice, Buyer shall have three (3) Business days to elect to assume the subject Contract(s) (a "Contract Assumption Notice"). Upon such designation, Sellers shall file a motion seeking the assumption and assignment of such Contracts to Buyer with a proposed cure amount reflecting an amount agreed upon by Sellers and Buyer (for which Buyer shall be responsible in accordance with clauses (x) and (y) below of this Section 1.5(b). For the avoidance of doubt, the parties acknowledge and agree that, if Buyer fails to deliver a Contract Assumption Notice within such 3-Business Day period, Sellers shall not be subject to any restriction hereunder relating to the rejection of such Contract(s). In addition, following the Closing, Buyer may require Sellers, upon three (3) Business Days notice, to file a motion seeking, as the case may be, the assumption and assignment of any Excluded Contract (other than a Definitively Excluded Contract) to Buyer, it being understood that, except as may otherwise be agreed to by the parties in writing, to the extent (x) such Excluded Contract is not listed on Schedule 1.6(b) with a proposed cure amount, then Sellers and Buyer shall reasonably and in good faith agree upon a cure amount for such Contract (for which Buyer shall be responsible in its entirety), and (y) such Excluded Contract is listed on Schedule 1.6(b), Buyer shall be responsible for such cure cost as set forth in Section 1.6(b) hereof. Following the Closing, except to the extent otherwise agreed to by Buyer pursuant to the terms hereof, Sellers shall not assume and assign to any entity other than Buyer any Excluded Contract pursuant to which any Acquired Intellectual Property is licensed without Buyer's prior written consent: provided, however, for the avoidance of doubt, the parties acknowledge and agree that Buyer's consent shall not be required to the extent that such Excluded Contracts relate to the Polaroid ID Systems Business or Sellers' 20x24 business.
- (c) Prior to the Closing, Buyer shall be permitted to designate (i) any equipment and machinery owned by any of the Sellers (other than any machinery and equipment specifically identified as an Excluded Asset in Section 1.2 (or in any of the schedules included in such Section)) as an Acquired Asset that will be transferred by Sellers to Buyer at Closing pursuant to the terms hereof, (ii) any Inventory located at the Inventory Locations as an Excluded Asset and (iii) any supplies located at the Supplies Locations as an Excluded Asset. No such designation by Buyer shall result in any increase or decrease to the Purchase Price. For the avoidance of doubt, the parties acknowledge and agree that under no circumstances shall Buyer be permitted to designate the Acquired Equity Interests (or, except as

specifically provided in this <u>Section 1.5</u>, any other Acquired Asset) as an Excluded Asset following the date hereof or (B) any Excluded Asset (except as specifically provided in this <u>Section 1.5</u>) as an Acquired Asset following the date hereof.

SECTION 1.6 Assumption/Rejection of Certain Contracts.

- (a) As of the Closing or the date of any order approving assumption and assignment of an Acquired Contract, as applicable, Sellers shall assume pursuant to Section 365(a) of the Bankruptcy Code and sell and assign to Buyer pursuant to Sections 363(b), (f), and (m) and Section 365(f) of the Bankruptcy Code each of the Acquired Contracts. Buyer shall assume and thereafter in due course pay, fully satisfy, discharge and perform all of the obligations under the Acquired Contracts pursuant to Section 365 of the Bankruptcy Code.
- (b) The cure amounts, as determined by the Bankruptcy Court, if any, necessary to cure all defaults under any Acquired Contract, if any (including, if any, and to the extent allowed by the Bankruptcy Court, all actual or pecuniary losses that have resulted from such defaults under the Acquired Contracts), shall be paid by Buyer, on or before the Closing, and not by Sellers and Sellers shall have no liability therefor; provided, however, that Buyer shall not be obligated to cure any such default that is greater than 110.0% of the total amount set forth in the cure amounts schedule attached as Schedule 1.6(b) hereof.
- (c) Sellers agree to file a motion within ten (10) Business Days after the Closing, seeking rejection of the executory contracts listed on Schedule 1.6(c).

SECTION 1.7 [Intentionally Omitted].

SECTION 1.8 Prorations.

(a) All personal property Taxes or similar ad valorem obligations levied with respect to the Acquired Assets for any taxable period that includes the Closing Date and ends after the Closing Date, whether imposed or assessed before or after the Closing Date, shall be prorated between Sellers, on the one hand, and Buyer, on the other hand, as of 12:01 a.m. (Central Standard Time) on the Closing Date. If any Taxes subject to proration are paid by Buyer, on the one hand, and Sellers, on the other hand, the proportionate amount of such Taxes paid (or in the event of a refund of any portion of such Taxes previously paid is received, such refund) shall be paid promptly by (or to) the other after the payment of such Taxes (or promptly following the receipt of any such refund).

ARTICLE 2

PURCHASE PRICE

SECTION 2.1 Purchase Price.

(a) The aggregate consideration and purchase price (the "<u>Purchase Price</u>") for the sale, transfer, assignment and conveyance of the Acquired Assets will be (i) Fifty-Five Million Dollars (\$55,000,000) (the "<u>Closing Cash Consideration</u>"), *plus* (ii) the Equity Consideration, *plus* (iii) the Receivables Sharing Portion. On the Closing Date, Buyer shall (A) pay the Closing Cash Consideration to Sellers by wire transfer of immediately available funds and (B) issue the Equity Consideration to Sellers, free and clear of any and all Encumbrances. Following the Closing, Buyer shall pay to Sellers on a quarterly basis (commencing on the three-month anniversary of the Closing Date) the Receivables Sharing Portion, if

any, collected by Buyer during the preceding three-month period. The apportionment of the payment of the Purchase Price among Sellers shall be as determined by Sellers and communicated in a written notice (the "Apportionment Notice") by Sellers' Representative to Buyer on or prior to the Closing Date.

- (b) As of the Closing Date, the cost of the Pogo Inventory shall be no less than \$18.6 million. In the event the cost of the Pogo Inventory at the Closing Date is less than \$18.6 million, then the Purchase Price shall be adjusted by 35% of such deficiency.
- (c) The cost of the Pogo Inventory for purposes of this <u>Section 2.1</u> shall be determined, at Buyer's election, pursuant to a review of the Sellers' books and records, a physical inventory taking of the Inventory (the "<u>Inventory Taking</u>") conducted in the presence in the presence of representatives of each of Sellers and Buyer, or a combination of both. To the extent that Buyer elects to conduct the Inventory Taking, it shall be conducted in accordance with the Buyer's reasonable instructions and the cost of the third party inventory taking service conducting the Inventory Taking shall be borne by Buyer.

SECTION 2.2 Deposit.

Buyer has made an earnest money deposit (the "<u>Deposit</u>") in the amount of Two Million Four Hundred Twenty Thousand Dollars (\$2,420,000.00) to Lindquist & Vennum PLLP, counsel to Sellers. The Deposit shall be applied against payment of the Purchase Price on the Closing Date. If this Agreement shall be terminated by any party hereto pursuant to <u>Section 8.1(a)</u>, (b), (d), (f), or (g) hereof, or in the event that a party other than Buyer or an Affiliate of Buyer purchases all or a significant portion of the Acquired Assets, then Sellers shall return the Deposit to Buyer within five (5) Business Days after Sellers' Representative's receipt of Buyer's written request therefor, except in the case of termination of this Agreement pursuant to <u>Section 8.1(f)</u> or (g) hereof, in which case Sellers shall return the Deposit to Buyer upon the closing of the alternative transaction. If this Agreement shall be terminated by Sellers pursuant to <u>Section 8.1(c)</u> or (e) hereof or otherwise by reason of the failure of any condition precedent under <u>Section 7.1</u> hereof resulting primarily from Buyer breaching any representation, warranty or covenant contained herein, then Sellers may retain the Deposit.

ARTICLE 3

CLOSING AND DELIVERIES

SECTION 3.1 Closing.

The consummation of the transactions contemplated hereby (the "Closing") shall take place at the offices of Lindquist & Vennum PLLP, 4200 IDS Center, 80 South Eighth Street, Minneapolis, Minnesota 55402 at 10:00 a.m., Minneapolis time, no later than fourteen (14) days following the satisfaction or waiver by the appropriate party of all the conditions contained in Article 7 or on such other date or at such other place and time as may be mutually agreed to by the parties (the "Closing Date"). All proceedings to be taken and all documents to be executed and delivered by all parties at the Closing shall be deemed to have been taken and executed simultaneously and no proceedings shall be deemed to have been taken nor documents executed or delivered until all have been taken, executed and delivered.

SECTION 3.2 Seller's Deliveries.

At the Closing, Sellers shall (a) sell, transfer, assign, convey and deliver the Acquired Assets to Buyer by bills of sale, endorsements, assignments and other instruments of transfer and conveyance reasonably satisfactory in form and substance to counsel for Buyer, including, without limitation, duly

executed copyright assignments, trademark assignments and patent assignment agreements in forms suitable for recording in the U.S. Trademark office or any similar office of a foreign country; and (b) deliver to Buyer the various certificates, consents, and documents referenced in <u>Section 7.2</u> hereof.

SECTION 3.3 Buyer's Deliveries.

At the Closing, Buyer shall (a) deliver to Sellers the Purchase Price (apportioned among the Sellers in accordance with the Apportionment Notice), less the Deposit; (b) assume the Assumed Liabilities by an assumption agreement reasonably satisfactory in form and substance to counsel for Sellers; and (c) deliver to Sellers the various certificates, consents, and documents referenced in Section 7.1 hereof.

ARTICLE 4

REPRESENTATIONS AND WARRANTIES

SECTION 4.1 Representations and Warranties of Sellers.

Sellers hereby represent and warrant to Buyer as follows:

- (a) <u>Organization</u>. Each Seller is duly organized, validly existing and in good standing under the laws of the State of Delaware. Each Acquired Subsidiary is duly organized, validly existing and in good standing under the laws of its respective jurisdiction of organization, each of which is set forth on <u>Schedule 4.1(a)</u>. Each Seller and Acquired Subsidiary has all requisite corporate or limited liability company power and authority to own its properties and assets and to conduct its businesses as now conducted.
- (b) <u>Qualification to Conduct Business</u>. Each Seller and Acquired Subsidiary is duly qualified to do business and is in good standing in every jurisdiction in which the character of the properties owned or leased by it or the nature of the businesses conducted by it makes such qualification necessary except where the failure to be so qualified would not result in a Material Adverse Effect.
- Authorization and Validity. Each Seller has, or on the Closing Date will have, as applicable, all requisite corporate or limited liability company power and authority to enter into this Agreement and any Ancillary Agreements to which each such Seller is or will become a party and, subject to the (i) Bankruptcy Court's entry of the Orders, and (ii) receipt of all Consents to perform each such Seller's obligations hereunder and thereunder, the execution and delivery of this Agreement and each Ancillary Agreement to which each such Seller is or will become a party and the performance of each such Seller's obligations hereunder and thereunder, have been, or on the Closing Date will be, duly authorized by all necessary corporate or limited liability company action of each such Seller, and no other corporate or limited liability company proceedings on the part of any Seller are necessary to authorize such execution, delivery and performance. This Agreement and each Ancillary Agreement to which each Seller is or will become a party have been, or on the Closing Date will be, duly executed by each such Seller, and, subject to the Bankruptcy Court's entry of the Orders, constitute, or will when executed and delivered constitute, each such Seller's valid and binding obligation, enforceable against each such Seller in accordance with their respective terms. The boards of directors or boards if managers (as applicable) of each Seller has resolved to request that the Bankruptcy Court approve this Agreement and the transactions contemplated hereby and each Ancillary Agreement to which each Seller is or will become a party.

- (d) Reports: Financial Statements. Schedule 4.1(d) contains (i) the consolidating unaudited balance sheets of Sellers and the Acquired Subsidiaries as of December 31, 2006 and December 31, 2007 and the related consolidating unaudited statements of earnings for the fiscal years then ended (the "Year-End Reports") and (ii) the consolidating unaudited balance sheet and statement of earnings of Sellers and the Acquired Subsidiaries for the fiscal year to date periods ended September 30, 2008 and December 18, 2008 (collectively, the "Interim Reports" and, together with the Year-End Financial Statements, the "Seller Reports"). Each of the Seller Reports included presents fairly, in all material respects, the financial position and results of operations of the Sellers as of the respective dates or for the respective periods set forth therein (it being understood that (A) such Seller Reports do not include any inter-company eliminations and (B) such Seller Reports include transactions with PGW and its affiliates (and, as such, the results of operations reflected in the Seller Reports may not be indicative of the results of operations that would be reflected if Sellers had not been affiliated with PGW and its affiliates).
- (e) No Conflict or Violation. Subject to the (i) receipt of all Consents and (ii) the Bankruptcy Court's entry of the Orders, the execution, delivery and performance by each Seller of this Agreement and each Ancillary Agreement to which any of them is or will become a party does not and will not (A) violate or conflict with any provision of the organizational documents (i.e., certificate of incorporation, certificate of formation, by-laws or operating agreement) of any Seller, (B) violate any provision of law, or any order, judgment or decree of any Government Authority applicable to any Seller except where any such violation would not result in a Material Adverse Effect, (C) result in or require the creation or imposition of any Liens (other than Permitted Liens) on any of the Acquired Assets or (D) violate or result in a breach of or constitute (with due notice or lapse of time or both) a default under any Acquired Contract entered into by Seller or by which the Seller is bound or to which the assets of Sellers are subject.
- (f) <u>Subsidiaries</u>. Except as set forth on <u>Schedule 4.1(f)</u>, there are no direct or indirect subsidiaries of Sellers other than the other Sellers and the Acquired Subsidiaries (to the extent that the Acquired Subsidiaries are considered to be direct or indirect subsidiaries of Sellers). Except as set forth on <u>Schedule 4.1(f)</u>, Sellers own, directly or indirectly, all of the issued and outstanding capital stock or other equity interests of each Acquired Subsidiary. There are no other shares of capital stock or other equity interests of any Acquired Subsidiary issued other than as set forth on <u>Schedule 4.1(f)</u>. The Acquired Equity Interests (i) have been duly authorized, validly issued, and are fully paid and nonassessable (in those jurisdictions in which such concepts are applicable), (ii) have not been issued in violation of any preemptive rights or of any terms of any agreement or other understanding binding upon any Acquired Subsidiary, and (iii) have been offered and sold in compliance with any and all applicable securities laws, rules and regulations. Upon the Closing, Buyer will be the sole owner of all right, title and interest in the Acquired Equity Interests.
- (g) <u>Consents and Approvals</u>. Other than pursuant to the HSR Act, <u>Schedule 4.1(g)</u> sets forth a true and complete list of each consent, waiver, authorization or approval of any Person and each material declaration to or filing or registration with any Government Authority that is required to be obtained by Sellers in connection with the execution and delivery by them of this Agreement and their respective Ancillary Agreements or the performance by them of their obligations hereunder or thereunder, including, without limitation, any and all material consents and approvals that are required to be obtained, or rights of first refusal, first offer or other similar preferential rights to purchase that are required to be complied with, in connection with the assignment or transfer of any Acquired Assets to Buyer in accordance with the terms of this Agreement (collectively, the "<u>Consents</u>").

- (h) <u>Litigation</u>. Except as set forth on <u>Schedule 4.1(h)</u>, there are no Claims, actions, suits, proceedings, orders or investigations pending or, to the knowledge of Sellers, threatened, that could reasonably be expected to affect the ability of Sellers to consummate the transactions contemplated by this Agreement and each Ancillary Agreement.
- (i) <u>Title to Acquired Assets</u>. Subject to the entry of the Orders, at the Closing, Sellers will obtain good and marketable title to, or a valid and enforceable right by Contract to use, the Acquired Assets which shall be transferred to Buyer free and clear of all Liens, other than Permitted Liens.
- Intellectual Property. Schedule 1.1(f) is an accurate and complete list of all (j) material items of Acquired Intellectual Property used in connection with the operation of the Business, as historically operated by Sellers, and, to the extent applicable, such list contains a complete and correct description of the owner, title (in the case of patents and copyrights) or trademark (in the case of trademarks), registration or application number, if in existence, and country of registration or application of each such listed item of Acquired Intellectual Property. Except as set forth on Schedule 4.1(j), no renewal and maintenance fees, annuities or other similar fees due and payable in respect of the Acquired Intellectual Property required to have been listed on Schedule 1.1(f) are overdue. Except as set forth on Schedule 4.1(j), the Acquired Intellectual Property except for that licensed to Sellers pursuant to an Acquired Contract is valid and enforceable and, to Sellers' knowledge, Sellers have not undertaken or omitted to undertake any acts that would invalidate, eliminate or materially impair the enforceability or scope of such rights and (ii) to Sellers' knowledge, no Claim has been made or is threatened challenging the legality, validity or enforceability of the Acquired Intellectual Property. Except as set forth on Schedule 4.1(j), to Sellers' knowledge, no rights in any Acquired Intellectual Property are being infringed, misappropriated or otherwise violated by any Person. Sellers are not aware of any diluting uses in the U.S. of the Polaroid trademark. Schedule 4.1(j) sets forth a complete and accurate list of all material agreements relating to the license of any Acquired Intellectual Property and Sellers have not received written notice from any party to any such agreement notifying Sellers of any grounds or circumstances that would give rise to a right of termination of any such agreement. Except as set forth on Schedule 4.1(j), no Seller is and, to Sellers' knowledge, no party to any such agreement (other than a Seller) is, in default under any such agreement. Other than as contemplated by this Agreement, Sellers have not licensed the rights to any of the Acquired Intellectual Property to any domestic or foreign Affiliates of Sellers that will continue following the Closing. Except as set forth on Schedule 4.1(j), all Intellectual Property used by Sellers in connection with the operation of the Business, as historically operated by Sellers, is part of the Acquired Intellectual Property transferred to Buyer hereunder.

(k) [Intentionally omitted]

- (1) <u>Permits</u>. Except as set forth on <u>Schedule 4.1(1)</u>, each Permit has been duly obtained, is valid and in full force and effect, and is not subject to any pending or, to the knowledge of Sellers, threatened administrative or judicial proceeding to revoke, cancel, suspend or declare such Permit invalid in any respect.
- (m) <u>Environmental Matters</u>. To the knowledge of Sellers, except as set forth on <u>Schedule 4.1(m)</u>, Sellers have provided or made available to Buyer copies of all information in possession of Sellers relating to the presence or migration of Hazardous Materials on, in or under the Real Property and the compliance with applicable Environmental Laws associated with activities conducted with respect to the Acquired Assets.
- (n) <u>Insurance</u>. Sellers, or their Affiliates, have maintained industry appropriate insurance at all times, and all premiums required to be paid under each insurance policy have been paid

when due, and all such policies are in full force and effect, with the exception of lawyer's E&O policies, which have not been renewed.

- (o) <u>Real Property</u>. Sellers have delivered to the Buyer correct and complete copies of the leases covering all leased Real Property listed in <u>Schedule 1.1(a)</u>.
- (p) <u>Brokerage and Finder's Fees</u>. No Seller, and none of Sellers' Affiliates or any of the officers or directors of any Seller or any Affiliate of any Seller, has any liability or obligation to pay any fees or commissions to any broker, finder or agent with respect to the transactions contemplated by this Agreement, with the exception of Houlihan, Lokey, Howard & Zukin.
- (q) <u>Disclaimer of Other Representations and Warranties</u>. Except as expressly set forth in this <u>Section 4.1</u>, Sellers make no representation or representation or warranty, express or implied, at law or in equity, in respect of Sellers, their Affiliates (including, but not limited to, the Acquired Subsidiaries), or their assets (including, *inter alia*, the Acquired Assets), liabilities (including, *inter alia*, the Assumed Liabilities) or operations (including, *inter alia*, the Business), including with respect to title, merchantability or fitness for any particular purpose and any such other representations or warranties are hereby expressly disclaimed. Buyer hereby acknowledges and agrees that, except to the extent expressly set forth in this <u>Section 4.1</u>, Buyer is purchasing the Acquired Assets on an "as is, where is, with all faults" basis and disclaims all warranties or guarantees, whether express or implied. Sellers shall not be liable in contract or in tort for any special, incidental, liquidated, punitive or consequential damages relating to the Acquired Assets. Buyer acknowledges and agrees that any consequences arising from Sellers' filing of the Bankruptcy Cases in accordance with the Agreement shall not be deemed a breach of any of the representations or warranties set forth in this Agreement
- (r) <u>Impaired Inventory</u>. A complete list of Impaired Inventory is set forth on <u>Schedule 4.1(r)</u>, which schedule shall be updated by Sellers on and as of the Closing Date.
- (s) <u>Contracts</u>. Attached hereto as <u>Schedule 4.1(s)</u> is a complete and accurate list of all material contracts to which Sellers are party.

SECTION 4.2 Representations and Warranties of Buyer.

Buyer hereby represents and warrants to Sellers as follows:

- (a) <u>Corporate Organization</u>. Buyer is a limited liability company duly organized, validly existing and in good standing under the laws of the State of Delaware and has all requisite limited liability company power and authority to own its properties and assets and to conduct its businesses as now conducted.
- (b) <u>Authorization and Validity</u>. Buyer has, or on the Closing Date will have, all requisite limited liability company power and authority to enter into this Agreement, any Ancillary Agreement to which Buyer is or will become a party, to issue the Equity Consideration as contemplated hereby and to perform its obligations hereunder and thereunder. The execution and delivery of this Agreement and any Ancillary Agreement to which Buyer is or will become a party and the performance of Buyer's obligations hereunder and thereunder have been, or on the Closing Date will be, duly authorized by all necessary limited liability company action by the Board of Managers of Buyer, and no other corporate proceedings on the part of Buyer are necessary to authorize such execution, delivery and performance. This Agreement and each Ancillary Agreement to which Buyer is or will become a party have been, or on the Closing Date will be, duly executed by Buyer and constitute, or will constitute, when executed and delivered, Buyer's valid and binding obligations, enforceable against it in accordance with

their respective terms except as may be limited by bankruptcy or other laws affecting creditors' rights and by equitable principles.

- (c) No Conflict or Violation. The execution, delivery and performance by Buyer of this Agreement (including the issuance of the membership interests comprising the Equity Consideration) and any Ancillary Agreement to which Buyer is or will become a party do not and will not (i) violate or conflict with any provision of the certificate of formation or limited liability company agreement of Buyer, (ii) violate any provision of law, or any order, judgment or decree of any court or Government Authority applicable to Buyer or (iii) violate or result in a breach of or constitute (with due notice or lapse of time or both) a default under any Contract to which Buyer is party or by which Buyer is bound or to which any of Buyer's properties or assets is subject.
- (d) <u>Consents and Approvals</u>. Other than pursuant to the HSR Act, no consent, waiver, authorization or approval of any Person and no declaration to or filing or registration with any Government Authority is required in connection with the execution and delivery by Buyer of this Agreement and each Ancillary Agreement to which Buyer is or will become a party or the performance by Buyer of its obligations hereunder or thereunder (including the issuance of the membership interests comprising the Equity Consideration).
- (e) <u>Adequate Assurances Regarding Acquired Contracts</u>. Buyer is (or, after the Closing, will be) capable of satisfying the conditions contained in Sections 365(b)(1)(C) and 365(f)(2)(B) of the Bankruptcy Code with respect to the Acquired Contracts.
- (f) <u>Brokerage and Finder's Fees</u>. None of Buyer, its Affiliates or any of their officers or directors has any liability or obligation to pay any fees or commissions to any broker, finder or agent with respect to the transactions contemplated by this Agreement.
- (g) <u>Buyer's Financing</u>. Buyer has, and at the Closing shall have, available under its current credit facilities, or otherwise, the funds necessary to consummate the transactions set forth in this Agreement, including, *inter alia*, the payment of the Purchase Price and the Deposit.
- (h) Capitalization. As of the Closing, (i) the membership interests comprising the Equity Consideration shall include all of the terms set forth in the Limited Liability Company Agreement (the "Membership Interests") and shall be duly authorized, validly issued, fully paid and nonassessable, (ii) the Membership Interests shall be issued to Sellers Representative free and clear of any and all Encumbrances and not in violation of any contractual obligation (whether verbal or written) binding upon Buyer or any of its Affiliates and (iii) Sellers' Representative shall own 25% of the fully-diluted equity of a newly form limited liability company which will hold the Intellectual Property purchased hereunder. Exhibit A to the Limited Liability Company Agreement sets forth, under the heading "Pro Forma Capitalization," the true and complete capitalization of Buyer immediately following the consummation of the transactions contemplated hereby. Except as contemplated by this Agreement, there are no agreements, arrangements, options, warrants, calls, rights or commitments of any character relating to the issuance, sale, purchase or redemption of any equity interest in Buyer. No holder of any equity interest in Buyer has any preemptive or other rights to acquire any such equity interests. As of the Closing, true and complete copies of the Certificate of Formation and Limited Liability Company Agreement, in each case as amended to date and taking into account the issuance of the Membership Interests contemplated hereby, shall have been delivered to Sellers' Representative.
- (i) <u>No Undisclosed Liabilities; No Subsidiaries</u>. Buyer is not subject to any liability or obligation of any kind or nature, whether absolute, contingent, accrued or otherwise. Buyer does not have any subsidiaries or any equity ownership interest in any Person. PLR IP Holdings, LLC, a Delaware

limited liability company ("<u>PLR Holdings</u>"), is an Affiliate of Buyer that will own and hold all of the Acquired Intellectual Property following the consummation of the transactions contemplated hereby.

SECTION 4.3 Warranties Are Exclusive.

The parties acknowledge that the representations and warranties contained in this <u>Article 4</u> are the only representations or warranties given by the parties and that all other express or implied warranties are disclaimed.

ARTICLE 5

COVENANTS AND OTHER AGREEMENTS

SECTION 5.1 Pre-Closing Covenants of Sellers.

Sellers covenant to Buyer that, during the period from the Execution Date through and including the Closing Date or the earlier termination of this Agreement:

- (a) <u>General</u>. Sellers will use reasonable efforts to take all action and to do all things necessary, proper, or advisable in order to consummate and make effective the transactions contemplated by this Agreement (including satisfaction, but not waiver, of the Closing conditions set forth in <u>Article 7</u> herein).
- (b) Operation of Business. Subject to any restrictions and obligations imposed by the Bankruptcy Court and applicable law, Sellers will not engage in any practice, take any action, or enter into any transaction outside the ordinary course of business, and Sellers (to the extent Sellers are able to control the actions of the Acquired Subsidiaries) will cause the Acquired Subsidiaries to not engage in any practice, take any action, or enter into any transaction outside the ordinary course of business. In particular, Sellers will refrain from doing any of the following in respect of the Acquired Assets or the operation of the Business: (i) disposing of, or transferring, any material Acquired Asset except for the sale of such assets in the ordinary course of business, (ii) amending, terminating or modifying the material terms of any of the Acquired Contracts, or (iii) making any change in the compensation payable or to become payable to the employees, other than increases or promotions in the ordinary course of business consistent with past practice; provided, however, notwithstanding the preceding, Sellers may in their reasonable discretion take such actions in connection with or as a result of the consequences (adverse or otherwise) of filing the Bankruptcy Cases, including the establishment of employee retention programs and negotiating the amounts, if any, to cure defaults in respect of the Acquired Contracts.
- (c) <u>Access to Records and Properties</u>. Buyer shall be entitled, and Sellers shall permit Buyer, to conduct such investigation of the condition (financial or otherwise), businesses, assets, properties or operations of the Seller Group as Buyer shall reasonably deem appropriate.
- (i) Sellers shall (A) provide Buyer access at any reasonable time during normal business hours to the facilities and offices of the Seller Group and to all of the books and records of the Seller Group, including, without limitation, to perform field examinations and inspections of the Seller Group's inventories, facilities, equipment and other assets and properties; and (B) cause the Seller Group's representatives to furnish Buyer with such financial and operating data and other information with respect to the condition (financial or otherwise) of the Business and the Seller Group's assets, properties or operations as Buyer shall reasonably request; <u>provided</u>, <u>however</u>, that Buyer and its representatives shall (x) coordinate all requests for access and information with the Chief Executive

Officer of Sellers' Representative and (y) use all commercially reasonable efforts to prevent any such investigation from interfering in any way with the operation of the business of the Seller Group.

- (ii) Buyer will treat and hold as such any Confidential Information it receives from the Seller Group, including, without limitation, any Confidential Information it received prior to the Execution Date. Buyer will not use any of the Confidential Information it (or its representatives) receives in connection with the transactions contemplated by this Agreement except in connection with this Agreement. If this Agreement is terminated for any reason whatsoever, Buyer will return to the Seller Group all tangible embodiments (and all copies) of the Confidential Information which are in its possession.
- (d) <u>Notice of Certain Events</u>. Sellers' Representative shall promptly notify Buyer of, and furnish Buyer any information it may reasonably request with respect to, the occurrence of any event or condition or the existence of any fact that would reasonably be expected to cause any of the conditions to Buyer's obligations to consummate the transaction(s) contemplated by this Agreement or by any Ancillary Agreement not to be fulfilled. Any such notice will be deemed to (i) amend any applicable Schedule, (ii) have qualified the representations and warranties contained in <u>Section 4.1</u> of this Agreement, and (iii) have cured any misrepresentation or breach of warranty that otherwise might have existed hereunder by the occurrence of such event or condition or the existence of such fact.
- (e) <u>Assignment of Transferred Internet Domain Names</u>. Sellers shall use commercially reasonable efforts to cause their Affiliates who are the registered holders of the Transferred Internet Domain Names to assign and transfer to Sellers, at or prior to the Closing, all of such Transferred Internet Domain Names, other than those set forth on <u>Schedule 5.1(e)</u> (the "<u>Subject Internet Domain Names</u>"). The Transferred Internet Domain Names (other than the Subject Internet Domain Names), shall be transferred to Buyer at the Closing pursuant to <u>Section 1.1</u> hereof.

SECTION 5.2 Pre-Closing Covenants of Buyer.

Buyer covenants to Sellers that, during the period from the Execution Date through and including the Closing Date or the earlier termination of this Agreement:

- (a) <u>General</u>. Buyer will use reasonable efforts to take all action and to do all things necessary, proper, or advisable in order to consummate and make effective the transactions contemplated by this Agreement (including satisfaction, but not waiver, of the Closing conditions set forth in <u>Article 7</u> herein).
- (b) <u>Adequate Assurances Regarding Acquired Contracts and Required Orders.</u> With respect to each Acquired Contract, Buyer shall provide adequate assurance of the future performance of such Acquired Contract by Buyer. Buyer shall take such actions as may be reasonably requested by Sellers to assist Sellers in obtaining the Bankruptcy Court's entry of the Orders and any other order of the Bankruptcy Court reasonably necessary to consummate the transactions contemplated by this Agreement.
- (c) <u>Permits.</u> Buyer shall use commercially reasonable efforts to promptly obtain or consummate the transfer to Buyer of any Permit required to own or operate the Acquired Assets under applicable laws.
- (d) <u>Notice of Certain Events</u>. Buyer shall promptly notify Sellers' Representative of, and furnish Sellers' Representative any information it may reasonably request with respect to, the occurrence of any event or condition or the existence of any fact that would reasonably be expected to (A) cause any of the conditions to Sellers' obligations to consummate the transactions contemplated by this

Agreement or by any Ancillary Agreement not to be fulfilled or (B) adversely affect the value of the Equity Consideration. Any such notice will be deemed to (i) amend any applicable Schedule, (ii) have qualified the representations and warranties contained in Section 4.2 of this Agreement, and (iii) have cured any misrepresentation or breach of warranty that otherwise might have existed hereunder by the occurrence of such event or condition or the existence of such fact.

- (e) <u>Access to Records and Properties</u>. Sellers shall be entitled, and Buyer shall permit Sellers, to conduct such investigation of the condition (financial or otherwise), businesses, assets, properties or operations of the Buyer Group as Sellers shall reasonably deem appropriate.
- (i) Buyer shall, to the extent relating to Sellers' evaluation of the Equity Consideration, (A) provide Sellers access at any reasonable time during normal business hours, to all of the books and records of the Buyer Group and (B) cause the Buyer Group's representatives to furnish Sellers with such financial and operating data and other information with respect to the condition (financial or otherwise) Buyer Group's assets, properties or operations as Buyer shall reasonably request; provided, however, that Sellers and their representatives shall (x) coordinate all requests for access and information with those parties of Buyer listed in Section 10.7 hereof and (y) use all commercially reasonable efforts to prevent any such investigation from interfering in any way with the operation of the business of the Buyer Group.
- (ii) Sellers will treat and hold as such any Confidential Information they receive from the Buyer Group, including, without limitation, any Confidential Information they received prior to the Execution Date. Sellers will not use any of the Confidential Information they (or their representatives) receive in connection with the transactions contemplated by this Agreement except in connection with this Agreement. If this Agreement is terminated for any reason whatsoever, Sellers will return to the Buyer Group all tangible embodiments (and all copies) of the Confidential Information which are in their possession.

SECTION 5.3 Other Covenants of Sellers and Buyer.

- (a) <u>Filings</u>. During the period from the Execution Date through and including the Closing Date or the earlier termination of this Agreement, Sellers and Buyer shall cooperate with one another in obtaining all authorizations, consents, and approvals of Government Authorities that may be or become necessary in connection with the consummation of the transactions contemplated by this Agreement, and to take all reasonable actions to avoid the entry of any order or decree by Government Authorities prohibiting the consummation of the transactions contemplated hereby, and shall furnish to the other all such information in its possession as may be necessary for the completion of the notifications to be filed by the other.
- (b) <u>Improper Receipt of Payment</u>. From and after the Closing Date, (i) Sellers shall promptly forward to Buyer any and all payments received by Sellers from customers or any other Persons that constitute part of the Acquired Assets and (ii) Buyer shall promptly forward to Sellers any and all payments received by Buyer from customers or any other Persons that constitute part of the Excluded Assets.
- (c) <u>Disclosure Schedule Supplements</u>. Sellers' Representative, on the one hand, shall notify Buyer of, and Buyer on the other hand, shall notify Sellers' Representative of, and shall supplement or amend the Schedules to this Agreement with respect to, any matter that (i) may arise after the delivery of the Schedules hereunder and that, if existing or occurring at or prior to such delivery of the Schedules, would have been required to be set forth or described in the Schedules to this Agreement or (ii) makes it necessary to correct any information in the Schedules to this Agreement or in any

representation and warranty of Sellers or Buyer, as applicable, that has been rendered inaccurate thereby. Each such notification and supplementation shall be made no later than three (3) days before the date set for the Closing by the parties, it being understood that, except with respect to the addition to the Schedules of Material Contracts entered into after the date hereof in accordance with Section 5.1, information contained in any supplement to the Schedules provided pursuant to this Section 5.3 shall not be deemed to cure any breach for purposes of Article 7 hereof.

- Transition Services. For the period commencing on the Closing Date and ending on the later of (x) the closing of the Bankruptcy Cases and (y) five (5) years after the Closing Date (the "Transition Period"), Sellers and their representatives shall have reasonable access to, and shall have the right to photocopy at their own expense, all of the following (to the extent the same are transferred to Buyer hereunder): books and records, including any computerized databases and files and programs and associated software relating to the pre-Closing operations and business of the Seller Group and/or the Acquired Assets and Assumed Liabilities as they existed as of the Closing Date, including but not limited to (i) the investigation, evaluation and prosecution of any and all claims and Causes of Action retained by Sellers, (ii) the evaluation, allowance, distribution and defense of any and all claims brought against Sellers, their Affiliates or their estates, and (iii) employee's records or other personnel and medical records, as of the Closing Date, required by law, legal process or subpoena. During the Transition Period, Buyer agrees to provide Sellers and any of their representatives, upon reasonable request and notice, with reasonable access to employees of Buyer (who may be former employees of Sellers) for purposes of, among other things, winding down the estates of Sellers, pursuing Claims against Persons and completing the Bankruptcy Cases. Access pursuant to this Section shall be afforded by Buyer upon receipt of reasonable advance notice, during normal business hours. Sellers and their representatives agree to treat confidentially any information obtained pursuant to this Section, including the books and records. If Buyer shall desire to dispose of any such books and records upon or prior to five (5) years after the Closing Date, Buyer shall, prior to such disposition, give Sellers a reasonable opportunity at Sellers' expense, to segregate and remove such books and records as Sellers may select.
- (e) <u>Subject Internet Domain Names</u>. If the Subject Internet Domain Names are not transferred to Buyer (or its designee) at the Closing, then Sellers shall, promptly following the Closing, cause their Affiliates who are the registered holders of the Subject Internet Domain Names to assign and transfer such Subject Internet Domain Names to Buyer (or its designee), such transfers to be effected free and clear of all Liens, Claims and Encumbrances.
- (f) <u>Collection of Accounts Receivable</u>. Following the Closing, Buyer shall use commercially reasonable efforts to collect the Accounts Receivable.

SECTION 5.4 Ownership and Use of Polaroid Name.

(a) Sellers covenant and agree that Sellers shall, and Sellers shall cause (i) all of their Affiliates (other than the Acquired Subsidiaries and the Specified Entities) and (ii) PGW and each of its Affiliates which use the Polaroid Name (other than the Acquired Subsidiaries and the Specified Entities), to pass all required resolutions to amend their respective certificates of incorporation or formation or other organizational documents to change their corporate or company name to a name that does not include the word "Polaroid" or any name intended or likely to be confused or associated with the Polaroid Name or product no later than the Closing Date. Sellers covenant and agree that Sellers shall cause each of the Specified Entities to pass all required resolutions and to amend their respective certificates of incorporation or formation or other organizational documents to change their corporate or company name to a name that does not include the word "Polaroid" or any name intended or likely to be confused or associated with the Polaroid Name or product no later than sixty (60) days following the Closing Date.

The parties acknowledge and agree that (i) Polaroid Asia Pacific Services Limited, a company organized under the laws of Hong Kong, is in the process of deregistration, (ii) Polaroid (Italia) S.p.A. is in the process of liquidating and (iii) Polaroid Eyewear (France) EURL is in the process of liquidating, and that, accordingly, in each instance, no action shall be required to change the name of such entities in connection with the consummation of the transactions contemplated hereby, provided, that, to the extent each such entity is still in the process of deregistration or liquidation, as applicable, on June 30, 2009, with respect to such entities, Sellers covenant and agree that Sellers shall cause each of such entities to pass all required resolutions and to amend their respective certificates of incorporation or formation or other organizational documents to change their corporate or company name to a name that does not include the word "Polaroid" or any name intended or likely to be confused or associated with the Polaroid Name or product no later than July 31, 2009.

- (b) Sellers acknowledge that the Polaroid Name shall be and remain, subsequent to the Closing, the sole and exclusive property of Buyer.
- (c) Subsequent to the Closing, subject to <u>Sections 5.4(d)</u> and <u>5.10</u>, none of Sellers or any of their Affiliates (other than the Acquired Subsidiaries) shall have any right, title or interest in or to, and Buyer is not granting Seller or any of its Affiliates, a license to use, the Polaroid Name.
- (d) The obligations in this <u>Section 5.4</u> shall not apply (i) to the extent use of the Polaroid Name is required by law or otherwise reasonably required pending the change of corporate names (as set out in this <u>Section 5.4</u>), (ii) to the extent use of the Polaroid Name is reasonably required in order to enable collection or payment of invoices issued by Seller or any of its Affiliates, (iii) to the extent the Polaroid Name is reasonably required in connection with the Bankruptcy Cases or (iv) to the extent use of the Polaroid Name is reasonably required to pursue a Claim.

SECTION 5.5 Bankruptcy Actions.

- (a) <u>Bankruptcy Sale Motion</u>. Sellers have filed with the Bankruptcy Court a motion seeking approval of the Bankruptcy Bidding Procedures Order and the Bankruptcy Sale Order (the "Bankruptcy Sale Motion").
- (b) <u>Bankruptcy Bidding Procedures Order</u>. For purposes of this Agreement, "<u>Bankruptcy Bidding Procedures Order</u>" shall mean the order of the Bankruptcy Court entered on February 21, 2009 and attached as <u>Exhibit B</u> hereto (A) approving sale procedures and bidding protections in connection with the sale of substantially all of the debtors' assets pursuant to Sections 363 and 365 of the Bankruptcy Code; (B) scheduling an auction and hearing to consider approval of the sale of substantially all of the debtors' assets (the "Auction"); and (C) granting related relief.
- Sale Order" shall mean the order of the Bankruptcy Court entered pursuant to Sections 363 and 365 of the Bankruptcy Code in the form attached as Exhibit C hereof, in part, (A) approving this Agreement and the transactions contemplated hereby; (B) approving the sale of the Acquired Assets to Buyer free and clear of all Liens, Claims and Encumbrances pursuant to Section 363(f) of the Bankruptcy Code, (C) finding that Buyer is a good-faith purchaser entitled to the protections of Section 363(m) of the Bankruptcy Code; (D) confirming that Buyer is acquiring the Acquired Assets free and clear of the Excluded Assets and the Excluded Liabilities; (E) providing that the provisions of Rules 6004(g) and 6006(d) of the Federal Rules of Bankruptcy Procedure are waived and there will be no stay of execution of the Bankruptcy Sale Order under Rule 62(a) of the Federal Rules of Civil Procedure; and (F) retaining jurisdiction of the Bankruptcy Court to interpret and enforce the terms and provisions of this Agreement.

- (d) <u>Bankruptcy Assignment Motion and Order</u>. Sellers have filed with the Bankruptcy Court a motion for an order (the "<u>Bankruptcy Assignment Order</u>") authorizing the assumption and assignment pursuant to Section 365 of the Bankruptcy Code of the Acquired Contracts (the "<u>Bankruptcy Assignment Motion</u>"). Buyer agrees to cooperate with Sellers in connection with furnishing information pertaining to the satisfaction of the requirement of adequate assurances of future performance as required under Section 365(f)(2)(B) of the Bankruptcy Code.
- (e) <u>Notice and Reasonable Efforts</u>. Sellers have provided notice of the hearing(s) on the Bankruptcy Sale Motion, the Bankruptcy Assignment Motion, and the Auction as required by the Bankruptcy Code and the Bankruptcy Rules and other applicable law to all parties entitled to notice, including, but not limited to, all parties to the Acquired Contracts and all taxing and environmental authorities in jurisdictions applicable to Sellers. Sellers have also published notice of the hearing(s) on the Bankruptcy Sale Motion, the Bankruptcy Assignment Motion, and the Action in at least one (1) national newspaper and one (1) regional newspaper. Sellers shall take all actions as may be reasonably necessary to cause the Orders to be issued, entered and become a Final Order.
- (f) <u>Excluded Assets.</u> Notwithstanding anything herein to the contrary, Sellers shall have the right, in their sole discretion (without any obligation to notify or otherwise consult with Buyer) to reject any or all Excluded Assets, including any or all contracts other than the Acquired Contracts, and to take all actions necessary to effectuate any such rejection or rejections, including prosecuting a motion in the Bankruptcy Court seeking authorization, as necessary, to reject such Excluded Assets.
- (g) <u>Appeals of Bankruptcy Orders</u>. If, following the Closing, any of the Orders or any other order of the Bankruptcy Court relating to this Agreement shall be appealed by any Person (or a petition for certiorari or motion for rehearing or reargument shall be filed with respect thereto), Buyer shall take all steps as may be reasonable and appropriate to defend against such appeal, petition or motion, at Buyer's sole cost and Sellers agree to cooperate in such efforts, and each party hereto shall endeavor to obtain an expedited resolution of such appeal.

SECTION 5.6 Employees.

- (a) From and after the Execution Date through and including the Closing, Buyer shall have reasonable access (during normal business hours and so as not to disrupt Sellers' operations) to all of Sellers' employees for the purpose of Buyer's customary screening and testing for new hires (which Buyer shall undertake in compliance with applicable law), provided that, at Sellers' option, all such access shall be overseen by such employees' supervisors. Buyer may offer to employ any of Sellers' employees on terms to be determined by Buyer, subject to the provisions of this Section 5.6. Sellers shall reasonably cooperate with Buyer in its efforts to extend offers of employment to employees upon the conditions described above.
- (b) Buyer shall offer employment to those employees (the "<u>Offered Employees</u>") listed on <u>Schedule 5.6(b)</u>, which schedule Buyer may update through and including the Closing Date.
- (c) It is understood and agreed that (i) Buyer's expressed intention to extend offers of employment as set forth in this Section 5.6 will not constitute a contract (express or implied) on the part of Buyer to a post-Closing employment relationship of any fixed term or duration or upon any terms or conditions other than those that Buyer may establish pursuant to individual offers of employment, and (ii) employment offered by Buyer is "at will" and may be terminated by Buyer or by an employee at any time for any reason (subject to any written commitments to the contrary made by Buyer or an employee and applicable laws governing employment). Nothing in this Agreement will be deemed to prevent or restrict in any way the right of Buyer to terminate, reassign, promote or demote any of the Offered

Employees after the Closing, or to change adversely or favorably the title, powers, duties, responsibilities, functions or locations, or terms or conditions of employment of such employees; <u>provided</u>, <u>however</u>, that Sellers shall not be responsible for and shall have no liability for any such action by the Buyer from and after the Closing.

(d) Sellers shall terminate the employment of each Offered Employee effective immediately before the Closing, and Buyer shall commence the employment of each Offered Employee with Buyer effective immediately thereafter. The defense of any claims and the satisfaction of any liabilities relating to or arising out of the employment, or cessation of employment, of any employee of Sellers or their Affiliates (whether or not an Offered Employee) on or before the Closing Date, and any claims for wages, benefits, or other remuneration allegedly due through the Closing, shall be the sole responsibility of Sellers.

SECTION 5.7 Employee Benefit Plans and Pre-Closing Employment Liabilities.

Buyer shall not assume any Employee Benefit Plan or liability or obligation under any Employee Benefit Plan, or any liability or obligation under any contract, payroll practice or other arrangement that any Seller sponsors, contributes to, or participates in (or may have sponsored, contributed to or participated in), or that any Seller has or may have any liability or obligation under, whether or not disclosed under this Agreement or any Schedule. Without limiting the generality of the foregoing, Buyer shall not be responsible for (i) the payment of any wages and other remuneration of any current or former employees with respect to their services as employees of Sellers through the Closing, including pro rata bonus and retention payments (including any key employee retention plan approved by the Bankruptcy Court), if any, earned before the Closing, and (ii) the payment to active employees of any termination or severance payments under Employee Benefit Plans or other benefit obligations of the Sellers and the provision of group health plan continuation coverage in accordance with the requirements of ERISA Section 601 et seq. and Code Section 4980 B.

SECTION 5.8 Other Actions.

Sellers will give any notices required by applicable law and take whatever other actions as may be necessary to carry out the arrangements described in <u>Sections 5.6</u> and <u>5.7</u>.

SECTION 5.9 [Intentionally Omitted].

SECTION 5.10 Use of Names.

days after the Closing Date, each of the Sellers' respective controlled Affiliates (collectively, the "Subject Affiliates") shall have a non-exclusive, royalty-free, non-transferable license in the Territory to use any and all of the trademarks, service marks, trade dress, logos, trade names and corporate names included in the Acquired Intellectual Property located on (i) any finished goods and other supplies owned by the Subject Affiliates as of the Closing Date (excluding, for these purposes, the Subject Inventory, which is the subject of Section 5.10(b) below), (ii) any advertising or promotional materials used by the Subject Affiliates as of the Closing Date and (iii) any stationery, business cards, business forms and other similar items owned by any of the Subject Affiliates as of the Closing Date; provided, however, that the Sellers shall only permit the Subject Affiliates to use such trademarks, service marks, trade dress, logos, trade names and corporate names in the operation of their respective businesses in the pre-Petition Date ordinary course consistent with past practice (including, without limitation, wholesale and retail channels of distribution, and marketing and advertising campaigns), it being understood and agreed that the Sellers shall not permit the Subject Affiliates to use any such Acquired Intellectual Property in connection with

any other activities. Without limiting the foregoing, during such 120-day period, the Sellers shall not permit any Subject Affiliate to (the "<u>Use Restrictions</u>"): (a) hold itself out as having any affiliation or relationship of any kind with Buyer or any Affiliate thereof or (b) use such Acquired Intellectual Property in a manner that would be reasonably likely to reflect negatively thereon or on Buyer or its Affiliates. In the event Sellers (A) receive an offer for a bulk sale of Inventory from a Person (who is not an Affiliate of Buyer) consistent with this <u>Section 5.10(a)</u> (a "<u>Bulk Sale Offer</u>") and (B) desire to sell such Inventory pursuant to the terms of such Bulk Sale Offer, then Sellers shall promptly notify Buyer of such offer (a "<u>ROFR Notice</u>") and Buyer shall have the right, within 48 hours of such notice, to acquire such Inventory on the same terms as set forth in the ROFR Notice. The parties acknowledge and agree that the Subject Affiliates shall be entitled to enforce the provisions of this <u>Section 5.10(a)</u>. For purposes hereof, the "Territory" shall mean worldwide excluding the United States.

(b) Notwithstanding anything contained herein to the contrary, for a period of twelve (12) months after the Closing Date, each of the Subject Affiliates shall have a non-exclusive, royalty-free, non-transferable license in the Territory to use, subject to the Use Restrictions, any and all of the trademarks, service marks, trade dress, logos, trade names and corporate names included in the Acquired Intellectual Property in connection with selling the Subject Inventory through the polapremium.com channel and the channels currently used by Nippon Polaroid Kabushiki Kaisha (the "NPKK Channels"); provided, however, that Buyer shall have the right, upon ten (10) days prior written notice, to purchase all remaining Subject Inventory from the Subject Affiliates at the price then payable by polapremium.com or, as applicable, offered through the NPKK Channels. The parties acknowledge and agree that the Subject Affiliates shall be entitled to enforce the provisions of this Section 5.10(b).

SECTION 5.11 Certain Affiliate Arrangements. Effective at the Closing, all Contracts between any Acquired Subsidiary, on the one hand, and any Seller or any of their respective Affiliates, on the other hand, shall be terminated as between them and shall be without any further force and effect, and there shall be no further obligations of any of the relevant parties thereunder. Without limiting the foregoing, all inter-company accounts, whether payables or receivables, or other amounts or obligations between any Seller or any of their respective Affiliates, on the one hand, and any Acquired Subsidiary, on the other hand, shall, effective as of the Closing, be extinguished and terminated with no amounts being paid or assets being transferred in connection therewith. At or prior to the Closing, Sellers shall deliver to Buyer such documentation evidencing compliance with this Section 5.11 as Buyer may reasonably request.

ARTICLE 6

TAXES

SECTION 6.1 Taxes Related to Purchase of Acquired Assets.

All Taxes, including, without limitation, all state and local Taxes in connection with the transfer of the Acquired Assets, and all recording charges (including, but limited to, any and all recording charges and other fees required for the recording of any copyright assignments, trademark assignments and patent assignments in the U.S. Patent and Trademark office or any similar office of a foreign country), registration fees, conveyance fees and other fees and charges (collectively, "Transaction Taxes"), that may be imposed by reason of the sale, transfer, assignment and delivery of the Acquired Assets shall be borne by Buyer. Buyer and Sellers shall cooperate to (a) determine the amount of Transaction Taxes payable in connection with the transactions contemplated under this Agreement, (b) provide all requisite exemption certificates and (c) prepare and file any and all required Tax Returns for or with respect to such Transaction Taxes with any and all appropriate Government Authorities. Without the prior written consent of Sellers' Representative (which Sellers' Representative may provide or withhold in its sole and

absolute discretion), Buyer shall not be entitled to make an election under Section 338(h)(10) of the Code with respect to Buyer's acquisition of the shares of capital stock or other ownership interest of the Acquired Subsidiaries.

SECTION 6.2 Cooperation on Tax Matters.

- (a) Buyer and Sellers shall furnish or cause to be furnished to each other, as promptly as practicable, such information and assistance relating to the Acquired Assets and the Assumed Liabilities as is reasonably necessary for the preparation and filing of any Tax Return, claim for refund or other required or optional filings relating to Tax matters, for the preparation for and proof of facts during any Tax audit, for the preparation for any Tax protest, for the prosecution or defense of any suit or other proceeding relating to Tax matters and for the answer to any Government Authority relating to Tax matters.
- (b) Sellers shall retain possession of all Tax Records for a period of at least six (6) years from the Closing Date. Sellers shall give Buyer notice and an opportunity to retain any Tax Records in the event that Sellers determine to destroy or dispose of them after such period. In addition, from and after the Closing Date, Sellers shall provide access to Buyer and its representatives (after reasonable notice and during normal business hours and without charge), to the Tax Records as Buyer may reasonably deem necessary to properly prepare for, file, prove, answer, prosecute and/or defend any Tax Return, claim, filing, tax audit, tax protest, suit, proceeding or answer.
- (c) Buyer shall retain possession of all accounting, business and financial records and information (other than Tax Records) (i) relating to the Acquired Assets or the Assumed Liabilities that are in existence on the Closing Date and transferred to Buyer hereunder and (ii) coming into existence after the Closing Date that relate to the Acquired Assets or the Assumed Liabilities before the Closing Date, for a period of at least six (6) years from the Closing Date. Buyer shall give Sellers notice and an opportunity to retain any such records in the event that Buyer determines to destroy or dispose of them after such period. In addition, from and after the Closing Date, Buyer shall provide access to Sellers and their representatives (after reasonable notice and during normal business hours and without charge), to the books, records, documents and other information relating to the Acquired Assets or the Assumed Liabilities as Sellers may reasonably deem necessary to (i) properly prepare for, file, prove, answer, prosecute and/or defend any such Tax Return, claim, filing, tax audit, tax protest, suit, proceeding or answer, (ii) administer or complete any cases under Chapter 11 of the Bankruptcy Code of Sellers or (iii) pursue any Claim. Such access shall include, without limitation, access to any computerized information retrieval systems relating to the Acquired Assets or the Assumed Liabilities.

SECTION 6.3 Allocation of Purchase Price.

Buyer and Sellers' Representative shall agree to an allocation of the Purchase Price among the Acquired Assets (the "Allocation"). Such Allocation will be binding upon Buyer and Sellers and their respective successors and assigns, and none of the parties to this Agreement will take any position (whether in returns, audits or otherwise) that is inconsistent with the Allocation. Buyer and Sellers will report the purchase and sale of the Acquired Assets on all tax returns, including, without limitation, Form 8594 as provided for in Section 1060 of the Code, in accordance with the Allocation and will cooperate in timely filing with the Internal Revenue Service their respective Forms 8594.

ARTICLE 7

CONDITIONS PRECEDENT TO PERFORMANCE BY PARTIES

SECTION 7.1 Conditions Precedent to Performance by Sellers.

The obligation of Sellers to consummate the transactions contemplated by this Agreement is subject to the fulfillment, at or before the Closing, of the following conditions, any one or more of which (other than the conditions contained in <u>Section 7.1(c)</u>) may be waived in writing by Sellers' Representative, in its sole discretion:

- (a) <u>Representations and Warranties of Buyer</u>. The representations and warranties of Buyer made in <u>Section 4.2</u> of this Agreement, in each case, shall be true and correct as of the Execution Date and in all material respects as of the Closing Date as though made by Buyer as of the Closing Date, except to the extent representations and warranties expressly relate to an earlier date, in which case such representations and warranties shall be true and correct on and as of such earlier date.
- (b) <u>Performance of the Obligations of Buyer</u>. Buyer shall have (i) performed in all material respects all obligations required under this Agreement or any Ancillary Agreement to which it is party which are to be performed by it on or before the Closing Date (except with respect to any obligations qualified by materiality, which obligations shall be performed in all respects as required under this Agreement), (ii) taken all actions in connection with the consummation of the transactions contemplated hereby, and (iii) delivered all certificates, instruments, and other documents required herein or otherwise in a form and substance reasonably satisfactory to Sellers required to effect the transactions contemplated hereby.
- (c) Governmental Consents and Approvals. If applicable, the waiting period applicable to the consummation of the transactions contemplated by this Agreement under the HSR Act shall have expired or been terminated. The Orders shall have been entered and shall not have been stayed. Notwithstanding the foregoing, nothing in this Agreement shall preclude Buyer from consummating the transactions contemplated herein if Buyer, in its sole discretion, waives the requirement that the Sale Order shall have become a Final Order. No notice of such waiver of this condition or any other condition to the Closing need be given except to Sellers, it being the intention of the parties that Buyer shall be entitled to, and is not waiving, the protection of Section 363(m) of the Bankruptcy Code, the mootness doctrine and any similar statute or body of law if the Closing occurs in the absence of the Bankruptcy Sale Order becoming a Final Order.
- (d) <u>No Violation of Orders</u>. No preliminary or permanent injunction or other order of any court or Government Authority that declares this Agreement invalid or unenforceable in any material respect or which prevents the consummation of the transactions contemplated hereby shall be in effect.
- (e) <u>Assignment and Assumption of Liabilities</u>. Buyer shall have executed and delivered to Sellers an instrument of assignment and assumption of liabilities with respect to the Assumed Liabilities reasonably satisfactory in form and substance to counsel for Sellers.
- (f) <u>No Litigation</u>. There shall not be pending or threatened in writing by any Government Authority any suit, action or proceeding, (i) challenging or seeking to restrain, prohibit, alter or materially delay the consummation of any of the transactions contemplated by this Agreement, or (ii) seeking to obtain from Sellers or any of its Affiliates any damages in connection with the transactions contemplated hereby.

- (g) <u>No Violation of Order in Petters Case</u>. The consummation of the transactions contemplated by this Agreement will not result in a violation of any order entered by the U.S. District Court for the District of Minnesota in connection with the case *U.S. v. Thomas Joseph Petters et. al.*, 08-364.
- (h) <u>Limited Liability Company Agreement; Issuance of Equity Consideration</u>. The Limited Liability Company Agreement shall (i) be in form and substance reasonably satisfactory to Sellers and (ii) have been executed and delivered by each of the other members of Buyer. Buyer shall have issued the Equity Consideration to Sellers in accordance with the terms hereof.
- (i) <u>Assignment and Assumption Agreement</u>. Sellers shall have received an executed assignment and assumption agreement, in form reasonably satisfactory to the Sellers, evidencing that the Acquired Intellectual Property has been transferred from Buyer to PLR IP Holdings.

SECTION 7.2 Conditions Precedent to the Performance by Buyer.

The obligations of Buyer to consummate the transactions contemplated by this Agreement are subject to the fulfillment, at or before the Closing, of the following conditions, any one or more of which (other than the conditions contained in Section 7.2(c)) may be waived by Buyer, in its sole discretion:

- (a) <u>Representations and Warranties of Sellers</u>. The representations and warranties of Sellers made in <u>Section 4.1</u> of this Agreement shall be true and correct as of the Execution Date and in all material respects as of the Closing Date as though made by Sellers as of the Closing Date, except to the extent such representations and warranties expressly relate to an earlier date, in which case such representations and warranties shall be true and correct on and as of such earlier date.
- (b) <u>Performance of the Obligations of Sellers</u>. Sellers shall have (i) performed in all material respects all obligations required under this Agreement or any Ancillary Agreement to which Sellers are party which are to be performed by Sellers on or before the Closing Date (except with respect to any obligations qualified by materiality, which obligations shall be performed in all respects as required under this Agreement), (ii) taken all actions in connection with the consummation of the transactions contemplated hereby, and (iii) delivered all certificates, instruments, and other documents required herein or otherwise in a form and substance reasonably satisfactory to Buyer required to effect the transactions contemplated hereby.
- (c) Governmental Consents and Approvals. If applicable, the waiting period applicable to the consummation of the transactions contemplated by this Agreement under the HSR Act shall have expired or been terminated. The Orders shall have been entered and shall not have been stayed. Notwithstanding the foregoing, nothing in this Agreement shall preclude Buyer from consummating the transactions contemplated herein if Buyer, in its sole discretion, waives the requirement that the Sale Order shall have become a Final Order. No notice of such waiver of this condition or any other condition to the Closing need be given except to Sellers, it being the intention of the parties that Buyer shall be entitled to, and is not waiving, the protection of Section 363(m) of the Bankruptcy Code, the mootness doctrine and any similar statute or body of law if the Closing occurs in the absence of the Bankruptcy Sale Order becoming a Final Order.
- (d) <u>No Violation of Orders</u>. No preliminary or permanent injunction or other order of any court or Government Authority that declares this Agreement invalid in any material respect or prevents the consummation of the transactions contemplated hereby shall be in effect.

- (e) <u>No Litigation</u>. There shall not be pending or threatened in writing by any Government Authority any suit, action or proceeding, (i) challenging or seeking to restrain, prohibit, alter or materially delay the consummation of any of the transactions contemplated by this Agreement or (ii) seeking to obtain from Buyer or any of its Affiliates any damages in connection with the transactions contemplated hereby.
- (f) <u>Third Party Consents and Approvals</u>. Those Consents identified on <u>Schedule 7.2(f)</u> as being material consents required as closing conditions shall have been obtained.
- (g) <u>Name Change</u>. Sellers shall have provided Buyer written confirmation acceptable to Buyer evidencing Sellers' compliance with Section 5.4(a).
- (h) <u>Bankruptcy Sale Order</u>. The Bankruptcy Sale Order shall be entered and approved by the Bankruptcy Court as contemplated by <u>Section 5.5(c)</u> in a form acceptable to Buyer in writing no later than April 14, 2009 and at such time shall not have been appealed, and is otherwise final and non-appealable, <u>provided</u>, <u>however</u> that Buyer may, in its sole discretion, waive the requirement that the Bankruptcy Sale Order has not been appealed, and is otherwise final and non-appealable.

(i) [Intentionally omitted]

- (j) <u>No Violation of Order in Petters Case.</u> The consummation of the transactions contemplated by this Agreement will not result in a violation of any order entered by the U.S. District Court for the District of Minnesota in connection with the case *U.S. v. Thomas Joseph Petters et. al.*, 08-364
- (k) <u>Transferred Internet Domain Names</u>. Sellers shall have provided Buyer a certificate signed by a duly authorized officer of Sellers and acceptable to Buyer evidencing that each of the Transferred Internet Domain Names (other than the Subject Internet Domain Names) registered in the name of any Affiliate of Sellers has been transferred to a Seller free and clear of all Liens (other than Permitted Liens), Claims and Encumbrances (it being understood that Sellers shall not be required to take any action to render any Seller the registered owner of such Transferred Internet Domain Names, which shall be Buyer's sole responsibility (at its sole cost and expense) post-Closing, subject to all such Transferred Internet Domain Names (other than the Subject Internet Domain Names) being transferred to Buyer at the Closing on the terms set forth herein).

ARTICLE 8

TERMINATION

SECTION 8.1 Conditions of Termination.

This Agreement may be terminated only in accordance with this <u>Section 8.1</u>. This Agreement may be terminated at any time before the Closing, as follows:

- (a) By mutual written consent of Sellers' Representative (on behalf of Sellers) and Buyer;
- (b) By Buyer, by written notice to Sellers' Representative, on or after May 14, 2009 (the "<u>Termination Date</u>"), subject, however, to extension by the mutual written consent of Sellers' Representative and Buyer, if the Closing shall not have occurred on or prior to the Termination Date; <u>provided, however</u>, that (i) if the condition to the Closing set forth in the first sentence of <u>Sections 7.1(c)</u>

and <u>7.2(c)</u> shall not have been satisfied by the Termination Date (and such condition is required to be satisfied in order to effect the Closing), but all other conditions to Closing set forth in <u>Article 7</u> hereof would be satisfied or waived if the Closing Date were to occur on such date, then the Termination Date shall be automatically extended to June 14, 2009, and (ii) Buyer shall not have the right to terminate this <u>Agreement</u> under this <u>Section 8.1(b)</u> if Buyer is then in material breach of this Agreement;

- (c) By Sellers' Representative (on behalf of Sellers), by written notice to Buyer, on or after the Termination Date, subject, however, to extension by the mutual written consent of Buyer and Sellers' Representative, if the Closing shall not have occurred on or prior to the Termination Date; provided, however, that (i) if the condition to the Closing set forth in the first sentence of Sections 7.1(c) and 7.2(c) shall not have been satisfied by the Termination Date (and such condition is required to be satisfied in order to effect the Closing), but all other conditions to Closing set forth in Article 7 hereof would be satisfied or waived if the Closing Date were to occur on such date, then the Termination Date shall be automatically extended to June 14, 2009, and (ii) Sellers' Representative shall not have the right to terminate this Agreement under this Section 8.1(c) if Sellers are then in material breach of this Agreement:
- (d) By Buyer, by written notice to Sellers' Representative, if Buyer has previously provided Sellers' Representative with notice of any inaccuracy of any representation or warranty of Sellers contained in Section 4.1, which inaccuracy would reasonably be expected to result in, individually or in the aggregate with the results of other inaccuracies, a Material Adverse Effect, or notice of a material failure to perform any covenant of Sellers contained in this Agreement or any Ancillary Agreement to which a Seller is party (including, without limitation, the failure to deliver and/or convey to Buyer any of the Acquired Assets free and clear of any and all interests, claims, encumbrances and/or liens), and Sellers have failed, within five (5) Business Days after such notice, to remedy such inaccuracy or perform such covenant or provide reasonably adequate assurance to Buyer of Sellers' ability to remedy such inaccuracy or perform such covenant; provided, however, that Buyer shall not have the right to terminate this Agreement under this Section 8.1(d) if Buyer is then in material breach of this Agreement;
- (e) By Sellers' Representative (on behalf of Sellers), by written notice to Buyer, if Sellers' Representative has previously provided Buyer with notice of any inaccuracy of any representation or warranty of Buyer contained in Section 4.2, which inaccuracy would reasonably be expected to result in, individually or in the aggregate with the results of other inaccuracies, the conditions set forth in Section 7.1 not being satisfied, or notice of a material failure to perform any covenant of Buyer contained in this Agreement or any Ancillary Agreement to which Buyer is party, and Buyer has failed, within five (5) Business Days after such notice, to remedy such inaccuracy or perform such covenant or provide reasonably adequate assurance to Sellers of Buyer's ability to remedy such inaccuracy or perform such covenant; provided, however, that Sellers' Representative shall not have the right to terminate this Agreement under this Section 8.1(e) if Sellers are then in material breach of this Agreement;
- (f) Buyer may terminate this Agreement by giving written notice to Sellers' Representative at any time prior to Closing in the event Sellers have accepted or selected and the Bankruptcy Court shall have approved by Final Order, the bid or bids (including a credit bid) of any Person or Persons other than Buyer or any of its Affiliates to purchase all or a significant portion of the businesses and assets of Sellers; or
- (g) Immediately and automatically (without any further action of Buyer or Sellers), if the Bankruptcy Court shall enter a Final Order approving the sale of the Acquired Assets (or any material portion of the Acquired Assets) to any Person other than the Buyer.

SECTION 8.2 Effect of Termination.

In the event of termination pursuant to <u>Section 8.1</u>, this Agreement shall become null and void and have no effect (other than those provisions of <u>Article 8</u>, <u>Article 10</u> and <u>Article 11</u> that expressly survive termination), and none of Sellers or Buyer, or their respective Affiliates or respective representatives, shall have any liability whatsoever with respect to this Agreement or any Ancillary Agreement. Accordingly, Buyer shall have no recourse whatsoever against Sellers or Sellers' Affiliates following any termination of this Agreement.

ARTICLE 9

SURVIVAL

SECTION 9.1 Survival.

Each of the representations and warranties, covenants and agreements of Sellers and Buyer made in this Agreement shall not survive the Closing Date; provided, however, that any covenant or agreement in this Agreement which, by its terms, is to survive the Closing Date, shall survive the Closing Date for the duration of such covenant or agreement.

SECTION 9.2 Specific Performance.

Sellers, on the one hand, and Buyer, on the other hand, each acknowledges that in case of any breach of their covenants or other obligations, the other would suffer immediate and irreparable harm, which money damages would be inadequate to remedy, and accordingly, in case of any such breach each non-breaching party shall be entitled to obtain specific performance and other equitable remedies, in addition to other remedies provided in this <u>Article 9</u>.

SECTION 9.3 Covenant Not to Sue.

- (a) On and after the Closing Date, Buyer covenants and agrees not to sue or otherwise bring any action against the Seller Group, any of the current directors and officers of the Seller Group or any of the current and former employees, agents, managers, advisors, attorneys and representatives (in their capacity as such and in no other capacity) of the Seller Group (collectively, the "Seller Group Representatives"), with respect to any and all Claims based in whole or in part upon any act, omission, transaction, event or other occurrence taking place at any time on or before the Closing Date, with the exception of (i) acts, omissions, transactions, events or occurrences resulting from or involving the gross negligence, breach of fiduciary duties, ultra vires acts or fraud of any such Persons, as determined by a final order of the Bankruptcy Court or other court of competent jurisdiction or (ii) cross claims by Buyer resulting from third party claims against Buyer for (A) any acts or omissions of the Seller Group or the Seller Group Representatives, or (B) any business, assets and/or liabilities of Seller not purchased or assumed by Buyer.
- (b) On and after the Closing Date, the Seller Group covenants and agrees not to sue or otherwise bring any action against any non-debtor parties to Acquired Contracts under which Acquired Intellectual Property is licensed to or by Buyer; <u>provided</u>, <u>however</u>, that notwithstanding anything contained herein to the contrary, this <u>Section 9.3(b)</u> shall not apply to, and the Seller Group shall be entitled to sue or otherwise bring an action against (which shall include the right to assert defenses, setoffs

and counterclaims against and object to claims filed, held or asserted by), any Person in connection with any matter set forth on Schedule 1.1(o) or that otherwise constitutes an Excluded Asset.

(c) Notwithstanding any other term in this Agreement to the contrary, the waivers, covenants and agreements contained in this <u>Section 9.3</u> shall survive the Closing and shall bind and inure to the benefit of, as the case may be, the Buyer and its successors and assigns and the Seller Group, their Affiliates, and their estate, creditors, successors and assigns, including, without limitation, any trustee in any case under Chapter 7 of the Bankruptcy Code.

ARTICLE 10

MISCELLANEOUS

SECTION 10.1 Joint Drafting.

The parties have participated jointly in the negotiation and drafting of this Agreement. In the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the parties and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any of the provisions of this Agreement.

SECTION 10.2 Further Assurances.

At the request and the sole expense of the requesting party, Buyer or Sellers, as applicable, shall execute and deliver, or cause to be executed and delivered, such documents as Buyer or Sellers, as applicable, or their respective counsel may reasonably request to effectuate the purposes of this Agreement and the Ancillary Agreements.

SECTION 10.3 Successors and Assigns.

This Agreement shall be binding on and inure to the benefit of Buyer and Sellers and their respective successors and permitted assigns, including, without limitation, any trustee appointed in the Bankruptcy Cases or subsequent Chapter 7 case. No party may assign either this Agreement or any of its rights, interests or obligations hereunder without the prior written consent of the other party.

SECTION 10.4 Governing Law; Jurisdiction.

This Agreement shall be construed, performed and enforced in accordance with, and governed by, the laws of the State of Minnesota (without giving effect to the principles of conflicts of laws thereof), except to the extent that the laws of such State are superseded by the Bankruptcy Code or other applicable federal law. For so long as Sellers are subject to the jurisdiction of the Bankruptcy Court, the parties irrevocably elect, as the sole judicial forum for the adjudication of any matters arising under or in connection with the Agreement, and consent to the exclusive jurisdiction of, the Bankruptcy Court. After Sellers are no longer subject to the jurisdiction of the Bankruptcy Court, the parties irrevocably elect, as the sole judicial forum for the adjudication of any matters arising under or in connection with this Agreement, and consent to the jurisdiction of, any state or federal court having jurisdiction over Hennepin County, Minnesota.

SECTION 10.5 Expenses.

Except as set forth in this <u>Section 10.5</u>, each of the parties will bear its own costs and expenses (including legal fees and expenses) incurred in connection with this Agreement and the transactions

contemplated hereby. Without limiting the generality of the foregoing, all transfer, documentary, sales, use, stamp, registration and other such Taxes, and all conveyance fees, recording charges and other fees and charges (including any penalties and interest) incurred in connection with the consummation of the transactions contemplated by this Agreement shall be paid by Buyer when due, and Buyer shall, at its own expense, file all necessary Tax Returns and other documentation with respect to all such Taxes, fees and charges, and, if required by applicable law, the parties will, and will cause their Affiliates to, join in the execution of any such Tax Returns and other documentation. Buyer shall be liable for and pay all Taxes applicable to the Acquired Assets and Assumed Liabilities that are attributable to taxable years or periods beginning on the Closing Date and with respect to any Straddle Period, the portion of such Straddle Period beginning on the Closing Date.

SECTION 10.6 Severability.

In the event that any part of this Agreement is declared by any court or other judicial or administrative body to be null, void or unenforceable, said provision shall survive to the extent it is not so declared, and all of the other provisions of this Agreement shall remain in full force and effect only if, after excluding the portion deemed to be unenforceable, the remaining terms shall provide for the consummation of the transactions contemplated hereby in substantially the same manner as originally set forth at the later of (a) the Execution Date and (b) the date this Agreement was last amended.

SECTION 10.7 Notices.

(a) All notices, requests, demands, consents and other communications under this Agreement shall be in writing and shall be deemed to have been duly given: (i) on the date of service, if served personally on the party to whom notice is to be given; (ii) on the day of transmission, if sent via facsimile transmission to the facsimile number given below; (iii) on the day after delivery to Federal Express or similar overnight courier or the Express Mail service maintained by the United States Postal Service addressed to the party to whom notice is to be given; or (iv) on the fifth day after mailing, if mailed to the party to whom notice is to be given, by first class mail, registered or certified, postage prepaid and properly addressed, to the party as follows:

If to a Seller or:

c/o Polaroid Holding Company 4400 Baker Road Minnetonka, MN 55343 Attn.: Mary L. Jeffries Fax: (952) 351-0330

With a copy (which shall not constitute notice) to:

Lindquist & Vennum PLLP 4200 IDS Center 80 South Eighth Street Minneapolis, MN 55402 Attn.: Charles P. Moorse, Esq. George H. Singer, Esq.

Fax: (612) 371-3207

If to Buyer:

PLR Acquisition, LLC

c/o Gordon Brothers Brands, LLC 101 Huntington Avenue, 10th Floor Boston, MA 02199 Attn: Rafael Klotz

Fax: (617) 531-7929

PLR Holdings, LLC c/o Hilco Consumer Capital, L.P. 5 Revere Drive, Suite 206 Northbrook, IL 60062 Attn: Eric Kaup

Fax: (847) 897-0766

With a copy (which shall not constitute notice) to:

Dorsey & Whitney LLP 50 South Sixth Street Minneapolis, MN 55402

Attn.: Mark J. Kalla, Esq.

Christopher J. Bellini, Esq.

Fax: (612) 340-2868

(b) Any party may change its address or facsimile number for the purpose of this Section 10.7 by giving the other parties written notice of its new address in the manner set forth above.

SECTION 10.8 Amendments; Waivers.

This Agreement may be amended or modified, and any of the terms, covenants, representations, warranties or conditions hereof may be waived, only by a written instrument executed by Buyer and Sellers' Representative, or in the case of a waiver, by the party waiving compliance. Any waiver by any party of any condition, or of the breach of any provision, term, covenant, representation or warranty contained in this Agreement, in any one or more instances, shall not be deemed to be or construed as a furthering or continuing waiver of any such condition, or of the breach of any other provision, term, covenant, representation or warranty of this Agreement.

SECTION 10.9 Public Announcements.

Sellers, on the one hand, and Buyer, on the other hand, shall not make any press release or public announcement concerning the transactions contemplated by this Agreement without the prior written agreement from Sellers' Representative (in the case of Sellers) or Buyer, as applicable, unless a press release or public announcement is required by law, the rules of any stock exchange or order of the Bankruptcy Court. If any such announcement or other disclosure is required by law, the rules of any stock exchange or order of the Bankruptcy Court, the form and content of any such announcing or other disclosure shall be subject to the prior written consent of Sellers' Representative (in the case of Sellers) or Buyer, as applicable, which consent shall not be unreasonably withheld.

SECTION 10.10 Entire Agreement.

This Agreement and the Ancillary Agreements contain the entire understanding among the parties with respect to the transactions contemplated hereby and supersede and replace all prior and contemporaneous agreements and understandings, oral or written, with regard to such transactions. All Schedules hereto and any documents and instruments delivered pursuant to any provision hereof are expressly made a part of this Agreement as fully as though completely set forth herein.

SECTION 10.11 No Third Party Beneficiaries.

Except as set forth in <u>Sections 5.10</u> and <u>9.3</u>, nothing in this Agreement is intended to or shall confer any rights or remedies under or by reason of this Agreement on any Persons other than Sellers and Buyer and their respective successors and permitted assigns. Nothing in this Agreement is intended to or shall relieve or discharge the obligation or liability of any third Persons to Sellers or to Buyer. Except as set forth in <u>Sections 5.10</u> and <u>9.3</u>, this Agreement is not intended and shall not give any third Persons any right of subrogation or action over or against Sellers or against Buyer.

SECTION 10.12 Headings.

The article and section headings in this Agreement are for reference purposes only and shall not affect the meaning or interpretation of this Agreement.

SECTION 10.13 Counterparts.

This Agreement may be executed in counterparts, each of which shall be deemed an original, but all of which shall constitute the same instrument.

SECTION 10.14 Construction.

Any reference to any federal, state, local or foreign statute or law shall be deemed also to refer to all rules and regulations promulgated thereunder, unless the context requires otherwise. The word "including" shall mean including without limitation. Any reference to the singular in this Agreement shall also include the plural and vice versa.

SECTION 10.15 Tax Disclosure.

Notwithstanding anything herein to the contrary, each party (and each Affiliate and Person acting on behalf of any such party) agrees that each party (and each employee, representative, and other agent of such party) may disclose to any and all Persons, without limitation of any kind, the Tax treatment and Tax structure of the transaction and all materials of any kind (including opinions or other tax analyses) that are provided to such party or such Person relating to such Tax treatment and Tax structure, but only to the extent necessary to comply with any applicable federal or state securities laws or tax laws. This authorization is not intended to permit disclosure of any other information including (without limitation) (a) any portion of any materials to the extent not related to the Tax treatment or Tax structure of the transaction, (b) the identities of participants or potential participants in the transaction, (c) the existence or status of any negotiations, (d) any pricing or financial information (except to the extent such pricing or financial information is related to the Tax treatment or Tax structure of the transaction), or (e) any other term or detail not relevant to the Tax treatment or the Tax structure of the transaction.

SECTION 10.16 Sellers' Representative.

Each Seller hereby irrevocably appoints PHC to act as representative, agent, proxy and attorneyin-fact for all Sellers for all purposes under this Agreement (PHC, in such capacity, being "Sellers' Representative"), including, without limitation, the full power and authority on each Seller's behalf to: (i) receive notices or service of process; (ii) negotiate, determine, compromise, settle and take any other action permitted or called for by Sellers under this Agreement; and (iii) to execute and deliver any termination of, amendment to or waiver under this Agreement. Each Seller agrees that such agency and proxy are coupled with an interest and are, therefore, irrevocable without the consent of Sellers' Representative and shall survive the bankruptcy, dissolution or liquidation of any Seller. All decisions and actions by Sellers' Representative shall be binding upon all Sellers, and no Seller shall have the right to object, dissent, protest or otherwise contest same. Sellers' Representative shall have no duties or obligations hereunder except those specifically set forth herein and such duties and obligations shall be determined solely by the express provisions of this Agreement. Each Seller agrees to indemnify and hold harmless Sellers' Representative from and against all expenses (including reasonable attorneys' fees), judgments, fines and amounts incurred by Sellers' Representative in connection with any action, suit or proceeding to which Sellers' Representative is made a party by reason of the fact it is or was acting as a Sellers' Representative under this Agreement. Neither Sellers' Representative nor any agent employed by Sellers' Representative shall incur any liability to any Seller relating to the performance of its duties hereunder except for actions or omissions constituting fraud or bad faith. Sellers' Representative shall have no liability in respect of any action, claim or proceeding brought against Sellers' Representative by any Seller if Sellers' Representative took or omitted taking any action in good faith.

ARTICLE 11

DEFINITIONS

As used in this Agreement, the following terms have the following meanings:

- "Accounts Receivable" has the meaning specified in Section 1.1(q).
- "Acquired Assets" has the meaning set forth in Section 1.1.
- "Acquired Contracts" has the meaning set forth in Section 1.1(c).
- "Acquired Equity Interests" has the meaning set forth in Section 1.1(m).
- "Acquired Intellectual Property" has the meaning set forth in Section 1.1(f).
- "Acquired IP Causes of Action" has the meaning set forth in Section 1.1(o).
- "<u>Acquired Subsidiaries</u>" means those entities in which a Seller holds an ownership interest that are listed on <u>Schedule 11(a)</u>. For the avoidance of doubt, the Acquired Subsidiaries shall not include any Seller, even if such Seller is a direct or indirect subsidiary of another Seller.
 - "Affiliate" has the meaning given that term in Section 101(2) of the Bankruptcy Code.
 - "Agreement" has the meaning set forth in the Preamble.
 - "Allocation" has the meaning set forth in Section 6.3.

- "Ancillary Agreement" means any other agreement that a Seller or Buyer, as applicable, may reasonably enter into in connection with the consummation of the transactions contemplated hereby.
 - "Apportionment Notice" has the meaning set forth in Section 2.1.
 - "Assumed Liabilities" has the meaning set forth in Section 1.3.
 - "Auction" has the meaning set forth in Section 5.5(c)(ii).
- "Avoidance Actions" means all Claims, rights and remedies of a debtor assertable or arising under Chapter 5 of the Bankruptcy Code or any other applicable law, including, without limitation, all preference, fraudulent transfer, and other Claims to avoid a transfer.
 - "Bankruptcy Assignment Motion" has the meaning set forth in Section 5.5(d).
 - "Bankruptcy Assignment Order" has the meaning set forth in Section 5.5(d).
 - "Bankruptcy Bidding Procedures Motion" has the meaning set forth in Section 5.5(c)(i).
 - "Bankruptcy Bidding Procedures Order" has the meaning set forth in Section 5.5(b).
 - "Bankruptcy Cases" has the meaning set forth in the Recitals.
 - "Bankruptcy Code" has the meaning set forth in the Recitals.
 - "Bankruptcy Court" has the meaning set forth in Section 5.5(a).
 - "Bankruptcy Rules" has the meaning set forth in the Recitals.
 - "Bankruptcy Sale Motion" has the meaning set forth in Section 5.5(a).
 - "Bankruptcy Sale Order" has the meaning set forth in Section 5.5(c).
 - "Bulk Sale Offer" has the meaning set forth in Section 5.10(a).
 - "Business" has the meaning set forth in the Recitals.
- "Business Day" means any day other than Saturday, Sunday and any day that is a legal holiday or a day on which banking institutions in Minneapolis, Minnesota are authorized by law or other governmental action to close.
 - "Buyer" has the meaning set forth in the Preamble.
 - "Buyer Group" means Buyer and its Affiliates.
 - "Cash" means all cash and cash equivalents.
- "Causes of Action" means all Claims and causes of action (of any kind or character and whether arising prior to, on or after the Petition Date), including, without limitation, the Avoidance Actions, that Sellers or any of their Affiliates may have against any other Person.

"Claim" has the meaning given that term in Section 101(5) of the Bankruptcy Code and includes, *inter alia*, all rights, claims, causes of action, defenses, debts, demands, damages, offset rights, setoff rights, recoupment rights, obligations, and liabilities of any kind or nature under contract, at law or in equity, known or unknown, contingent or matured, liquidated or unliquidated, and all rights and remedies with respect thereto.

"Closing" has the meaning set forth in Section 3.1.

"Closing Cash Consideration" has the meaning set forth in Section 2.1(a).

"Closing Date" has the meaning set forth in Section 3.1.

"Code" means the Internal Revenue Code of 1986, as amended.

"Confidential Information" means any information concerning the operations and affairs of the Seller Group or Buyer Group, as applicable, their respective Affiliates and/or their respective businesses that is not already generally available to the public.

"Consents" has the meaning set forth in Section 4.1(g).

"Contract" means any written contract, agreement, lease or sublease, license or sublicense, instrument, indenture, commitment or undertaking.

"Contract Assumption Notice" has the meaning set forth in Section 1.5(b).

"Definitively Acquired Contracts" means the Contracts identified on Schedule 11(b).

"<u>Definitively Excluded Contracts</u>" means any and all Contracts (i) between any Seller or their Affiliates, on the one hand, and any of the following entities on the other hand, (A) Target Chip Ganassi Racing or (B) ArcSoft, or (ii) relating to (I) Polaroid ID Systems, an operating division of PC or (II) the 20" x 24" studio photography business owned by PC, including all of the Contracts set forth on <u>Schedule 11(c)</u>.

"Deposit" has the meaning set forth in Section 2.2.

"Employee Benefit Plans" means all employee benefit plans as defined in Section 3(3) of ERISA, all compensation, pay, severance pay, salary continuation, bonus, incentive, stock option, retirement, pension, profit sharing or deferred compensation plans, Contracts, programs, funds or arrangements of any kind and all other employee benefit plans, programs, funds or arrangements (whether written or oral, qualified or nonqualified, funded or unfunded, foreign or domestic, currently effective or terminated, and whether or not subject to ERISA) and any trust, escrow or similar agreement related thereto, whether or not funded.

"Encumbrance" means any lien, pledge, charge, security interest, option, right of first refusal, mortgage, easement, right of way, lease, sublease, license, sublicense, adverse claim, title defect, encroachment, other survey defect, or other encumbrance of any kind, including, with respect to real property, any covenant or restriction relating thereto.

"Environmental Laws" means all applicable federal, state and local statutes, ordinances, rules, orders, judgments, junctions, decrees, regulations and other provisions having the force of law, all judicial and administrative orders and determinations, and all common law concerning pollution or protection of

human health and the environment, including, without limitation, all those relating to the presence, use, production, generation, handling, transportation, treatment, storage, disposal, distribution, labeling, testing, processing, discharge, release, threatened release, control or cleanup of any Hazardous Materials.

"<u>Equity Consideration</u>" means 25% of the fully-diluted equity of PLR IP Holdings, LLC, a Delaware limited liability company and a wholly owned subsidiary of Buyer, in all respects governed by the terms of the Limited Liability Company Agreement.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"ERISA Affiliate" means, with respect to any Person, any trade or business (whether or not incorporated) (i) under common control within the meaning of Section 4001(b)(1) of ERISA with such Person or (ii) which together with such Person is treated as a single employer under Sections 414(b), (c), (m), (n) or (o) of the Code.

"Excluded Assets" has the meaning set forth in Section 1.2.

"Excluded Computer-Related Assets" means any and all computer hardware, related equipment and software (i) currently used in and/or reasonably necessary to operate the Data Center and/or (ii) reasonably necessary for winding down the Sellers' bankruptcy estates, including without limitation, all computer hardware and related equipment and software listed on Schedule 1.2(e).

"Excluded Contracts" has the meaning set forth in Section 1.2(c).

"Excluded Liabilities" has the meaning set forth in Section 1.4.

"Excluded Receivables" means all accounts and notes receivable of Sellers relating to Circuit City Stores, Inc.

"Execution Date" has the meaning set forth in the Preamble.

"<u>Final Order</u>" means an order or judgment of the Bankruptcy Court, or other court of competent jurisdiction, the implementation, operation or effect of which has not been stayed and as to which order (or any revision, modification or amendment thereof) the time to appeal or seek review, rehearing or writ of certiorari has expired and as to which no appeal or petition for review, reconsideration, rehearing or certiorari has been taken and is pending.

"Government Authority" means any agency, division, subdivision or governmental or regulatory authority or any adjudicatory body thereof, of the United States, or any state thereof.

"Hazardous Materials" means any hazardous or toxic substance or waste or any contaminant or pollutant regulated or otherwise creating liability under Environmental Laws, including, without limitation, "hazardous substances" as defined by the Comprehensive Environmental Response Compensation and Liability Act, as amended, "toxic substance" as defined by the Toxic Substance Control Act, as amended, "hazardous wastes" as defined by the Resource Conservation and Recovery Act, as amended, "hazardous materials" as defined by the Hazardous Materials Transportation Act, as amended, thermal discharges, radioactive substances, PCBs, natural gas, petroleum products or byproducts and crude oil.

"HSR Act" means the Hart-Scott-Rodino Anti-trust Improvements Act of 1976, as amended.

"Improvements" means the buildings, improvements and structures of Sellers now existing on the Real Property or demised under any lease of, or other Contract for the use of, real property, and any and all fixtures appurtenant thereto.

"Intellectual Property" shall mean (i) patents and applications therefor, including continuations, divisionals, continuations-in-part, or reissue patent applications and patents issuing thereon, invention and patent disclosures, inventions, improvements and trade secrets, (ii) registered and unregistered trademarks, service marks, trade names, service names, brand names, derivative brands, fictional or assumed names, all trade dress rights, logos, slogans, uniform resource locators, Transferred Internet Domain Names and corporate names, together with the goodwill associated with the foregoing, and applications and registrations therefor and renewals thereof (iii) all copyrights and registrations and applications therefor and works of authorship and mask work rights, (iv) Technology, (v) brand related property and advertising materials, copy, commercials, images, package designs, product designs and artwork, (vi) records and documents, whether in hard copy or electronic, relating to any of the foregoing, (vii) any and all customer lists and contact information, including email addresses and (viii) any other proprietary rights used by Sellers in connection with the operation of the Business, as historically operated by Sellers. The definition of Intellectual Property includes all such assets on a worldwide basis and all foreign equivalent or counterpart rights having similar effect in any jurisdiction throughout the world.

"Interim Reports" has the meaning set forth in Section 4.1(d).

"Inventory" means all the finished goods, raw materials, work in process and other supplies owned by Sellers on the Closing Date.

"Inventory Locations" has the meaning set forth in Section 1.1(d).

"Inventory Taking" has the meaning set forth in Section 2.1(c).

"Lien" has the meaning given to that term in Section 101(37) of the Bankruptcy Code.

"<u>Limited Liability Company Agreement</u>" means the Limited Liability Company Agreement of Buyer, substantially in the form attached hereto as <u>Exhibit A</u>.

"Material Adverse Effect" means a state of facts, event, change or effect with respect to the Acquired Assets or the Assumed Liabilities, that has had, or is reasonably likely to result in, a material adverse effect on the value of the Acquired Assets, a material increase in the amount of the Assumed Liabilities or any material limitation on the rights to use the Acquired Intellectual Property, but excludes (a) any state of facts, event, change or effect caused by events, changes or developments relating to (i) general business or economic conditions, including such conditions related to the Business or manufacturing industries generally, (ii) national or international political or social conditions, including the engagement by the United States in hostilities, whether or not pursuant to the declaration of a national emergency or war, or the occurrence of any military or terrorist attack upon the United States, or any of its territories, possessions, or diplomatic or consular offices or upon any military installation, equipment or personnel of the United States, except to the extent that any such conditions or acts have a material adverse effect on (x) the physical condition of the Acquired Assets, taken as a whole, or (y) the ability of the Acquired Assets, taken as a whole, to be physically used in a manner consistent with past-practice, (iii) financial, banking, or securities markets (including any disruption thereof and any decline in the price of any security or any market index), (iv) changes in United States generally accepted accounting principles, (v) changes in law, rules, regulations, orders, or other binding directives issued by any governmental entity and applicable generally to the Business, or (vi) the taking of any action

contemplated by this Agreement or the Ancillary Agreements contemplated hereby (including, without limitation, the filing of the Petition of Relief and the Bankruptcy Cases), (b) any existing event, occurrence, or circumstance with respect to which the Buyer has knowledge as of the Execution Date, and (c) any adverse change in or effect on the business of the Seller Group that is cured by Sellers before the earlier of (i) the Closing Date or (ii) the date on which this Agreement is terminated pursuant to Section 8.1 hereof.

"Membership Interests" has the meaning set forth in Section 4.2(h).

"NPKK Channels" has the meaning specified in Section 5.10(b).

"Offered Employees" has the meaning set forth in Section 5.6(b).

"Orders" means the Bankruptcy Sale Order, the Bankruptcy Bidding Procedures Order, and the Bankruptcy Assignment Order.

"Owned Machinery and Equipment" has the meaning set forth in Section 1.1(b).

"PAP" has the meaning set forth in the Preamble.

"PC" has the meaning set forth in the Preamble.

"PCAP" has the meaning set forth in the Preamble.

"PCE" has the meaning set forth in the Preamble.

"Permits" has the meaning set forth in Section 1.1(h).

"Permitted Liens" mean: (a) Liens for Taxes, assessments or other similar charges that are not yet due and payable; (b) easements, licenses, unrecorded real estate agreements, restrictions and other matters of record which either (i) the title company has agreed to affirmatively insure against loss caused thereby in the applicable title policy, by way of ALTA coverage or other affirmative cover, reasonably acceptable to Buyer, or (ii) do not materially and adversely affect the operation of the Real Property in question as currently and previously used in the operation of the Seller Group's business and (c) any state of facts a survey or other visual inspection would show that do not materially and adversely affect the operation of the Real Property in question as currently and previously used in the operation of the Seller Group's business; and (d) Liens arising from the Assumed Liabilities.

"<u>Person</u>" means any individual, corporation, partnership, limited liability company, joint venture, association, joint-stock company, trust, unincorporated organization or Government Authority.

"Petition Date" means the date of commencement of the Bankruptcy Cases.

"PGW" means Petters Group Worldwide, LLC, a Delaware limited liability company.

"PHC" has the meaning set forth in the Preamble.

"PINT" has the meaning set forth in the Preamble.

"PLA" has the meaning set forth in the Preamble.

"PLR IP Holdings" has the meaning set forth in Section 4.2(i).

- "PNB" has the meaning set forth in the Preamble.
- "PNOR" has the meaning set forth in the Preamble.
- "<u>Pogo Inventory</u>" means all Inventory associated with the Polaroid PoGoTM Instant Mobile Printer.
 - "PWALT" has the meaning set forth in the Preamble.
- "Polaroid Name" means any name including the word "Polaroid" and any other trade names, trademarks or service marks owned by Sellers or licensed to Sellers.
 - "Purchase Price" has the meaning set forth in Section 2.1.
 - "Real Property" has the meaning set forth in Section 1.1(a).
- "Receivables Sharing Portion" means an amount equal to Eighty Five Percent (85%) of the gross proceeds of Accounts Receivable which are actually collected in cash by Buyer.
 - "Rejection Notice" has the meaning set forth in Section 1.5(b).
 - "ROFR Offer" has the meaning set forth in Section 5.10(a).
 - "Seller Benefit Plans" has the meaning set forth in Section 1.2(d).
 - "Seller Controlled Group" has the meaning set forth in Section 1.2(d).
 - "Seller Group" means Sellers and the Acquired Subsidiaries.
 - "Seller Group Representatives" has the meaning set forth in Section 9.3(a).
 - "Seller Reports" has the meaning set forth in Section 4.1(d).
 - "Sellers" has the meaning set forth in the Preamble.
 - "Sellers' Representative" has the meaning set forth in Section 10.16.
 - "Specified Entities" means the entities set forth on Exhibit D.
- "Straddle Period" shall mean any taxable period beginning on, or before and ending after the Closing Date.
 - "Subject Affiliates" has the meaning set forth in Section 5.10.
 - "Subject Internet Domain Names" has the meaning set forth in Section 5.1(e).
 - "Subject Inventory" has the meaning set forth in Section 1.2(s).
- "Supplies" means all supplies, items and materials (including spare parts) owned by Sellers on the Closing Date.
 - "Supplies Locations" has the meaning set forth in Section 1.1(e).

"Tax Records" has the meaning set forth in Section 1.2(m).

"<u>Tax Return</u>" means any report, declaration, return, information return, filing, claim for refund or other information relating to Taxes, including any Schedules, exhibits or attachments thereto, and any amendments to any of the foregoing required to be filed or maintained in connection with the calculation, determination, assessment or collection of any Taxes (including estimated Taxes).

"Taxes" means all taxes, however denominated, including any interest, penalties or additions to tax that may become payable in respect thereof, imposed by any Government Authority, whether payable by reason of contract, assumption, transferee liability, operation of law or Treasury Regulation Section 1.1502-6(a) (or any predecessor or successor thereof or any analogous or similar provision under state, local or foreign law), which taxes shall include all income taxes, payroll and employee withholding, unemployment insurance, social security (or similar), sales and use, excise, franchise, gross receipts, occupation, real and personal property, stamp, transfer, workmen's compensation, customs duties, registration, documentary, value added, alternative or add-on minimum, estimated, environmental (including taxes under Section 59A of the Code) and other assessments or obligations of the same or a similar nature, whether arising before, on or after the Closing Date.

"<u>Technology</u>" means any and all inventions, discoveries, ideas, processes, formulae, designs, models, industrial designs, know-how, confidential information and proprietary information, whether or not patented or patentable, writings and other copyrightable works and works in progress, databases and software.

"Termination Date" has the meaning set forth in Section 8.1(b).

"Territory" has the meaning set forth in Section 5.10(a).

"Transaction Taxes" has the meaning set forth in Section 6.1.

"<u>Transferred Internet Domain Names</u>" means all internet domain names used in connection with the Business that are transferred to Buyer by Sellers following the assignment to Sellers by Sellers' Affiliates pursuant to <u>Section 5.1(e)</u>.

"Transition Period" has the meaning set forth in Section 5.3(d).

"Use Restrictions" has the meaning set forth in Section 5.10(a).

"Year-End Reports" has the meaning set forth in Section 4.1(d).

"Zink Entities" has the meaning set forth in Section 1.1.

* * * * *

IN WITNESS WHEREOF, the parties hereto have caused this Asset Purchase Agreement to be executed by their respective officers thereunto duly authorized as of the Execution Date.

BUYER:

PLR ACQUISITION, LEC

By: Fire W. KAN

Tro: Intelligence of Sections (

SELLERS:

POLAROID HOLDING COMPANY

POLAROID ASIA PACIFIC, LLC

By: Mary L. Jeffries

Its: President and Chief Executive Officer

By: Mary L. Jeffries
Its: Authorized Signatory

POLAROID CORPORATION

POLAROID INTERNATIONAL HOLDING,

LLC

By: Mary L. Jeffries

Its: President and Chief Executive Officer

By: Mary L. Jeffries

Its: Authorized Signatory

POLAROID CONSUMER ELECTRONICS, LLC

POLAROID NORWOOD REAL ESTATE, LLC

By: Mary L. Jeffries

Its: President and Chief Executive Officer

By: Mary L. Jeffries

Its: Authorized Signatory

POLAROID CAPITAL, LLC

POLAROID NEW BEDFORD REAL ESTATE,

LLC

By: Mary L. Jeffries

Its: Authorized Signatory

By: Mary L. Jeffries

Its: Authorized Signatory

POLAROID LATIN AMERICA I

CORPORATION

POLAROID WALTHAM REAL ESTATE, LLC

By: Mary L. Jeffries

Its: Authorized Signatory

By: Mary L. Jeffries Its: Authorized Signatory

IN WITNESS WHEREOF, the parties hereto have caused this Asset Purchase Agreement to be executed by their respective officers thereunto duly authorized as of the Execution Date.

BUYER:	
PLR ACQUISITION, LLC	
By:	
SELLERS:	
POLAROID HOLDING COMPANY	POLAROID ASIA PACIFIC, LLC
By: Mary L. Teffries Its: President and Chief Executive Officer	By: Mary L. Jeffries Its: Authorized Signatory
POLAROID CORPORATION By: Mary L. Jeffries Its: President and Chief Executive Officer	POLAROID INTERNATIONAL HOLDING, LLC By: Mary L. Jeffries Its: Authorized Signatory
POLAROID CONSUMER ELECTRONICS, LLC	POLAROID NORWOOD REAL ESTATE, LLC
By: Mary L Jeffries Its: President and Chief Executive Officer	By: Mary L Jeffries Its: Authorized Signatory
POLAROID CAPITAL, LLC By: Mary L. Jeffries Its: Authorized Signatory	POLAROID NEW BEDFORD REAL ESTATE, LLC By: Mary L. Jeffries Its: Authorized Signatory
POLAROID LATIN AMERICA I CORPORATION By: Mary Lifeffries Its: Authorized Signatory	POLAROID WALTHAM REAL ESTATE, LLC By: Mary L. Jeffries Its: Authorized Signatory

Its: Authorized Signatory

THE SECURITIES REPRESENTED BY AND ISSUED PURSUANT TO THIS OPERATING AGREEMENT HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR THE SECURITIES LAWS OF ANY STATE OR FOREIGN JURISDICTION. THE SECURITIES MAY NOT BE TRANSFERRED WITHOUT REGISTRATION UNDER SUCH LAWS OR THE AVAILABILITY OF AN EXEMPTION FROM REGISTRATION UNDER SUCH LAWS.

LIMITED LIABILITY COMPANY AGREEMENT

FOR

PLR IP HOLDINGS, LLC

A DELAWARE LIMITED LIABILITY COMPANY

LIMITED LIABILITY COMPANY AGREEMENT FOR PLR IP HOLDINGS, LLC

	AABILITY COMPANY AGREEMENT is entered into as of and among those parties whose names are set forth on the			
RECITALS				
WHEREAS, on	, a Certificate of Formation of PLR IP Holdings,			

LLC, a limited liability company organized under the laws of the State of Delaware (the

WHEREAS, the parties hereto desire to enter into this Agreement to provide for the management of the business and the affairs of the Company, the allocation of profits and losses, the distribution of cash of the Company among the Members, the rights, obligations and interests of the Members to each other and to the Company, and certain other matters.

NOW, THEREFORE, the parties hereto agree as follows:

"Company"), was filed with the Delaware Secretary of State; and

ARTICLE I DEFINITIONS

Capitalized terms used in this Agreement have the meanings specified in this Article or elsewhere in this Agreement. In referring to sections or provisions of the Code or Regulations, it is intended that the terms "partner" and "partnership" (or variations thereof) appearing therein shall be read, respectively, as Member or Company (or variations thereof).

- 1.1 "Advisory Board" has the meaning set forth in Section 5.2.
- 1.2 "Act" means the Delaware Limited Liability Company Act, codified in the Delaware General Corporation Law, Section 18-101 et seq., as the same may be amended from time to time.
- 1.3 "Additional Capital Return Account" means, with respect to each Member, as of any relevant date, an amount equal to such Member's aggregate Additional Capital Contributions less the aggregate amount of distributions made to such Member prior to that relevant date pursuant to Section 4.4(b).
- 1.4 "Additional Capital Contribution" has the meaning ascribed thereto in Section 3.3.
- 1.5 "Adjusted Capital Account Deficit" means, with respect to any Person, the deficit balance, if any, in such Person's Capital Account as of the end of the relevant Fiscal Year, after giving effect to the following adjustments:

- (a) credit to such Capital Account any amounts which such Person is obligated to restore pursuant to any provision of this Agreement or is deemed to be obligated to restore pursuant to the next to the last sentence of Regulations Sections 1.704-2(g)(1) and 1.704-2(i)(5) after taking into account any changes during such year in Company Minimum Gain and Member Minimum Gain; and
- (b) debit to such Capital Account the items described in Regulations Section 1.704-1(b)(2)(ii)(d)(4), (5) and (6).

The foregoing definition of Adjusted Capital Account Deficit is intended to comply with the provisions of Regulations Section 1.704-1(b)(2)(ii)(d) and shall be interpreted consistently therewith

- 1.6 "Additional ROFR Right" has the meaning set forth in Section 8.6(c).
- 1.7 "Affiliate" means with respect to a specified Person: (a) any Person that directly or indirectly through one or more intermediaries, alone or through an affiliated group, controls, is controlled by, or is under common control with, such specified Person, (b) any Person that is an officer, director, partner or trustee of, or serves in a similar capacity with respect to, such specified Person (or an Affiliate of such specified Person), (c) any Person that, directly or indirectly, is the beneficial owner of ten percent (10%) or more of any class of equity securities of, or otherwise has a substantial beneficial interest in, the specified Person or of which the specified Person is directly or indirectly the owner of ten percent (10%) or more of any class of equity securities or in which the specified Person has a substantial beneficial interest, or (d) the spouse of the specified Person.
- 1.8 **"Agreement"** means this limited liability company agreement, as originally executed and as amended from time to time.
- 1.9 "Asset Purchase Agreement" means the Asset Purchase Agreement dated of April ___, 2009 by and between the Company and the Bankruptcy Estate.
- 1.10 **"Assignee"** means a Person who has acquired a Member's Economic Interest in the Company by way of a Transfer, but who has not been admitted as a Member.
- 1.11 **"Assigning Member"** means a Member who by means of a Transfer has assigned or otherwise caused an Economic Interest in the Company to be transferred to an Assignee.
- 1.12 **"Bankruptcy Estate**" means Polaroid Corporation, debtor-in-possession, a Delaware corporation and a Member of the Company and its debtor affiliates and subsidiaries.
 - 1.13 "Bankruptcy Estate Interest" has the meaning set forth in Section 8.6.
- 1.14 "**Bona Fide Offer**" means a written offer from any proposed transferee of any Offered Interest that states (a) the form and amount of consideration being offered by such proposed transferee for such Offered Interest, and (b) the proposed timetable for the consummation of the proposed Transfer.

- 1.15 **"Book Value"** means, with respect to any asset of the Company, the asset's adjusted basis for federal income tax purposes, except as follows:
- (a) The initial Book Value of any asset contributed by a Member to the Company shall be such asset's gross fair market value at the time of such contribution, as determined by the Manager;
- (b) The Book Value shall be adjusted in the same manner as would the asset's adjusted basis for federal income tax purposes, except that the depreciation deduction taken into account each Fiscal Year for purposes of adjusting the Book Value of an asset shall be the amount of Depreciation with respect to such asset taken into account for purposes of computing Net Income or Net Loss for the Fiscal Year:
- (c) The Book Value of any asset distributed to a Member by the Company shall be such asset's gross fair market value at the time of such distribution, as determined by the Manager; and
- (d) Upon election by the Company, the Book Value of all Company assets shall be adjusted upon the events and in the manner specified in Regulations Section 1.704-1(b)(2)(iv)(f).
- 1.16 "**Business**" means the business of acquiring, owning, operating, licensing, selling and otherwise exploiting the Acquired Intellectual Property and Acquired Contracts (as such terms are defined in the Asset Purchase Agreement) and such other assets which are otherwise contributed or acquired by the Company pursuant to the Asset Purchase Agreement.
- 1.17 "**Business Day**" means any day that is not a Saturday, Sunday or other day on which the Federal Reserve Bank in Chicago, Illinois is closed.
- 1.18 "Capital Account" means, in respect of any Member, the account reflecting the capital interest of the Member in the Company, consisting of the Member's initial Capital Contribution and any additional Capital Contributions, maintained and adjusted for such Member in accordance with the provisions of Regulations Section 1.704-1(b)(2)(iv), and the provisions of this Agreement.
- 1.19 "Capital Contribution" means, with respect to any Member, the amount of money and the fair market value of any property (other than money) contributed (or deemed contributed pursuant to Regulation Section 1.704-1(b)(2)(iv)(d)) to the Company (net of liabilities secured by such contributed property that the Company is considered to assume or take "subject to" under Code Section 752) with respect to the Membership Interest held by such Member. A Capital Contribution shall not be deemed a loan.
- 1.20 "Capital Member" means any Member, other than the Bankruptcy Estate, who made an Initial Capital Contribution, or any Assignee thereof or any transferee thereof admitted as a Member pursuant to the terms of this Agreement. To the extent that any action requires the consent of the Capital Members, such action shall be made by the affirmative

vote of the Capital Members then entitled to vote representing a majority in interest of the total interests held by all of the Capital Members then entitled to vote.

- 1.21 "Cash Flow" means all cash, revenues and funds received by the Company, less the following: (i) interest, principal and other amounts due with respect to any indebtedness of the Company, (ii) cash funds used to pay expenses of the Company (including any fees or guaranteed payments paid to the Manager for services rendered to the Company), and (iii) and reasonable reserves for future expenses, debt payments, capital improvements and replacements, or any other purpose related to the Business, as determined by the Manager in its reasonable discretion.
- 1.22 "Certificate of Formation" means the Certificate of Formation of the Company filed pursuant to <u>Section 2.1</u>.
- 1.23 "Code" means the Internal Revenue Code of 1986, as amended, or any corresponding provision of any succeeding law.
- 1.24 "Company" means PLR IP Holdings, a Delaware limited liability company.
- 1.25 **"Company Minimum Gain"** has the meaning ascribed to the term "partnership minimum gain" in the Regulations Section 1.704-2(d).
 - 1.26 "Company Option Period" has the meaning set forth in Section 8.6(b).
- 1.27 "Controls," "Control," "Controlling," whether or not capitalized, means the power, directly or indirectly, to direct or cause the direction of the management and policies of a person or entity through ownership of voting securities, contract or otherwise.
 - 1.28 "Covered Transfer" has the meaning set forth in Section 8.7(a)
- 1.29 "Depreciation" means an amount equal to the depreciation, amortization or other cost recovery deduction allowable with respect to an asset for the Fiscal Year or other period, except that if the Book Value of an asset differs from its adjusted tax basis at the beginning of the Fiscal Year or other period, Depreciation will be an amount which bears the same ratio to the beginning Book Value as the Federal income tax depreciation, amortization or other cost recovery deduction for the Fiscal Year or other period bears to the beginning adjusted tax basis; *provided, however*, that if the Federal income tax depreciation, amortization or other cost recovery deduction for the Fiscal Year or other period is zero, Depreciation will be determined by reference to the beginning Book Value using any reasonable method.
 - 1.30 "Drag-Along Price" has the meaning set forth in Section 8.8(a).
 - 1.31 "**Drag-Along Right**" has the meaning set forth in <u>Section 8.8(a)</u>.
 - 1.32 "Eligible Purchasers" has the meaning set forth in Section 8.6(a).

- 1.33 **"Economic Interest"** means a Person's right to share in the income, gains, losses, deductions, credit or similar items of, and to receive distributions from, the Company, but does not include any other rights of a Member, including the right to vote, participate in the management of the Company, or the right to information concerning the business and affairs of the Company.
- 1.34 **"Economic Risk of Loss"** shall have the meaning specified in Regulations Section 1.752-2.
 - 1.35 "**Effective Date**" means the date first above written.
 - 1.36 "Exercise Notice" has the meaning set forth in Section 8.6(b).
- 1.37 **"Fiscal Year"** means the Company's fiscal year, which shall be the calendar year (except as otherwise required by law), and any partial year with respect to the fiscal years in which the Company is organized and dissolved or terminated.
- 1.38 **"Gordon Brothers**" means Gordon Brothers Brands, LLC, a Delaware limited liability company.
- 1.39 "Hilco" means Hilco PLR Company, LLC, a Delaware limited liability company.
- 1.40 "**Hilco/Gordon Brothers**" means Hilco/Gordon Brothers PLR IP Holdings, LLC, a Delaware limited liability company.
 - 1.41 "**Indemnified Parties**" has the meaning set forth in <u>Section 10.2</u>.
 - 1.42 "Initial Capital Contributions" has the meaning set forth in Section 3.2.
- 1.43 "**Initial Capital Return Account**" means, with respect to each Capital Member, as of any relevant date, an amount equal to such Capital Member's aggregate Initial Capital Contributions less the aggregate amount of distributions made to such Capital Member prior to that relevant date pursuant to <u>Section 4.4(d)</u>.
- 1.44 **"Initial Members"** means the Members set forth on Exhibit A on the date of this Agreement. A reference to an "Initial Member" means any of the Initial Members.
- 1.45 **"Interest Holder"** means any Person who holds an Economic Interest, whether as a Member or as an Assignee.
 - 1.46 "Liabilities" has the meaning set forth in Section 10.2.
- 1.47 **"Majority-in-Interest"** means Members holding more than fifty percent (50%) of the Percentage Interests.

- 1.48 **"Management Fee"** has the meaning set forth in <u>Section 5.1</u>.
- 1.49 **"Manager"** means the Person designated initially as such in <u>Section</u> 5.1(a), and its successors-in-interest.
- 1.50 **"Member"** means an Initial Member or a Person who otherwise acquires a Membership Interest and has joined this Agreement as a Member, as permitted under this Agreement, whose Membership Interest has not been terminated.
- 1.51 **"Member Minimum Gain"** has the meaning ascribed to the term "partner nonrecourse debt minimum gain" in Regulations Section 1.704-2(i)(2).
- 1.52 **"Member Nonrecourse Debt"** has the meaning ascribed to the term "partner nonrecourse debt" in Regulations Section 1.704-2(b)(4).
- 1.53 **"Member Nonrecourse Deductions"** means items of Company loss, deduction, or Code Section 705(a)(2)(b) expenditures that are attributable to Member Nonrecourse Debt within the meaning of Regulations Section 1.704-2(i).
 - 1.54 "Member Option Period" has the meaning set forth in Section 8.6(b).
- 1.55 **"Membership Interest"** means a Member's collective rights in the Company, including the Member's Economic Interest, Percentage Interest, right to vote and to participate in the management of the Company, and the right to information concerning the business and affairs of the Company.
- 1.56 "Net Income" and "Net Loss" means, for each fiscal year of the Company (or other period for which Net Income and Net Loss must be computed), an amount equal to the Company's taxable income or loss for such year or period, determined in accordance with Code Section 703(a) and the Regulations, and, for this purpose, all items of income, gain, loss or deduction required to be stated separately pursuant to Code Section 703(a)(1) shall be included in taxable income or loss. The determination of Net Income and Net Loss pursuant to the previous sentence shall be subject to the following adjustments:
- (a) Any income of the Company that is exempt from federal income tax and not otherwise taken into account in computing Net Income or Net Loss shall be added to such taxable income or loss:
- (b) Any expenditures of the Company described in Code Section 705(a)(2)(B) or treated as Code Section 705(a)(2)(B) expenditures pursuant to Regulations Section 1.704-1(b)(2)(iv)(i) and not otherwise taken into account in computing Net Income or Net Loss shall be subtracted from Net Income or Net Loss;
- (c) Gains or losses resulting from any disposition of a Company asset with respect to which gains or losses are recognized for federal income tax purposes shall be computed with reference to the Book Value of the Company asset disposed of, notwithstanding the fact that the adjusted tax basis of such Company asset differs from its Book Value;

- (d) In lieu of the depreciation, amortization and other cost recovery deductions taken into account in computing the taxable income or loss, there will be taken into account Depreciation;
- (e) If the Book Value of any Company asset is adjusted pursuant to the definition of "Book Value," the amount of the adjustment will be taken into account as gain or loss from the disposition of the asset for purposes of computing Net Income or Net Loss; and
- (f) Notwithstanding any other provision of this subsection, any items of income, gain, loss or deduction that are specially allocated shall not be taken into account in computing Net Income or Net Loss.
 - 1.57 "Non-Contributing Member" has the meaning set forth in Section 3.3.
- 1.58 "**Nonrecourse Liability**" has the meaning set forth in Regulations Section 1.752-1(a)(2).
- 1.59 "Notice" means a written notice required or permitted under this Agreement. A Notice shall be deemed given or sent when deposited, as certified mail, return receipt requested, postage and fees prepaid, in the United States mails; when personally delivered to the recipient; when transmitted by electronic means, and such transmission is electronically confirmed as having been successfully transmitted; or when delivered to the home or office of a recipient in the care of a person whom the sender has reason to believe will promptly communicate the Notice to the recipient.
 - 1.60 **"Offering Person"** has the meaning set forth in Section 8.6(a).
 - "Other Activities" has the meaning set forth in Section 5.1.
 - 1.62 "Other Members" has the meaning set forth in Section 8.8(a)
- 1.63 "Partially Adjusted Capital Accounts" means, with respect to any Member for any Fiscal Year, the Capital Account of such Member at the beginning of such year, adjusted for all Capital Contributions and distributions during such year and all special allocations pursuant to Sections 4.2 and 4.3 with respect to such year before giving effect to any allocations of Net Income or Net Loss pursuant to Section 4.1(a).
- 1.64 **"Percentage Interest"** means, with respect to a Member, the Units held by such Member, as a percentage of the total of all issued and outstanding Units. The number of Units held by each Member, and the Percentage Interest of such Member, shall be as set forth opposite such Member's name on Exhibit A attached hereto, which shall be amended from time to time in accordance with the terms of this Agreement.
- 1.65 **"Person"** means any natural person, corporation, firm, joint venture, partnership, limited liability company, trust, unincorporated organization, government or any department, political subdivision or agency of a government.

- 1.66 "**Preferred Return II Account**" means, with respect to each Member, as of any relevant date, the amount equal to the excess of (i) that Member's cumulative Preferred Return II Amount over (ii) the cumulative distributions to that Member pursuant to Section 4.4(a).
- 1.67 "**Preferred Return II Amount**" means, with respect to each Member, a return on equity investment computed like interest at the rate per Fiscal Year equal to ten percent (10%) cumulative and compounded annually (prorated for any partial Fiscal Year) on the unpaid sum of that Member's Additional Capital Return Account.
- 1.68 "**Pro Rata Portion**" means, with respect to any opportunity that a Member may have to make any Remaining Contribution, the percentage of such Remaining Contribution equal to the quotient of (i) such Member's Percentage Interest as of the applicable date, <u>divided by</u> (ii) the aggregate Percentage Interests of all Members who are not Noncontributing Members with respect to the Additional Capital Contribution that gave rise to the Remaining Contribution.
 - 1.69 "**Proprietary Information**" has the meaning set forth in Section 12.1.
- 1.70 **"Regulations"** means the income tax regulations promulgated by the United States Department of the Treasury and published in the Federal Register for the purpose of interpreting and applying the provisions of the Code, as such Regulations may be amended from time to time, including corresponding provisions of applicable successor regulations.
 - 1.71 "Regulatory Allocations" has the meaning set forth in Section 4.2(f).
- 1.72 "**Remaining Contribution**" means that portion of any Additional Capital Contribution that Noncontributing Members (if any), as a whole, elect not to contribute to the Company pursuant to <u>Section 3.3(b)</u>.
 - 1.73 "**Remaining Member**" means the Members other than the Offeror.
 - 1.74 **"ROFR Notice"** has the meaning set forth in Section 8.6(a).
 - 1.75 **"ROFR Purchase Price"** has the meaning set forth in Section 8.6(a).
 - 1.76 **"ROFR Purchase Terms"** has the meaning set forth in <u>Section 8.6(a)</u>.
 - 1.77 "Selling Member" has the meaning set forth in Section 8.7(a).
 - 1.78 **"Specified Securities"** has the meaning set forth in <u>Section 8.6(a)</u>.
- 1.79 "**Subsidiary**" means, with respect to any Person, any corporation, partnership, limited liability company, association, joint venture or other business entity of which more than fifty percent (50%) of the total voting power of Units or other ownership interests

entitled (without regard to the occurrence of any contingency) to vote in the election of the Person or Persons (whether directors, managers, trustees or other Persons performing similar functions) having the power to direct or cause the direction of the management and policies thereof is at the time owned or controlled, directly or indirectly, by that Person or one or more of the other Subsidiaries of that Person or a combination thereof.

- 1.80 "**Tag-Along Buyer**" has the meaning set forth in <u>Section 8.7(a)</u>.
- 1.81 "**Tag-Along Notice**" has the meaning set forth in <u>Section 8.7(a)</u>.
- 1.82 "**Tag-Along Participant**" has the meaning set forth in Section 8.7(b).
- 1.83 "**Tag-Along Participation Notice**" has the meaning set forth in Section 8.7(b).
 - 1.84 "**Tag-Along Transfer**" has the meaning set forth in <u>Section 8.7(a)</u>.
- Year, an amount (either positive or negative) equal to the hypothetical distribution such Member would receive, or hypothetical contribution such Member would be required to make, as the case may be, if: (i) all Company assets, including cash, were sold for cash at an aggregate price equal to their Book Value (taking into account any adjustments to Book Value for such Fiscal Year), (ii) all liabilities allocable to such assets were then satisfied according to their terms (limited, with respect to each Nonrecourse Liability, to the Book Value of the assets securing such liability), and (iii) all such proceeds from the disposition were distributed pursuant to Section 4.4, reduced by such Member's share of Member Minimum Gain and Company Minimum Gain immediately prior to such sale.
 - 1.86 "**Tax Distribution**" has the meaning set forth in <u>Section 4.5</u>.
- 1.87 **"Tax Matters Member"** means such Person as may be designated pursuant to Section 7.6.
- 1.88 **"Transfer"** means and includes, in respect of a Membership Interest, or any element thereof, when used as a noun, any sale, hypothecation, pledge, assignment, attachment, gift or other disposition of a Membership Interest or any element thereof, and, when used as a verb, to sell, hypothecate, pledge, assign, attach, bequest or otherwise dispose of a Membership Interest or any element thereof.
- 1.89 **"Transferee"** means a Person who obtains or receives a Membership Interest or any element thereof by means of a Transfer.
 - 1.90 "Transferring Members" has the meaning set forth in Section 8.8(a).

1.91 "Unit" means a unit of limited liability company interest, which represents a Member's interest in the capital, income, gains, losses, deductions and expenses of the Company and the right to vote on certain Company matters as provided in this Agreement.

ARTICLE II ORGANIZATIONAL MATTERS

2.1	Filing of Certificate	e of Formation.	The parties	have o	rganized the
Company pursuant to	the Act and the prov	visions of this Ag	reement and, f	or that p	ourpose, have
caused the Certificate	e of Formation to b	be prepared, exec	uted and filed	l with t	he Delaware
Secretary of State on		The Memb	ers agree that	the right	ts, duties and
liabilities of the Mem	bers and the Manage	r shall be as provi	ided in the Act	, except	as otherwise
expressly provided her	rein.				

- 2.2 <u>Name of Company</u>. The name of the Company is "PLR IP Holdings, LLC" The Company may do business under that name and under any other name or names which the Manager selects. If the Company does business under a name other than that set forth in its Certificate of Formation, then the Company shall comply with any requirements of the Act or applicable law.
- 2.3 <u>Address of Company</u>. The principal executive office of the Company shall be situated at One Northbrook Place, 5 Revere Drive, Suite 206, Northbrook, IL 60091, or such other place or places as may be determined by the Manager from time to time.
- 2.4 <u>Agent for Service of Process</u>. The agent for service of process on the Company shall be Corporation Service Company, 2711 Centerville Road, Suite 400, Wilmington, Delaware, 19808, or such other agent as may be determined by the Manager from time to time.
- 2.5 <u>Business Purposes</u>. The purpose of the Company is to engage in any lawful act or activity for which a limited liability company may be organized under the Act. The Company shall possess and may exercise all powers necessary or convenient to the conduct and promotion of the Business or activities of the Company or any other business.
- 2.6 <u>Tax Treatment as Partnership.</u> It is the intent of the Members that the Company shall always be operated in a manner consistent with its treatment as a "partnership" for Federal and State income tax purposes. Except as provided in the foregoing sentence, the Members intend the Company to be a limited liability company under the Act, and that they be Members, and not partners in a partnership. No Member or Manager shall take any action inconsistent with the express intent of the parties hereto.
- 2.7 <u>Term of Company's Existence</u>. The term of existence of the Company commenced on the effective date of filing of the Certificate of Formation with the Delaware Secretary of State, and shall continue in perpetuity, unless sooner terminated by the provisions of this Agreement or as provided by law.

2.8 <u>Units.</u> Each Member's interest in the Company, including such Member's interest in the capital, income, gains, losses, deductions and expenses of the Company and the right to vote on certain Company matters as provided in this Agreement, shall be represented by Units. The Company initially shall have one (1) authorized class of Units, and the Company shall have one class of Members. The Company has issued on the date hereof One Thousand (1,000) Units. The ownership by a Member of Units shall entitle such Member to allocations of Net Income and Net Loss and other items of income, gain, loss or deduction, and distributions of cash and other property, as set forth in <u>Article IV</u>. For purposes of this Agreement, Units held by the Company shall be deemed not to be outstanding. The Company may issue fractional Units and all Units shall be rounded to the third decimal place. The names of the Members shall be as set forth on Exhibit A attached hereto.

ARTICLE III CAPITAL ACCOUNTS AND CAPITAL CONTRIBUTIONS

- 3.1 <u>Capital Accounts</u>. An individual Capital Account shall be maintained for each Member in accordance with the requirements of Regulations Section 1.704-1(b)(2)(iv), and the provisions of this Agreement respecting the maintenance of Capital Accounts shall be interpreted and applied in a manner consistent with those Regulations. If any Membership Interest (or portion thereof) is Transferred pursuant to and in accordance with this Agreement, the Assignee of such Membership Interest (or portion thereof) shall succeed to the Assigning Member's Capital Account attributable to such Membership Interest (or portion thereof).
- 3.2 <u>Initial Capital Contributions</u>. In exchange for their initial Membership Interests in the Company, on the date hereof, the Initial Members shall contribute cash or property to the Company in the amounts set forth opposite such Member's name on <u>Exhibit A</u> hereto (the "**Initial Capital Contributions**"). Each Member shall initially own the number of Units, and have the initial Percentage Interest, in each case as set forth opposite such Member's name on <u>Exhibit A</u>, which shall be amended from time to time in accordance with the terms of this Agreement.

3.3 Additional Capital Contributions.

- (a) In the event that the Company requires additional capital at any time after the Effective Date, the Manager, in its sole and absolute discretion, shall deliver a Notice to each Member requesting in writing that the Members make additional Capital Contributions to the Company (individually and in the aggregate, as the case may be an "Additional Capital Contribution"). Each Member shall have the right, but not the obligation, to make a Capital Contribution to the Company within ten (10) Business Days after receipt of such request in an amount equal to the product of (i) the aggregate Additional Capital Contributions requested by the Manager, multiplied by (ii) such Member's Percentage Interest.
- (b) In the event that any Member elects not to make an Additional Capital Contribution pursuant to a request by the Manager in accordance with <u>Section 3.3(a)</u> (a "**Noncontributing Member**"), the remaining Members shall have the right, but not the

obligation, to contribute the entire Remaining Contribution to the Company (together with its initial <u>pro rata</u> share of the Additional Capital Contribution).

- (c) Each Member shall receive a credit to his Capital Account in the amount of any Additional Capital Contribution which such Member contributes to the Company.
- (d) Additional Capital Contributions may not result in dilution of any Member's Percentage Interest without such Members prior written consent, and therefore, no adjustment to Percentage Interests of the Members shall be made as a result of any Additional Capital Contributions absent such consent.
- (e) The Members may, in lieu of making a permitted Additional Capital Contribution, elect to loan to the Company, without security, an amount equal to such Member's permitted Additional Capital Contribution pursuant to a note substantially in the form of Exhibit B, and the Company will promptly, duly execute, and deliver such note to such Member.
- 3.4 <u>Return of Capital Contributions</u>. Except in accordance with the terms of this Agreement, no Member shall be entitled to withdraw, or to receive a return of, any part of a Capital Contribution or to receive any distributions, whether of money or property, from the Company.
- 3.5 <u>No Interest on Capital Contributions or Capital Accounts.</u> Except as otherwise provided in this Agreement, no interest shall be paid on any Capital Contributions or on the balance of any Capital Account.

ARTICLE IV PROFITS, LOSSES AND DISTRIBUTIONS

4.1 Basic Allocations.

- (a) <u>In General</u>. After taking into account the special allocations set forth in this <u>Article IV</u>, Net Income and Net Loss for each Fiscal Year (or portion thereof), shall be allocated among the Members in the manner that will cause their Partially Adjusted Capital Accounts to equal, as soon as possible, their Targeted Accounts.
- (b) <u>Limitation on Net Loss Allocations</u>. If any allocation of Net Losses would cause a Member to have an Adjusted Capital Account Deficit, those Net Losses instead shall be allocated to the other Members in proportion to their Percentage Interests.
- 4.2 <u>Regulatory Allocations</u>. Notwithstanding any other provision of this Agreement, the following special allocations shall be made in the following order:
- (a) <u>Minimum Gain Chargeback</u>. If there is a net decrease in Company Minimum Gain during any fiscal year, each Member shall be specially allocated items of Company income and gain for such year (and, if necessary, for subsequent years) in an amount equal to such Member's share of the net decrease in Company Minimum Gain, as determined under Regulations Section 1.704-2(g). Allocations pursuant to the previous sentence shall be

made in proportion to the respective amounts required to be allocated to each Member pursuant thereto. The items to be so allocated shall be determined in accordance with Regulations Sections 1.704-2(f)(6) and 1.704-2(j)(2). This Section 4.2(a) is intended to comply with the "minimum gain chargeback" requirements of Regulations Section 1.704-2(f) and shall be interpreted consistently therewith.

- (b) <u>Chargeback Attributable to Member Nonrecourse Debt</u>. If there is a net decrease in Member Minimum Gain during any Fiscal Year, each Member with a share of Member Minimum Gain at the beginning of such Fiscal Year shall be specially allocated items of income and gain for such Fiscal Year (and, if necessary, for subsequent Fiscal Years) in an amount equal to such Member's share of the net decrease in Member Minimum Gain, determined in accordance with Regulations Section 1.704-2(i)(4) and (5). Allocations pursuant to the previous sentence shall be made in proportion to the respective amounts required to be allocated to each Member pursuant thereto. The items to be so allocated shall be determined in accordance with Regulations Sections 1.704-2(i)(4) and 1.704-2(j)(2)(i). This <u>Section 4.2(b)</u> is intended to comply with the "partner minimum gain chargeback" requirements of Regulations Section 1.704-2(i)(4) and shall be interpreted consistently therewith.
- (c) Qualified Income Offset. If any Member unexpectedly receives any adjustment, allocation or distribution described in Regulations Section 1.704-1(b)(2)(ii)(d)(4), (5) or (6) which results in an Adjusted Capital Account Deficit for the Member, such Member shall be allocated items of income and book gain in an amount and manner sufficient to eliminate such Adjusted Capital Account Deficit as quickly as possible; *provided, however*, an allocation pursuant to this Section 4.2(c) shall be made if and only to the extent that such Member would have an Adjusted Capital Account Deficit after all other allocations provided for in this Article IV have been tentatively made as if this Section 4.2(c) were not in the Agreement. This Section 4.2(c) is intended to constitute a "qualified income offset" as provided by Regulations Section 1.704-1(b)(2)(ii)(d) and shall be interpreted consistently therewith.
- (d) <u>Member Nonrecourse Deductions</u>. Member Nonrecourse Deductions shall be allocated among the Members who bear the Economic Risk of Loss for the Member Nonrecourse Debt to which such Member Nonrecourse Deductions are attributable in the ratio in which they share Economic Risk of Loss for such Member Nonrecourse Debt. This provision is to be interpreted in a manner consistent with the requirements of Regulations Section 1.704-2(b)(4) and (i)(1).
- (e) <u>Nonrecourse Deductions</u>. Any Nonrecourse Deductions (as defined in Regulations Section 1.704-2(b)(1)) for any Fiscal Year or other period shall be specially allocated to the Members in proportion to their Percentage Interests.
- (f) <u>Regulatory Allocations</u>. The allocations set forth in this <u>Section 4.2</u> (the "**Regulatory Allocations**") are intended to comply with certain requirements of the applicable Regulations promulgated under Code Section 704(b). Notwithstanding any other provision of this <u>Article IV</u>, the Regulatory Allocations shall be taken into account in allocating Net Income, Net Loss and other items of income, gain, loss and deduction to the Members for Capital Account purposes so that, to the extent possible, the net amount of such allocations of Net

Income, Net Loss and other items shall be equal to the amount that would have been allocated to each Member if the Regulatory Allocations had not occurred.

- Allocations of Built-in Items. In accordance with Code Sections 704(b) and 704(c) and the Treasury Regulations thereunder, income, gain, loss and deduction with respect to property actually or constructively contributed to the capital of the Company shall, solely for tax purposes, be allocated among the Members so as to take account of any variation at the time of the contribution between the tax basis of the property to the Company and the fair market value of that property. Except as otherwise provided herein, any elections or other decisions relating to those allocations shall be made by the Manager with the consent of all the Members, after consultation with the Company's accountant, in any manner that reasonably reflects the purpose and intent of this Agreement. Allocations of income, gain, loss and deduction pursuant to this Section 4.3 are solely for purposes of federal, state and local taxes and shall not affect, or in any way be taken into account in computing, the Capital Account of any Member or the share of Net Income, Net Loss, other tax items or distributions of any Member pursuant to any provision of this Agreement.
- 4.4 <u>Distributions of Cash Flow.</u> Subject to <u>Section 4.5</u>, the Cash Flow for each taxable year of the Company shall be distributed to the Members, as and when determined by the Manager (in its sole discretion), in the following order of priority:
- (a) First, to the Members to the extent of, and in proportion to, their Preferred Return II Accounts;
- (b) Second, to the Members to the extent of, and in proportion to, their Additional Capital Return Accounts;
 - (c)
- $\mbox{(d)}$, Third, to the Capital Members to the extent of, and in proportion to, their Initial Capital Return Accounts; and
- (e) Fourth, to the Members in proportion to their respective Percentage Interests.
- 4.5 <u>Distributions with Respect to Taxes</u>. Within ninety (90) days after the conclusion of each Fiscal Year, and to the extent of the Company's Cash Flow, the Company shall make a distribution to each Member (a "Tax Distribution") which is equal to the amount by which (A) the product of (i) forty percent (40%) and (ii) the Company's taxable income for federal income tax purposes allocated to such Member for such Fiscal Year, exceeds (B) the aggregate amount of distributions made by the Company to such Member pursuant to <u>Section 4.4</u> with respect to such Fiscal Year; *provided, however*, that to the extent possible, the Company shall make quarterly distributions in respect of the amounts to be distributed annually pursuant to this <u>Section 4.5</u> in order to facilitate the Members' ability to make quarterly estimated tax payments with respect to the taxable income of the Company allocated to them, and in determining and making the required Tax Distribution after the end of each Fiscal Year, the Company shall make appropriate adjustments to reflect the actual results of such Fiscal Year and take into account any quarterly Tax Distributions made during such Fiscal Year. Tax

Distributions shall be treated as an advance on distributions otherwise payable pursuant to Section 4.4.

4.6 Record Dates. All Net Income and Net Loss shall be allocated, and all distributions shall be made, to the Persons shown on the records of the Company to have been Members as of the last day of the taxable year for which the allocation or distribution is to be made. Notwithstanding the foregoing, unless the Company's taxable year is separated into segments, if there is a Transfer of a Membership Interest during the taxable year, the Net Income and Net Loss shall be allocated between the original Member and the successor on the basis of the number of days each was a Member during the taxable year; provided, however, the Company's taxable year shall be segregated into two or more segments in order to account for any extraordinary non-recurring items of the Company.

4.7 Withholding Taxes.

- (a) The Company shall withhold taxes from distributions to, and allocations among, the Members to the extent required by law. Except as otherwise provided in this Section 4.7, any amount so withheld by the Company with regard to a Member shall be treated for purposes of this Agreement as an amount actually distributed to such Member pursuant to Section 4.4. An amount shall be considered withheld by the Company if, and at the time, remitted to a governmental agency without regard to whether such remittance occurs at the same time as the distribution or allocation to which it relates; *provided, however*, that an amount actually withheld from a specific distribution or designated by the Company as withheld from a specific allocation shall be treated as if distributed at the time such distribution or allocation occurs.
- (b) Each Member hereby agrees to indemnify the Company and the other Members for any liability they may incur for failure to properly withhold taxes in respect of such Member; moreover, each Member hereby agrees that neither the Company nor any other Member shall be liable for any excess taxes withheld in respect of such Member's interest and that, in the event of overwithholding, a Member's sole recourse shall be to apply for a refund from the appropriate governmental authority.
- (c) Taxes withheld by third parties from payments to the Company shall be treated as if withheld by the Company for purposes of this Section 4.7. Such withholding shall be deemed to have been made in respect of all the Members in proportion to their respective allocable shares of the underlying items of Net Income to which such third party payments are attributable. In the event that the Company receives a refund of taxes previously withheld by a third party from one or more payments to the Company, the economic benefit of such refund shall be apportioned among the Members in a manner reasonably determined to offset the prior operation of this Section 4.7(c) in respect of such withheld taxes.
- 4.8 <u>No Restoration of Negative Capital Accounts.</u> No Member shall be obligated to restore a Capital Account with a balance of less than zero.
- 4.9 <u>Compliance with Laws and Regulations</u>. It is the intent of the Members that each Member's distributive share of Company tax items be determined in accordance with

this Agreement to the fullest extent permitted by Code Sections 704(b) and 704(c). Therefore, notwithstanding anything to the contrary contained herein, if the Company is advised, as a result of the adoption of new or amended regulations pursuant to Code Sections 704(b) and 704(c), or the issuance of authorized interpretations, that the allocations provided in this Agreement are unlikely to be respected for Federal income tax purposes, the Manager is hereby granted the power to amend the allocation provisions of this Agreement, on advice of accountants and legal counsel, to the minimum extent necessary to cause such allocation provisions to be respected for Federal income tax purposes.

ARTICLE V MANAGEMENT

5.1 <u>Management</u>.

- Management by the Manager. In accordance with Section 18-401 of the Act, the business, property and affairs of the Company shall be managed, and all powers of the Company shall be exercised, by or under the direction of the "manager" or co-managers of the Company within the meaning of the Act (the "Manager"). Hilco/Gordon Brothers are hereby designated the initial Manager. Except for matters for which approval by the Members is expressly required by this Agreement or the mandatory provisions of the Act, all decisions concerning the management, operation and policy of the Company's business shall be made by the Manager and the Manager shall have full, complete and exclusive authority, power and discretion to manage and control the business, property and affairs of the Company, to make all decisions regarding those matters and to perform any and all acts or activities customary or incident to the management, operation and policy of the Company's business, property or affairs. For the purpose of clarification, each reference in the immediately preceding sentence to the word "business" includes, but is not limited to, the Business. Decisions of the Manager within its scope of authority shall be binding upon the Company and each Member (as such decision affects the Company and, therefore, each Member's Membership Interest). Except for matters for which approval by the Members is expressly required by this Agreement or the mandatory provisions of the Act, no Member shall have the right to vote on any matters concerning the affairs of the Company.
- (b) <u>Actions on Behalf of Company</u>. Without limiting the generality of <u>Section 5.1</u>, the Manager or its designee hereby is authorized and empowered to carry out and implement any and all of the following actions on behalf of the Company:
- (i) to engage personnel, including the officers of the Company, and to do such other acts and incur such other expenses on behalf of the Company as it may deem necessary or advisable in connection with the conduct of the Company's affairs;
- (ii) to engage independent attorneys, accountants, investment advisers, agents or other such Persons as it may deem necessary or advisable;
- (iii) to open, maintain, conduct and close accounts, including depositary, custodial, brokerage, margin, client or discretionary accounts, with banks, brokers,

investment advisers, or other Persons and to pay the fees and charges for transactions in such accounts; and

- (iv) to execute, deliver and perform such other contracts, agreements, and such other undertakings as it may deem necessary or advisable for the conduct of the business of the Company.
- (c) <u>Restriction on Members' Authority</u>. No Member is an agent of the Company solely by virtue of being a Member, and no Member has the authority to act for or bind the Company or any other Member solely by virtue of being a Member.
- Devotion of Time as Manager. It is acknowledged that the Manager has other business interests to which it may devote part of its time. The Manager shall devote to the Company such efforts as the Manager in its sole discretion shall deem reasonably necessary to manage the business and affairs of the Company, it being understood that nothing herein shall require the Manager to devote its full time to the business and affairs of the Company. Nothing contained in this Agreement shall preclude the Manager or any of its employees or agents from acting as a director, stockholder, officer, official, consultant or employee of any Person, from receiving compensation for services rendered in connection with the foregoing, from acting as a principal or employee of any Person with whom the Company may contract for services or otherwise, or participating in profits derived from investments in any such Person, or from investing in any securities or other property for its, his or her own account, provided that such participation or involvement shall not interfere with the performance by such Person of its, his or her duties for the Company or its, his or her obligations under this Agreement or any other agreement between such Person and the Company. In addition to the foregoing, and notwithstanding anything to the contrary contained in this Agreement, the Manager shall be entitled to engage in any business activity of any kind ("Other Activities"), and none of the Members or the Company shall have any right in or to the Other Activities or to the income or proceeds derived therefrom.
- (e) <u>Management Fee and Reimbursements; Transaction Fee.</u> The Company shall pay to the Manager (i) a one-time transaction fee equal to \$500,000 <u>plus</u> actual, documented and reasonable expenses of Hilco and Gordon Brothers arising from or in connection with the negotiation and approval of the Asset Purchase Agreement and this Agreement not to exceed \$500,000, both payable at the Closing (as defined in the Asset Purchase Agreement) and (ii) an annual management fee equal to \$500,000 payable in equal installments on the first Business Day of each month (the "**Management Fee**"). The Company shall reimburse the Manager for reasonable out-of-pocket expenses incurred by the Manager in connection with its service as Manager. Reimbursements of officers and employees shall be subject to approval by the Manager.
- (f) <u>Exculpation</u>. The Manager shall not be liable to the Company or any Members for any claims, costs, expenses, damages or losses arising out of or in connection with the performance of its duties as the Manager, or for any act or omission performed or omitted to be performed by the Manager in good faith and pursuant to the authority granted to the Manager under this Agreement, other than those directly attributable to the Manager's own gross

negligence or willful misconduct. The Manager shall not be liable to any Member for claims, costs, expenses, damages or losses due to circumstances beyond the Manager's control, including, without limitation, due to the negligence, dishonesty, bad faith or misfeasance of any employee, broker or other agent of the Company.

- 5.2 <u>Major Decisions</u>. Notwithstanding any other provision in this Agreement, Manager shall not be authorized to take, and shall not take, any of the actions or make any of the decisions that are set out below (each, a "**Major Decision**") without the prior written consent of the Bankruptcy Estate:
- (a) Enter into any material amendment to this Agreement, including without limitation, changing the composition of the Advisory Board;
 - (b) Make any material change in the business objectives of the Company; or
- (c) Sell more than 51% of the Membership Interests held by any Member to an Affiliate of Manager or any Member, except as permitted pursuant to Section 8.4.

5.3 <u>Advisory Board</u>.

- (a) An advisory board to the Company (the "Advisory Board") will be established initially consisting of up to five (5) representatives. Hilco, so long as it is a Member of the Company, shall have the power to appoint or remove two members of the Advisory Board. Gordon Brothers, so long as it is a Member of the Company, shall have the power to appoint or remove two members of the Advisory Board. The Bankruptcy Estate, so long as it is a Member of the Company, shall have the power to appoint or remove one member of the Advisory Board.
- (b) Members of the Advisory Board shall not receive any compensation or consideration for such member's service on the Advisory Board, *provided, however*, that the Company shall reimburse any reasonable expenses of members incurred in the performance of their duties as members of the Advisory Board.
- (c) The Advisory Board shall meet quarterly to advise and assist the Company and discuss the Company's strategic objectives and performance. Other than as set forth below, the Advisory Board shall not have the authority to bind or act on behalf of the Company.
- 5.4 Officers. The Manager may appoint officers of the Company in its discretion, which may include a chief executive officer, chief financial officer, one or more vice presidents, a secretary and one or more assistant secretaries. Any number of offices may be held by the same person. The Manager may choose such other officers and agents, as it shall deem necessary, who shall hold their offices for such terms and shall exercise such powers and perform such duties as shall be determined from time to time by the Manager. The officers of the Company shall be empowered to carry out the day-to-day operations of the Company and to implement the actions authorized by the Manager. Any officer may be removed either with or without cause by the Manager at any time. Any officer may resign at any time by giving written notice to the Manager. No officer need be a Member.

- 5.5 <u>Title to Assets</u>. The Manager shall cause all assets of the Company (including, without limitation, all intellectual property assets) to be held in the name of the Company.
- 8.6 Resignation, Removal and Replacement of the Manager. The Manager may resign at any time by giving written notice to the Members. The Manager may only be removed without cause pursuant to an affirmative vote of Members holding in the aggregate at least seventy-five percent (75%) of the Percentage Interests. The Manager may be removed for cause by a Majority-in-Interest upon (i) the Manager's gross negligence or willful misconduct regarding any material item of the Company's business, or (ii) in the event of a voluntary bankruptcy filing by the Manager or an involuntary bankruptcy filing involving the Manager that is not rescinded by the applicable bankruptcy court within fifteen (15) days of such involuntary filing. The resignation or removal of the Manager shall not affect the former Manager's rights as a Member, if applicable, and shall not constitute a withdrawal from the Company. Vacancies in the position of the Manager shall be filled pursuant to an affirmative vote of a Majority-in-Interest.

ARTICLE VI MEMBERSHIP, MEETINGS, VOTING

6.1 Members and Voting Rights.

- (a) Except as expressly set forth in this Agreement, no Member shall have any rights or preferences in addition to or different from those possessed by any other Member. Members shall have the right to vote upon all of those matters as to which this Agreement or the Act requires such Member action. Each Member shall vote in proportion to his, her or its Percentage Interest as of the governing record date, determined in accordance with Section 6.3 hereof. Unless otherwise provided in this Agreement, actions of Members shall be pursuant to the prevailing vote of a Majority-in-Interest. Unless otherwise expressly provided in this Agreement, no Member shall be prohibited from voting merely by reason of the fact that the Member would be voting on a matter of particular interest to that Member.
- 6.2 <u>Admission of Additional Members</u>. Additional or new Members may be admitted only with the approval of the Manager.
- 6.3 <u>Record Dates.</u> The record date for determining the Members entitled to Notice at any meeting or to vote, or entitled to receive any distribution, or to exercise any right in respect of any other lawful action, shall be the date set by the Manager.
- 6.4 <u>Membership Certificates</u>. The Company may, but shall not be required, to issue certificates evidencing Units to Persons who, from time to time, are Members of the Company; provided that once such certificates have been issued, they shall continue to be issued as necessary to reflect current Units held by Members. Certificates shall be in such form as may be approved by the Manager, shall be manually signed by the Manager, and shall bear conspicuous legends evidencing the restrictions on transfer described in, and the purchase rights of the Company and Members set forth in, <u>Article VIII</u>. All issuances, reissuances, exchanges

and other transactions in Units involving Members shall be recorded in a permanent ledger as part of the books and records of the Company. The failure of any person signing as Manager to continue to be Manager shall not affect the validity of the certificates.

- 6.5 <u>Meetings: Call, Notice and Quorum.</u> The Company shall not be required to hold an annual meeting of Members. Special meetings of the Members may be called at any time by the Manager, for the purpose of addressing any matters on which the Members may vote by delivering Notice to the Members. Meetings may be held at the principal executive office of the Company or at such other location as may be designated by the Manager. Following the call of a meeting, the Manager shall give Notice of such meeting not less than five (5) or more than sixty (60) days prior to the date of the meeting to all Members entitled to vote at the meeting. The Notice shall state the place, date, and hour of the meeting and the general nature of business to be transacted. No other business may be transacted at the meeting. A quorum at any meeting of Members shall consist of a Majority-in-Interest, represented in person or by proxy. The Members present at a duly called or held meeting at which a quorum is present may continue to transact business until adjournment, notwithstanding the withdrawal of enough Members to leave less than a quorum, if the action taken, other than adjournment, is approved by the requisite Percentage Interests as specified in this Agreement or the Act.
- Adjournment of Meetings. A meeting of Members at which a quorum is present may be adjourned to another time or place and any business which might have been transacted at the original meeting may be transacted at the adjourned meeting. If a quorum is not present at an original meeting, that meeting may be adjourned by the vote of a Majority-in-Interest represented at that meeting either in person or by proxy. Notice of the adjourned meeting need not be given to Members entitled to Notice if the time and place thereof are announced at the meeting at which the adjournment is taken, unless the adjournment is for more than forty-five (45) days, or if, after the adjournment, a new record date is fixed for the adjourned meeting, in which cases Notice of the adjourned meeting shall be given to each Member of record entitled to vote at the adjourned meeting in the manner provided in Section 6.5.
- Maiver of Notice. The transactions of any meeting of Members, however called and noticed, and wherever held, shall be as valid as though consummated at a meeting duly held after regular call and notice, if a quorum is present at that meeting, either in person or by proxy, and if, either before or after the meeting, each of the persons entitled to vote, not present in person or by proxy, signs either a written waiver of notice, a consent to the holding of the meeting, or an approval of the minutes of the meeting. Attendance of a Member at a meeting shall constitute waiver of notice, except when that Member objects, at the beginning of the meeting, to the transaction of any business on the ground that the meeting was not lawfully called or convened. Attendance at a meeting is not a waiver of any right to object to the consideration of matters required to be described in the notice of the meeting and not so included, if the objection is expressly made at the meeting.
- 6.8 <u>Proxies</u>. At all meetings of Members, a Member may vote in person or by proxy, which must be in writing. Such proxy shall be filed with the Manager before or at the time of the meeting, and may be filed by facsimile transmission to the Manager at the principal

office of the Company or such other address as may be given by the Manager to the Members for such purposes.

- 6.9 <u>Participation in Meetings by Conference Telephone</u>. Members may participate in a meeting through use of conference telephone or similar communications equipment, so long as all Members participating in such meeting can hear one another. Such participation shall be deemed attendance at the meeting.
- 6.10 Action by Members Without a Meeting. Any action that may be taken at any meeting of the Members may be taken without a meeting if a consent in writing, setting forth the action so taken, is signed by all the Members entitled to vote on such matter.

ARTICLE VII ACCOUNTING AND FINANCIAL REPORTING

- 7.1 Accounts and Accounting. Proper and complete books of account of the Company's business, in which each Company transaction shall be fully and accurately entered, shall be kept at the Company's principal executive office, and at such other locations as the Manager shall determine from time to time, and shall be open to inspection and copying by each Member or his, her or its authorized representatives upon reasonable Notice and for a proper purpose during normal business hours. The costs of such inspection and copying shall be borne by the Member.
- 7.2 <u>Accounting</u>. The financial statements of the Company shall be prepared in a form which is appropriate and adequate for the Company's business and for carrying out the provisions of this Agreement. The annual accounting period of the Company shall be its taxable year. The Company's taxable year shall be selected by the Manager, subject to the requirements of the Code.
- 7.3 <u>Records.</u> At all times during the term of existence of the Company, and beyond that term if the Manager deems it necessary, the Manager shall keep or cause to be kept the books of account referred to in Section 7.1, together with:
- (a) A current list of the full name and last known business or residence address of each Member, together with the Capital Contributions and the share in Net Income and Net Loss of each Member;
 - (b) A copy of the Certificate of Formation and all amendments thereto;
- (c) Copies of the Company's Federal, state, and local income tax or information returns and reports, if any, for the six (6) most recent taxable years, if available;
- (d) An executed counterpart of this Agreement, as amended from time to time;

- (e) Any powers of attorney pursuant to which the Certificate of Formation or any amendments thereto were executed;
- (f) Copies of the financial statements of the Company for the six (6) most recent fiscal years, if available; and
- (g) The books and records of the Company as they relate to the Company's internal affairs for the current and past four (4) Fiscal Years of the Company.

7.4 <u>Member's Rights to Records.</u>

- (a) Upon the request of any Member, for purposes reasonably related to the interest of such a Member, the Manager shall cause to be promptly delivered to such Member, at the expense of the Member, a copy of the information required to be maintained pursuant to Section 7.3.
- (b) Each Member has the right, upon reasonable request, for purposes reasonably related to the interests of such a Member, to each of the following:
- (i) To inspect and copy during normal business hours any of the records required by Section 7.3;
- (ii) To obtain from the Company promptly after becoming available, a copy of the Company's Federal, state and local income tax or information returns for each year; or
- (iii) At the Member's expense, to cause an audit of the Company's books and records to be prepared by independent accountants for the period requested by the Member.
- (c) The Manager shall cause to be promptly furnished to each Member a copy of any amendment to the Certificate of Formation.
- 7.5 <u>Financial Reports; Budget</u>. In connection with its management of the day-to-day operations of the Company, the Manager shall, among other things:
- (a) Within sixty (60) days after the end of each Fiscal Year (commencing with the Fiscal Year ending on December 31, 2009), deliver to the Members annual audited financial statements for the Company prepared in accordance with generally accepted accounting principles;
- (b) Within thirty (30) days after the end of each of the first three fiscal quarters of each Fiscal Year (commencing with the fiscal quarter ending June 30, 2009), deliver to the Members unaudited quarterly financial reports of the Company, which will provide narrative and summary financial information of the Company; and

- (c) Use commercially reasonable efforts to transmit to each Member within 120 days after the end of each Fiscal Year such information as such Member reasonably may request to enable it to complete its tax returns or to fulfill any other reporting requirements, provided that the Manager can obtain such additional information without unreasonable effort or expense.
- 7.6 <u>Tax Matters Member</u>. The Manager shall act as Tax Matters Member of the Company pursuant to Code Section 6231(a)(7).

ARTICLE VIII WITHDRAWAL OF MEMBERS; TRANSFERS OF MEMBERSHIP INTERESTS

- 8.1 <u>Transfer and Assignment of Interests</u>. Except for Transfers made pursuant to <u>Sections 8.4</u>, <u>8.6</u>, <u>8.7</u> and <u>8.8</u>, and subject to the requirements of this <u>Article VIII</u>, a Member may only Transfer all or any part of his, her or its Membership Interest upon the prior written approval of the Manager. After the consummation of any Transfer of any part of a Member's Membership Interest, the Membership Interest so Transferred shall continue to be subject to the terms and provisions of this Agreement and any further Transfers shall be required to comply with all the terms and provisions of this Agreement.
- 8.2 <u>Further Restrictions on Transfer of Interests.</u> In addition to other restrictions contained in this Agreement, no Member shall Transfer all or any part of its Membership Interest: (i) without compliance with all federal and state securities laws to the extent applicable; (ii) unless the transferor pays all expenses reasonably incurred by the Company, including reasonable attorneys' fees and costs, in connection with the Transfer; and (iii) if the Membership Interest to be transferred, when added to the total of all other Membership Interests transferred in the preceding twelve (12) consecutive months prior thereto, would cause the tax termination of the Company under Code Section 708(b)(1)(B), unless such tax termination would not have a direct material adverse effect on any non-transferring Members.
- 8.3 <u>Substitution of Members.</u> An Assignee of a Membership Interest shall have the right to become a substitute Member only if (i) the requirements of <u>Section 8.1</u> and <u>8.2</u> relating to approval of the Manager, securities and tax requirements hereof are met; (ii) the Assignee executes an instrument reasonably satisfactory to the Members accepting and adopting the terms and provisions of this Agreement as such were applicable to the transferring Member at the time of transfer; and (iii) the Assignee pays any reasonable expenses in connection with its admission as a substitute Member. The admission of an Assignee as a substitute Member shall not result in the release of the Member who assigned the Membership Interest from any liability that such Member may have to the Company.
- 8.4 <u>Permitted Transfers</u>. Notwithstanding the provisions of <u>Sections 8.1</u>, <u>8.6</u> and <u>8.7</u>, the Bankruptcy Estate may, subject only to compliance with Section 8.2, transfer its Membership Interest to a creditor trust upon confirmation of its Plan of Reorganization; provided that such creditor trust agrees to be bound by the terms and conditions of this Agreement that were binding on the Bankruptcy Estate at the time of such transfer. Upon such transfer, the

successor shall be deemed to hold all rights, status and privileges of the Bankruptcy Estate under this Agreement as if such successor was originally named herein as the holder of the Bankruptcy Estate Interest. In addition, so long as a Member retains sole voting rights with respect to a transferred Membership Interest, the Membership Interest of any Member may be transferred, with or without consideration, subject to compliance with Section 8.2, and without the prior consent of the Manager, to (i) an inter vivos trust for estate planning purposes, (ii) a spouse or any lineal descendant of a Member or owner of an Economic Interest, and with respect to a Member or owner of an Economic Interest that is a trust, the spouse or lineal descendant of its trustor, or (iii) any Affiliate of such Member.

8.5 <u>Effective Date of Permitted Transfers.</u> Any permitted Transfer of all or any portion of a Member's Membership Interest shall be effective on the day following the date upon which the requirements applicable to such Transfer pursuant to this Agreement have been satisfied. The Member that is a party to the Transfer shall provide all other Members with written notice of such Transfer as promptly as possible after the requirements applicable to such Transfer pursuant to this Agreement have been met, as well as any such documents as reasonably requested by the other Members upon such request. Any transferee of all or any portion of a Member's Membership Interest shall take subject to the terms and provisions of this Agreement.

8.6 Right of First Refusal.

- (a) In the event that any Member shall wish to Transfer any Membership Interest to a third party or third parties other than as set forth in Section 8.4 such Member (the "Offering Person") shall deliver a written notice (the "ROFR Notice") to the Company and each of the other Members (collectively, the "Eligible Purchasers") offering such Membership Interests, first to the Company, and then to the remaining Eligible Purchasers, and specifying in reasonable detail the number of Membership Interests proposed to be Transferred (the "Specified Securities"), specifying in reasonable detail the offered purchase price therefor, which must be in cash in United States dollars (the "ROFR Purchase Price"), and any other material terms and conditions (the "ROFR Purchase Terms").
- (b) For a period of 20 days after the ROFR Notice has been delivered to the Company (the "Company Option Period"), the Company shall have the right to purchase all or any portion of the Specified Securities by delivering a written notice (an "Exercise Notice") to the Offering Person and each of the Eligible Purchasers prior to the expiration of the Company Option Period, specifying the number of Specified Securities to be purchased at the ROFR Purchase Price, and its acceptance of the ROFR Terms. The failure of the Company to respond within the Company Option Period shall be deemed to be a waiver of the Company's rights under this Section 8.5(b) with respect to such Transfer. If the Company waives or is deemed to have waived its rights under this Section 8.5(b), then for a period of 20 days (the "Member Option Period") after the later to occur of (i) the expiration of the Company Option Period and (ii) the expiration of the 10-day period following the Offering Person's receipt of an Exercise Notice from the Company with respect to a portion of the Specified Securities, each of the Members shall have the right to purchase all of the remaining Specified Securities pro rata in proportion to their respective Membership Interests by delivering an Exercise Notice to the Offering Person and each of the other Eligible Purchasers prior to the expiration of the Member

24

Option Period, of such Members agreement to purchase all or a portion of its pro rata share, based upon relative Membership Interests, of the remaining Specified Securities under the ROFR Purchase Terms. The failure of a Member to respond within the Member Period shall be deemed to be a waiver of such Member's rights under this Section 8.5(b) with respect to such Transfer.

- Each Eligible Purchaser shall have the additional right (the "Additional ROFR Right") to elect to purchase any or all of the shares of the Specified Securities not accepted for purchase by the Company or any other Eligible Purchaser, in which event such shares of Specified Securities not accepted by any other Eligible Purchaser shall be deemed to have been offered to and accepted by the Eligible Purchasers exercising such Additional ROFR Right in proportion to their relative Membership Interest on the same terms and at the same price per share as those specified in the Exercise Notice, but in no event shall any Eligible Purchaser exercising its Additional ROFR Right be allocated a number of shares of Additional Specified Securities in excess of the maximum number such Eligible Purchaser has offered to purchase in its notice of exercise. In the event that the Offering Person receives Exercise Notices with respect to less than all of the Specified Securities from the Company and the Eligible Purchasers, the Offering Person shall have the right to either (a) proceed with the sale of all of the Specified Securities designated for purchase by the Company and the Eligible Purchasers and to sell the remaining Specified Securities to a third party at a cash price (in United States dollars) and on other terms and conditions, taken as a whole, that are equal to or more favorable to the Offering Person than the ROFR Purchase Price and the other ROFR Purchase Terms, during the 90-day period immediately following the expiration of the Member Option Period, with any Specified Securities not transferred within such 90-day period subject to the provisions of this Section 8.5(c); or (b) elect not to Transfer any of the Specified Securities to any of the Company and of the Eligible Purchasers and to proceed with the sale of all, but not less than all of the Specified Securities to a third party at a price and on terms and conditions, taken as a whole, that are equal to or more favorable to the Offering Person than the ROFR Purchase Price and the other ROFR Purchase Terms.
- (d) The closing of the purchase by the Company or the Members of any Specified Securities under Sections 8.5(b) and 8.5(c) above, shall be held within 30 days after the Offering Person agreed in writing to sell Specified Securities to the Company or the Members, as the case may be, or at such other time as the Company or such Members, as the case may be, and the Offering Person shall agree. At such closing the Offering Person shall deliver certificates evidencing all its Specified Securities being purchased under this Section 8.5, duly endorsed, or accompanied by written instruments of transfer in form satisfactory to the proposed purchaser, duly executed by such Offering Person, and accompanied by all requisite transfer taxes, if any, and such Specified Securities shall be free and clear of any Liens and such Offering Person shall so represent and warrant, and further represent and warrant that it is the beneficial owner of such Specified Securities.

8.7 Tag-Along Rights.

(a) In the event that a Capital Member (the "Selling Member") agrees to sell more than one-half of the Membership Interest then held by such Capital Member (a "Tag Along Transfer") to any Person or group of Persons other than another Member (a

"Tag Along Buyer") or pursuant to Section 8.4, such Capital Member must first notify the other Members in writing (a "Tag-Along Notice") of such intended Transfer at least fifteen (15) days prior to the proposed date for the consummation of such transfer; which notice will contain all of the terms of the proposed Transfer including the name and address of the prospective purchaser(s), the purchase price (which is to be determined on the basis of all consideration paid or to be paid in connection with such transfer) and other terms and conditions of payment (or the basis for determining the purchase price and other terms and conditions), and the date on or about which such sale is to be consummated and the Membership Interest to be transferred.

- Along Participation Notice") each other Capital Member and, provided that no Member has any Unrecovered Capital, the Bankruptcy Estate shall have the option, exercisable by written notice to the Selling Member to participate in the proposed Tag-Along Transfer (such other Members, each a "Tag Along Participant"). In the event the Tag-Along Buyer is unwilling to so purchase all of the Selling Member's Membership Interest and all of the Tag Along Participants' Membership Interest, then each of the Selling Member and each of the Tag-Along Participants will be entitled to sell to the Tag-Along Buyer Membership Interests in an amount determined by multiplying the total Percentage Interest to be purchased by the Tag-Along Buyer by a fraction, the numerator of which is the Percentage Interest of such Member and the denominator of which is the aggregate Percentage Interests held by the Selling Member and the Tag Along Participants together.
- (c) No Member shall be required to make representations and warranties in connection with a Transfer to a Tag-Along Buyer other than customary representations and warranties, on a several and not joint basis, regarding the power and authority of such Member to engage in the Transfer, the receipt of appropriate corporate or similar authorizations, the absence of any consents or approvals applicable to such Member (other than those which have been obtained), and that such Member has good and marketable title to its Membership Interests, free and clear of all liens, claims and other encumbrances.
- (d) At the closing of any proposed Tag-Along Transfer the Tag-Along Participants shall deliver, free and clear of all liens, to the Tag-Along Buyer, the Membership Interests or Membership Interests, applicable to be transferred and shall receive in exchange therefore, pro-rata based upon Percentage Interests sold, the consideration to be paid or delivered by the Tag-Along Buyer as described in the Tag-Along Notice.

8.8 Drag-Along Rights.

(a) If at any time after the date hereof, Capital Members holding in the aggregate more than 51% of the then outstanding aggregate Membership Interest held by all Capital Members (collectively, the "Transferring Members") desire to transfer in a bona fide arm's-length sale to Persons who are unrelated third parties (for purposes of this Section 8.8, collectively, the "Proposed Transferee") all of their Membership Interests in the Company, whether in a single transaction or in a series of related transactions, then the Transferring Members shall have the right (the "Drag-Along Right") to require the Bankruptcy Estate and all Capital Members which are not Transferring Members (collectively, the "Other Members") to

26

sell to the Proposed Transferee their entire Membership Interests. The sale price to be paid to each Member will be an amount equal to the amount that would be distributed to such Member assuming that the Company distributed the aggregate sales price for all of the Membership Interests, net of closing and transaction costs, escrows and closing adjustments, in accordance with Section 4.4 (the "Drag-Along Price"). The Transfer of Membership Interests in connection with the exercise of the Drag-Along Right is referred to herein as a "Covered Transfer".

- (b) To exercise the Drag-Along Right, the Transferring Members shall give the Bankruptcy Estate and the Other Members a Notice, delivered not less than 20 days prior to the consummation of the proposed Covered Transfer, containing (i) the name and address of the Proposed Transferee, (ii) the terms and conditions of the proposed Covered Transfer and (iii) the Drag-Along Price (based on a reasonable estimate of transaction costs, escrows and closing adjustments).
- (c) In connection with a Covered Transfer pursuant to Section 8.8(a), the Bankruptcy Estate and the Other Members will, if requested by the Proposed Transferee, execute, deliver and perform agreements with the Proposed Transferee relating to the sale containing terms that are no more onerous in any material respect than those contained in the comparable agreements to be executed, delivered and performed by the Transferring Members, but shall relate solely to the Membership Interests being sold by the Bankruptcy Estate and the Other Members.
- Rights of Legal Representatives. If a Member who is an individual dies or is adjudged by a court of competent jurisdiction to be incompetent to manage the Member's person or property, the Member's executor, administrator, guardian, conservator, or other legal representative shall be entitled to receive such Member's share of income, losses and distributions but shall not be entitled to exercise any other right of the Member other than its right to sell such Member's Membership Interest as set forth herein (but not purchase the Membership Interests of other Members). If a Member is a corporation, trust, or other entity and is dissolved or terminated, the powers of that Member may be exercised by its legal representative or successor, and shall also be subject to such limitations.
- 8.10 <u>Involuntary Transfers</u>. Upon any involuntary Transfer of a Membership Interest in violation of this <u>Article VIII</u> (including, without limitation, by means of the dissolution, death or mental disability of a Member, a court award in a divorce or similar proceeding or by other operation of law), the transferee shall have no right to vote or participate in the management of the Business, property and affairs of the Company or to exercise any rights of a Member. Such transferee shall be entitled to become an Assignee only and thereafter shall only receive the share of one or more of the Company's Net Profits, Net Losses and distributions of the Company's assets to which the transferor of such Economic Interest would otherwise be entitled.

ARTICLE IX DISSOLUTION AND WINDING UP

9.1 <u>Events of Dissolution</u>. The Company shall be dissolved immediately upon the first to occur of the following events:

- (a) The happening of any event of dissolution specified in the Certificate of Formation;
 - (b) The sale of all or of substantially all of the Company's assets; and
- (c) The entry of a decree of judicial dissolution pursuant to Section 18-802 of the Act.
- 9.2 <u>Winding Up.</u> Upon the dissolution of the Company, the Company shall engage in no further business other than that necessary to wind up the business and affairs of the Company. The Manager shall wind up the affairs of the Company in an orderly manner. The Manager shall give Notice of the commencement of winding up by mail to all known creditors and claimants against the Company whose addresses appear in the records of the Company. After paying or adequately providing for the payment of all known debts and liabilities of the Company (including all costs of dissolution) and all indebtedness to any Member, the remaining assets of the Company shall be distributed or applied to the Members in accordance with Section 4.4.
- 9.3 <u>Deficits</u>. Each Member shall look solely to the assets of the Company for the return of his, her or its investment, and if the Company property remaining after the payment or discharge of the debts and liabilities of the Company is insufficient to return the investment of each Member, such Member shall have no recourse against any other Member for indemnification, contribution or reimbursement except as specifically provided in this Agreement.

ARTICLE X LIABILITY/INDEMNIFICATION

10.1 Liability.

- (a) No Member shall be personally liable for any debt, obligation or liability of the Company, whether that liability or obligation arises in contract, tort or otherwise, except as otherwise provided in the Act or in this Agreement.
- (b) No Member shall be liable, responsible or accountable, in damages or otherwise, to any Member or to the Company for any act or omission by such Member within the scope of the authority conferred on such Member by this Agreement, except for any liability that results from such Member's gross negligence or willful misconduct.

10.2 Indemnification.

(a) Except as may be prohibited by applicable law, the Company shall indemnify and hold harmless each Member, the Manager, and Company officer, their respective partners, officers, directors, shareholders, managers, members and trustees and the partners, officers, directors, shareholders, managers, members and trustees of such parties and, in the

discretion of the Manager, may indemnify and hold harmless any Affiliate thereof, and any employee or agent of, or adviser to, each Member, the Manager and officer, and their respective Affiliates (such Persons, to the extent they are required to be indemnified hereunder or are provided indemnity hereunder by the Manager, are herein collectively referred to as "Indemnified Parties") from and against any and all losses, claims, demands, costs, damages, liabilities, expenses of any nature (including reasonable attorneys' fees and expenses), judgments, fines, settlements and other amounts (collectively, the "Liabilities") arising from, or related or incidental to, any and all claims, demands, actions, suits or proceedings, civil, criminal, administrative or investigative, in which such Indemnified Party may be involved, or threatened to be involved, as a party or otherwise, and arising out of or related to the initial offering of Membership Interests in the Company, any other offering of Membership Interests in the Company, or the business, operation, administration or termination of the Company, including, without limitation, Liabilities under federal or state securities laws (and regardless of whether such Indemnified Party continues to be a Member, Manager and officer, any Affiliate of such, or a partner, officer, director, shareholder, trustee, employee or agent of, or adviser to, a Member, Manager and officer, or any of their respective Affiliates at the time any such Liabilities are paid or incurred), if such Indemnified Party acted in good faith and in a manner it believed to be in, or not opposed to, the interests of the Company, and, with respect to any criminal proceeding, did not in good faith believe its conduct was unlawful; provided, however, that such Indemnified Party shall not be indemnified against any such Liabilities, (and the Indemnified Party shall repay all amounts previously advanced by the Company pursuant to and in accordance with Section 10.2(b) hereof), that were caused by such Indemnified Party's gross negligence or willful misconduct. Notwithstanding the foregoing, to the extent that an Indemnified Party has been successful on the merits or otherwise in defense of any action, suit or proceeding referred to in this Section 10.2, or in connection with any appeal therein, or in defense of any claim, issue or matter therein, the Company shall indemnify such Indemnified Party against the expenses, including, without limitation, attorneys' and accountants' fees and expenses, incurred by such Indemnified Party in connection therewith. The termination of any pending or threatened action, suit or proceeding by judgment, order, settlement, conviction, or upon a plea of *nolo contendere* or its equivalent, shall not, in and of itself, create a presumption or otherwise constitute evidence that the Indemnified Party did not satisfy standards for indemnification set forth in this Section 10.2.

- (b) Liabilities incurred by any Indemnified Party in defending any pending or threatened claim, demand, action, suit or proceeding shall, from time to time, be paid by the Company in advance of the final disposition or settlement of such claim, demand, action, suit or proceeding, to the extent the Company has the necessary cash available, upon receipt of an undertaking by or on behalf of the Indemnified Party to repay such amounts (or a proportionate share of such amounts determined in accordance Section 10.2(c) if applicable) if it is ultimately determined that the Indemnified Party is not to be indemnified by the Company as provided in this Section 10.2.
- (c) If for any reason (other than the gross negligence or the willful misconduct or bad faith of the Indemnified Party), the foregoing indemnification is unavailable to such Indemnified Party, then the Company shall contribute to the amount paid or payable by such Indemnified Party as a result of such Liabilities in such proportion as is appropriate to reflect not

only the relative benefits received by the Company, on the one hand, and such Indemnified Party on the other hand, but also the relative fault of the Company and such Indemnified Party as well as any relevant equitable considerations.

- (d) The indemnification (or contribution) and advancement of amounts provided pursuant to this <u>Section 10.2</u> shall not be deemed exclusive of, but shall be in addition to, any other rights to which those persons seeking indemnification (or contribution) or advancement of amounts may otherwise be entitled and shall continue as to any Indemnified Party notwithstanding the dissolution or other cessation to exist of such Indemnified Party or the withdrawal, adjudication of bankruptcy or insolvency of such Indemnified Party, such Indemnified Party's no longer serving in the capacity entitling it to indemnification under the provisions of this <u>Section 10.2</u>, or the termination of the Company.
- (e) The Company may purchase and maintain insurance on behalf of any of the Indemnified Parties, and such other Persons as the Manager shall determine, against any Liabilities that may be asserted against or that may be incurred by such persons arising out of or related to the initial offering of Membership Interests, any other offering of Membership Interests or other interests in the Company, and/or the business, operation, administration or termination of the Company, regardless of whether the Company would be required to indemnify any such persons against such Liabilities under the provisions of this Agreement.
- (f) The advancement, indemnity and contribution obligations of the Company under this Section 10.2 shall be in addition to any obligation which the Company may otherwise have, shall be binding upon and inure to the benefit of any successors, assigns, heirs and personal representatives of the Company and each of the Indemnified Parties and shall not be deemed to create any rights for the benefit of any other party. The provisions of this Section 10.2 shall survive any termination of this Agreement.

ARTICLE XI POWER OF ATTORNEY

11.1 Appointment of Manager as Attorney-in-Fact.

- (a) Each Member, by the execution of this Agreement, irrevocably constitutes and appoints the Manager its true and lawful attorney-in-fact with full power and authority in its name, place and stead to execute, acknowledge, deliver, swear to, file and record at the appropriate public offices the following documents as may be necessary or appropriate to carry out the provisions of this Agreement:
- (i) All fictitious name certificates and all certificates and other instruments (including the Certificate of Formation and counterparts of this Agreement), and any amendment or restatement thereof, which the Manager deems appropriate to form, qualify or continue the Company as a limited liability company in the jurisdictions in which the Company may conduct business or in which such formation, qualification or continuation is, in the opinion of the Manager, necessary or desirable to protect the limited liability of the Members;

- (ii) All amendments to this Agreement and the Certificate of Formation adopted in accordance with the terms hereof; and
- (iii) All conveyances and other instruments which the Manager deems appropriate to reflect the dissolution and termination of the Company in accordance with the terms hereof.
- (b) The foregoing appointment shall be deemed to be a power coupled with an interest, in recognition of the fact that each of the Members under this Agreement will be relying upon the power of the Manager to act as contemplated by this Agreement in any filing and other action by it on behalf of the Company, and shall survive the bankruptcy, death, adjudication of incompetence or insanity, or dissolution of any Member hereby giving such power and the transfer or assignment of all or any part of the Membership Interest of such Member; *provided*, *however*, that in the event of the Transfer by a Member of all of its Membership Interest, the foregoing power of attorney of a transferor Member shall survive such transfer only until such time as the transferee shall have been admitted to the Company as a Member or Assignee, as the case may be, and all required documents and instruments shall have been duly executed, filed and recorded to effect such substitution.
- (c) For greater certainty, the Manager shall not take any action in contravention of this Agreement.

ARTICLE XII CONFIDENTIALITY

- 12.1 <u>Proprietary Information</u>. The Members each acknowledge and agree that it, he or she will receive and become aware of certain information which is proprietary to the Company, including, without limitation, prices, costs, personnel, knowledge, data and techniques, other non-public information concerning the business or finances of the Company, and other information the disclosure of which might harm or destroy the competitive advantage of the Company (all of the foregoing shall hereinafter be referred to as the "**Proprietary Information**"). Notwithstanding the foregoing, the Proprietary Information shall not include any information which (a) a Member obtains other than as a result of being a Member, (b) is generally known or becomes publicly available through no fault of a Person, or (c) is required to be disclosed in the context of any administrative or judicial proceeding.
- 12.2 <u>Confidentiality</u>. The Members each agree that it, he or she shall not, directly or indirectly, during the period such Person owns or holds any interest in the Company or renders services to the Company (as the case may be) and for a period of two (2) years thereafter, disclose any Proprietary Information to third parties, copy or use any Proprietary Information, or publish any Proprietary Information, except for the purpose of fulfilling its, his or her obligations to the Company.
- 12.3 <u>Equitable Relief.</u> The Members each hereby acknowledge and agree that the breach by such Person of its covenants and obligations under this Article XII will cause irreparable harm and significant injury to the Company which could be difficult to limit or

quantify. Accordingly, such Person agrees that the Company will have the right to seek an immediate injunction, specific performance or other equitable relief due to any such breach, without posting any bond therefor, in addition to any other remedies that may be available to the Company or the other Members at law or in equity.

ARTICLE XIII SECURITIES LAWS AND INVESTMENT REPRESENTATIONS

13.1 <u>Securities Laws.</u> The initial sale of Membership Interests in the Company to the Members has not been qualified or registered under the securities laws of any state, nor registered under the Securities Act of 1933, as amended, in reliance upon exemptions from the registration provisions of those laws. In addition, no attempt has been made to qualify the offering and sale of Membership Interests to Members under any state's "blue sky" laws, also in reliance upon an exemption from the requirement that a permit for issuance of securities be procured.

Each Member hereby represents and warrants (as to only such Member) to the other Members and the Company as follows:

- Preexisting Relationship or Experience. He, she or it has a preexisting personal or business relationship with the Company, the Manager or one or more of its officers or controlling persons, or by reason of his, her or its business or financial experience, or by reason of the business or financial experience of his, her or its financial advisor who is unaffiliated with and who is not compensated, directly or indirectly, by the Company or any Affiliate of the Company, he or she is capable of evaluating the risks and merits of an investment in the Company and of protecting his, her or its own interests in connection with this investment. He, she or it has been afforded ample opportunity to investigate the business of the Company, and its proposed Business, as well as to ask any questions of the Company or the Manager and has been satisfied with the responses to any such questions.
- 13.3 <u>High Risk Investment</u>. Each Member understands that there is an extremely high degree of risk in this investment. Investment into this Company should not be purchased by any purchaser who cannot afford the loss of his, her or its entire investment. An investment in a Membership Interest is riskier than an investment in publicly traded securities of companies traded on exchanges or over-the-counter, mutual funds, certificates of deposit, municipal bonds, corporate bonds, government obligations or securities purchased in firmly underwritten offerings. Only those investors who can tolerate such risk should purchase the Membership Interest. Furthermore, the Company has negligible cash and is dependant upon proceeds of this offering to finance its business.
- 13.4 <u>Disclaimer Regarding Projections</u>. Any financial projections and assumptions in the Company's use of funds, or other materials, have been provided outside of this Agreement for illustrative purposes only and must not be relied upon. The projections reflect estimates of future operating results developed by the Company without independent evaluation or analysis, based on assumptions that may not occur and over which the Company has very little control. To the extent that the assumptions are inaccurate or incomplete, the

financial condition of the Company may be adversely affected. There can be no assurance that any of the assumptions in the projections will be accurate. In the event that the assumptions are inaccurate or incomplete, the Company's profitability would be materially affected. There can be no assurance that actual results will correspond with the financial projections. If the Company were unable to sell all of the Membership Interests, it is possible that the Company may not be able to meet its projections.

- 13.5 <u>No Advertising.</u> He, she or it has not seen, received, been presented with, or been solicited by any leaflet, public promotional meeting, article or any other form of advertising or general solicitation with respect to the sale of the Membership Interest.
- 13.6 <u>Investment Intent</u>. He, she or it is acquiring the Membership Interest for investment purposes for his, her or its own account and not with a view to or for sale in connection with any distribution of all or any part of the Membership Interest. No other person will have any direct or indirect beneficial interest in or right to the Membership Interest.
- 13.7 <u>Accredited Investor</u>. Such Capital Member hereby represents that it is an "accredited investor" as defined by Rule 501 under Regulation D of the Securities Act of 1933, as amended; *provided, however*, the Bankruptcy Estate does not make any representation regarding its status as an "accredited investor" other than that the bankruptcy court with jurisdiction over the Estate has authorized the Bankruptcy Estate to enter into an Agreement on substantially the terms set forth herein.

ARTICLE XIV GENERAL PROVISIONS

- 14.1 <u>Notices</u>. Any Notice which may or must be given under this Agreement shall be addressed to a Member (c/o the Manager) at the address set forth under the Manager's name in <u>Exhibit A</u> hereto, or, if such Notice is by means of facsimile, to the facsimile number set forth under Manager's name in <u>Exhibit A</u> hereto.
- 14.2 <u>Entire Agreement; Amendment</u>. This Agreement shall constitute the whole and entire agreement of the parties hereto with respect to the matters set forth herein and shall not be modified or amended in any respect except in accordance with this Agreement.
- 14.3 <u>Choice of Law.</u> THIS AGREEMENT AND THE OBLIGATIONS OF THE PARTIES, WILL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF DELAWARE, WITHOUT REGARD TO ANY CONFLICTS-OF-LAWS PROVISION THEREOF THAT WOULD OTHERWISE REQUIRE THE APPLICATION OF THE LAW OF ANY OTHER JURISDICTION, AND NOTWITHSTANDING THE FACT THAT ONE OR MORE COUNTERPARTS HEREOF MAY BE EXECUTED OUTSIDE OF THE STATE, OR ONE OR MORE OF THE OBLIGATIONS OF THE PARTIES HEREUNDER ARE TO BE PERFORMED OUTSIDE OF THE STATE.

- 14.4 <u>Jurisdiction</u>. Each Member hereby consents to the exclusive jurisdiction of the state and federal courts sitting in Delaware. Each Member further agrees that personal jurisdiction over it may be effected by service of process by registered or certified mail addressed as provided in <u>Section 14.1</u>, and that when so made shall be as if served upon him or her personally within the State of Delaware.
- 14.5 <u>Successors and Assigns</u>. This Agreement shall be binding upon and inure to the benefit of the Members, Interest Holders, Assignees, and their respective legal representatives and successors.
- Injunctive Relief; Specific Performance. The parties hereby agree and acknowledge that a breach of any material term, condition or provision of this Agreement that provides for an obligation other than the payment of money would result in severe and irreparable injury to the other party, which injury could not be adequately compensated by an award of money damages, and the parties therefore agree and acknowledge that they shall be entitled to injunctive relief in the event of any breach of any material term, condition or provision of this Agreement, or to enjoin or prevent such a breach, including without limitation an action for specific performance hereof, and the parties hereby irrevocably consent to the issuance of any such injunction. The parties further agree that no bond or surety shall be required in connection therewith.
- 14.7 <u>Counterparts</u>. This Agreement may be executed in multiple counterparts, each of which shall be deemed an original, but all of which shall constitute one instrument.
- 14.8 <u>Number and Gender</u>. The use of the neuter gender herein shall be deemed to include the feminine and masculine genders. The use of either the singular or the plural includes the other unless the context clearly requires otherwise.
- 14.9 <u>Further Assurances</u>. Each of the parties hereto shall timely execute and deliver any and all additional documents, instruments, notices, and other assurances, and shall do any and all acts and things reasonably necessary in connection with the performance of their obligations hereunder and to carry out the intent of the parties hereto.
- 14.10 <u>Partition</u>. Each party hereto irrevocably waives any right which it may have to maintain an action for partition with respect to property of the Company.
- 14.11 <u>Authority to Contract</u>. Each Member hereby represents and covenants to the other Members that he, she or it has the capacity and authority to enter into this Agreement without the joinder of any other person. All undertakings and agreements herein shall be binding upon the Members hereto, their permitted successors and assigns.
- 14.12 <u>Titles and Headings.</u> The Article, Section and Paragraph titles and headings contained in this Agreement are inserted only as matter of convenience and for ease of reference and in no way define, limit, extend or proscribe the scope of this Agreement or the intent or content of any provision hereof. All references to sections, articles, schedules or

34

exhibits contained herein mean sections, articles, schedules or exhibits of this Agreement unless otherwise stated.

- 14.13 <u>Validity and Severability</u>. If any provision of this Agreement is held invalid or unenforceable, such decision shall not affect the validity or enforceability of any other provision of this Agreement, all of which other provisions shall remain in full force and effect.
- 14.14 <u>Statutory References</u>. Each reference in this Agreement to a particular statute or regulation, or a provision thereof, shall be deemed to refer to such statute or regulation, or provision thereof, or to any similar or superseding statute or regulation, or provision thereof, as is from time to time in effect.

[SIGNATURE PAGES FOLLOW]

IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the day and year first above written. Member and Manager

_	O/GORDON BROTHERS PLR IP DINGS, LLC, a Delaware limited liability any	
Comp		
D		
By:	N	
	Name:	
	Title:	

IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the day and year first above written.

Memb	<u>oers</u>	
HILC	O PLR C	OMPANY, LLC
By:	Name: Title:	
[]
By:	Name: Title:	
[]
By:		
	Name: Title:	

EXHIBIT A

Members, Capital Contributions and Units

<u>Members</u>	<u>Units</u>	Percentage Interest	<u>Capital</u> <u>Contributions</u>
Hilco PLR Company, LLC Address: One Northbrook Place 5 Revere Drive, Suite 206, Northbrook, IL 60091 Fax: Tel:	375	37.5%	\$24,250,000
Gordon Brothers Brands, LLC Address: 101 Huntington Avenue, 10 th Floor Boston, MA 02199 Fax: Tel:	375	37.5%	\$24,250,000
The Estate of Polaroid Corporation, et al. Address: c/o Lindquist & Vennum PLLP 4200 IDS Center 80 South Eighth Street Minneapolis, MN 55402 Fax: (612) 371-3207 Attn: Charles P. Moorse, Esq. George H. Singer, Esq.	250	25%	\$0
Total	1,000	100%	<u>\$48,500,000</u>

EXHIBIT B

FORM OF NOTE

DEMAND PROMISSORY NOTE

J		
FOR VALUE RECEIVED, the undersi	gned,	, LLC, a Delaware
limited liability company (the "Maker"), hereby	uncondit	itionally promises to pay to the order of
, a	_ (the " <u>Pa</u>	ayee"), ON DEMAND by Payee at any
time following the one year anniversary of the	he date o	o this Demand Promissory Note (this
"Note"), the principal sum of	(\$), in lawful money of the United
States of America, together with accrued intere	est on the	e unpaid principal balance from day-to-
day remaining, computed at the lesser of (i) the	Maximun	m Rate (as defined below) and (ii) a rate
per annum equal to ten (10%) (the "Interes	t Rate"),	, compounded annually to the extent
outstanding. All payments hereunder shall be r	nade at $_$, or such
other address as Payee may designate to Maker fr	om time t	to time.

- 1. Interest shall be payable monthly in arrears commencing on the first day of the first calendar month following the date hereof, and continuing regularly and monthly on the first day of each calendar month thereafter.
- 2. Interest in respect of the principal amount of this Note remaining outstanding from day-to-day shall be calculated and accrue at the Interest Rate per annum on the basis of the actual number of days elapsed over a 365 or 366-day year, as appropriate. Payee's books and records reflecting: (a) the calculation of interest; (b) the compounding of interest; and (c) the payment of principal and interest by Maker hereunder, in each case shall constitute prima facie evidence of such facts and shall be conclusive in the absence of manifest error.
- 3. Payments made to Payee by Maker hereunder shall be applied first to accrued interest and then to principal. All payments of principal and interest upon this Note shall be made by Maker to Payee in federal or other immediately available funds without setoff or other reduction. Maker may elect to prepay this Note in whole or in part from time to time without premium, penalty or notice.
- 4. Notwithstanding anything contained in this Note to the contrary, Payee shall never be deemed to have contracted for or be entitled to receive, collect or apply as interest on this Note, any amount in excess of the amount permitted and calculated at the maximum nonusurious interest rate, if any, that at any time, or from time to time, may be contracted for, taken, reserved, charged, or received on the indebtedness evidenced by this Note under the laws that are presently in effect of the United States and the State of Delaware or, to the extent permitted by law, under such applicable laws of the United States and the State of Delaware which may hereafter be in effect and which allow a higher maximum nonusurious interest rate than applicable laws now allow (the "Maximum Rate"). In the event Payee ever receives, collects or applies as interest any amount in excess of the amount permitted and calculated at the

Maximum Rate, such amount that would be excessive interest shall be applied to the reduction of the unpaid principal balance of this Note, and, if the principal balance of this Note is paid in full, any remaining excess shall forthwith be paid to Maker. In determining whether or not the interest paid or payable under any specific contingency exceeds the Maximum Rate, Maker and Payee shall, to the maximum extent permitted under applicable law: (a) characterize any non-principal payment (other than payments which are expressly designated as interest payments hereunder) as an expense, fee, or premium, rather than as interest; (b) exclude voluntary prepayments and the effect thereof; and (c) spread the total amount of interest throughout the entire contemplated term of this Note.

- 5. Maker, and any surety, endorser, guarantor or other party ever liable for payment of any sums of money on this Note, jointly and severally waive notice of acceptance, diligence, demand for payment, presentment, protest, notice of interest and non-payment, or other notice of default, notice of intention to accelerate the maturity of this Note, and notice of acceleration of the maturity of this Note, and agree that their liability under this Note shall not be affected by any renewal or extension in the time of payment hereof, or by any indulgences, or by any release or change in the security for the payment of this Note, and hereby consent to any and all renewals, extensions, indulgences, releases, or changes, regardless of the number of such renewals, extensions, indulgences, releases, or changes.
- 6. No waiver by Payee of any of its rights or remedies hereunder or under any other document evidencing or securing this Note or otherwise shall be considered a waiver of any other subsequent right or remedy of Payee; no delay or omission in the exercise or enforcement by Payee of any rights or remedies shall ever be construed as a waiver of any right or remedy of Payee; and no exercise or enforcement of any such rights or remedies shall ever be held to exhaust any right or remedy of Payee.
- 7. THE **VALIDITY OF THIS** NOTE. ITS CONSTRUCTION. INTERPRETATION, AND ENFORCEMENT, AND THE RIGHTS OF MAKER AND PAYEE HEREUNDER SHALL BE DETERMINED UNDER, GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF DELAWARE WITHOUT REGARD TO ANY CONFLICTS-OF-LAWS PROVISION THEREOF THAT WOULD OTHERWISE REQUIRE THE APPLICATION OF THE LAW OF ANY OTHER JURISDICTION. MAKER AND PAYEE AGREE THAT ALL ACTIONS OR PROCEEDINGS ARISING IN CONNECTION WITH THIS NOTE SHALL BE TRIED AND LITIGATED ONLY IN THE STATE OR FEDERAL COURTS LOCATED IN THE COUNTY OF WILMINGTON, STATE OF DELAWARE. MAKER AND PAYEE WAIVE, TO THE EXTENT PERMITTED UNDER APPLICABLE LAW, ANY RIGHT EACH MAY HAVE TO ASSERT THE DOCTRINE OF FORUM NON CONVENIENS OR TO OBJECT TO VENUE TO THE EXTENT ANY PROCEEDING IS BROUGHT IN ACCORDANCE WITH THIS SECTION 7.
- 8. If this Note is placed in the hands of an attorney for collection, or if it is collected through any legal proceedings at law or in equity or in bankruptcy, receivership or other court proceedings, Maker promises to pay all costs and expenses of collection including, but not limited to, court costs and the reasonable attorneys' fees of the holder hereof.

IN WITNESS WHEREOF, this Note	e is exe	cuted as of t	he first date written above.
	liabilit	y company	_, LLC, a Delaware limited
	By:	Name:	
		Title:	

UNITED STATES BANKRUPTCY COURT DISTRICT OF MINNESOTA

In re		JOINTLY ADMINISTERED UNDER CASE NO. 08-46617:
	POLAROID CORPORATION, ET AL.,	08-46617 (GFK)
	Debtors.	
	(includes:	
	Polaroid Holding Company;	08-46621 (GFK)
	Polaroid Consumer Electronics, LLC;	08-46620 (GFK)
	Polaroid Capital, LLC;	08-46623 (GFK)
	Polaroid Latin America I Corporation;	08-46624 (GFK)
	Polaroid Asia Pacific LLC;	08-46625 (GFK)
	Polaroid International Holding LLC;	08-46626 (GFK)
	Polaroid New Bedford Real Estate, LLC;	08-46627 (GFK)
	Polaroid Norwood Real Estate, LLC;	08-46628 (GFK)
	Polaroid Waltham Real Estate, LLC)	08-46629 (GFK)
		Chapter 11 Cases
		Judge Gregory F. Kishel

CERTIFICATE OF SERVICE

Gretchen Luessenheide of the City of New Hope, County of Hennepin, State of Minnesota, being first duly sworn on oath, states that on April 17, 2009 she served the following document:

Revised Notice of Filing of Prevailing Bidder

electronically by Notice of Electronic Filing upon all parties who have requested electronic service in these cases by filing the same via ECF with the Bankruptcy Court in the District of Minnesota.

/e/Gretchen Luessenheide	
Gretchen Luessenheide	

Schedule 1.2(q)

Artwork, Archival Documents, Images and Scientific Notes

Polaroid Collection

The collection of art, artifacts, archival documents, archival images artist notes, books, exhibition crates and memorabilia known as the "Polaroid Collection," which is located in part at an environmentally controlled art storage facility in Somerville, Massachusetts and various other locations.

Approximately 5,000 pieces of the Polaroid Collection are owned by a Seller, but are in the possession of the Musee de l'Elysee, located in Switzerland, pursuant to that certain Agreement dated as of December 14, 1990, by and between PC and the Musee de L'Elysee, as amended (the "<u>Polaroid Collection Bailment Agreement</u>").

The Polaroid Collection includes, without limitation, photographs from the artists and in approximately the quantities listed below

Artist	Approximate Photo
	Quantity
12 Instant Images Portfolio: SX-70 Experience	59
(artist unknown)	
12 Instant Images Portfolio-T105 (artist	48
unknown)	
A. Bruce Goldman	2
Aaron Siskind	11
Abe Rezny	2
Achim Liebold	1
Ad Windig	28
Adrien Sin	3
Adrienne Salinger	4
Advertisements	2
Akira Kumagaya	3
Akira Sato	4
Al Fisher	3
Alain Filhol	9
Alain Fleischer	4
Alain HervŽou	3
Alain Janssens	5
Alain Marouani	1
Alan Jaras	1
Alan Cruickshank	7
Alan Kikuchi-Yngojo	3
Alan LaMont	3
Alan Ross	32
Alan Weissman	1
Albano Guatti	2
Albert Chong	3
Albert Giordan	1

Artist	Approximate Photo Quantity
Albert Jungmann	17
Alberto Petra	2
Aldo Andredi	1
Aldo Poggi	6
Aldo Sessa	3
Alex Kayser	1
Alex Rakowsky	2
Alexander Berg	1
Alexander Carroux	12
Alexander Laidlaw/Christine Marsen	2
Alexander Marius	1
Alexandra Edwards	4
Alfonso Garces Toledano	1
Alfonso Zirpoli	3
Alice Garik	2
Alison Bradley	6
Allan Penn	22
Allan Humrore	5
Allan Maxwell	2
Allan Porter	7
Alma Davenport	72
Alvin Langdon Coburn	32
Amy Melious	2
Amy Nugent	1
Anait Stephens	4
Analuisa Fieueredo	1
Andre Gelpke	14
Andre Haluska	3
Andre Kertesz	2
Andre Thijssen	44
Andrea Blanch	3
Andrea Salomoni	2
Andrea Wolff	11
Andreas MŸller-Pohle	2
Andreas & Stephan Lauhoff	3
Andreas Mahl	40
Andreas Raab	3
Andreas Rentsch	2
Andree Tracey	13
Andres Serrano	6
Andrew Bordwin	2
Andrew L. Strout	2
Andrew Phelps	9
Andrew Shumaker	9
Andy Brilliant	2
Andy Warhol	13
Angelo Palumbo	2

Artist	Approximate Photo Quantity
Ania Bien	1
Anita Douthat	7
Ann Lovett	18
Ann McQueen	25
Anna Tomczak	6
Anne Gabriele	4
Anne C. Savedge	3
Anne Katrine Senstad	1
Anne Marie Louvet	1
Anne Mealhie	2
Anne Neely	8
Anne Rearick	1
Anne Rowland	3
Anne Warrington	11
Anne-Francoise Brillot	2
Annu Palakunnathu Matthew	1
Ansel Adams	627
Anthony Barboza	25
Anton Grassl	2
Antonia Contro	2
Antonio Bueno	2
Antonio Strati	5
APERTURE	3
Aram Dervent	3
Ardine Nelson	22
Are Flagan	3
Ariele Bonzon	1
Arlette Mosher	3
Arne Lewis	1
Arno Bauman	6
Arno Griener	1
Arno Jansen	4
Arno Rafael Minkkinen	31
Arnold Newman	39
Art Kane	7
Arthur Bacon	1
Arthur Ollman	1
Arthur Rothstein	13
Arunas Kulikauskas	2
Augusto de Luca	13
Auke Bergsma	15
Axel GrŸnewald	4
B Grad	4
B.A. Bosaiya	6
Barbara Alper	12
Barbara Bordnick	26
Barbara Bosworth	10

Artist	Approximate Photo Quantity
Barbara Cole	4
Barbara Crane	103
Barbara E. Cohen	1
Barbara Forshay	1
Barbara Hitchcock	26
Barbara Jaffe	1
Barbara Karant	3
Barbara Kasten	79
Barbara Krobath	1
Barbara Norfleet	12
Barbara Pfiff	1
Barbara Yoshida	1
Barend Houtmuller	4
Baria Ireland	1
Barnaby Evans	10
Barney Edwards	37
Barron Claiborne	1
Barry Lategan	1
Barry Louis Bell	2
Bart Glym	1
Bart Parker	11
Bart Parker/Rita Dewitt	6
Bea Nettles	22
Beat Schmidlin	5
Beate Priolo	3
Beatrice Ernst	2
Beatrice Helg	25
Beaumont Newhall	1
Bela Kalman	2
Ben Kerns	2
Ben Sollich	2
Beppe Bolchi	4
Beppe Buccafusca	4
Bernard Faucon	3
Bernard Gillet	1
Bernard Plossu	2
Bernd Fessler	5
Bernd Kracke	1
Bernd Oehmen	4
Bernhard Schmidt	1
Bernice Ficek-Swenson	4
Bernie Kessler	
Bernie Kessier Bernis von zur Muehlen	1
	1
Bert Lowenherz	2
Bert Stern	26
Betsy Connors	7
Bettina Rheims	3

Artist	Approximate Photo Quantity
Betty Hahn	14
Bill Allen	1
Bill Brandt	1
Bill Burke	39
Bill Gallery	3
Bill Owens	31
Bill Ravanesi	4
Bill Wray	1
Billie Glaser	4
Birgit Kahle	4
Bjorn Sterri	22
Blanka Wasowicz	2
Blythe Bohnen	4
Bob Chaplin	8
Bob Denton	1
Bob H. Miller	16
Bob Tycast	5
Bonnie Donohue	2
Books	2
Boyd Webb	2
Brandt Schmitt	2
Brauce Habegger	1
Brenda Black	3
Brent Sikkema	8
Brett Hilder	1
Brett Weston	34
Brian Graham	12
Brian Swift	1
Bronlyn Jones	1
Bruce Beck	2
Bruce Charlesworth	12
Bruce Martin	1
Bruce Nicoll	6
Bruce Rogovin	1
Bruno Bourel	8
Bruno Cattani	1
Bruno Stevens	6
C. Barton van Flymen	10
C. C. Church	2
CAMERAS	9
CAPTIVA	16
Carl Schultz	1
Carl Chiarenza	41
Carl Fisher	1
Carl Purcell	4
Carl Sesto	8
Carla Strache	8

Artist	Approximate Photo Quantity
Carla Weber	1
Carley Fonville	2
Carlo Savicoli	4
Carlo Tarzuolo	2
Carlos Eguiguren	5
Carlos Spaventa	2
Carlyn M. Elstrom	2
Carol Lavrillier	7
Carol Palmer	2
Carolina Furque	5
Caroline Vaughan	41
Carolyn Bates	7
Carolyn Ross	25
Carrie Mae Weems	16
CarrŽre	1
Cary Wasserman	4
Cassio Vasconcellos	6
Caterine Millinaire	1
Catherine Poncin	2
Catherine Vanaria	5
Catherine Wagner	7
Cay Lang	8
Cecelia Condit	5
Cees van Gelderen	3
Ceferino Lopez Reyes	6
Chamouret	1
Charles Corda	1
Charles Eames	19
Charles Hrbek	21
Charles Leavitt	15
Charles Meyer	2
Charles Pervis	3
Charles Purvis	8
Charles Sieburth	19
Charles Sillery	1
Charles Traub	2
Charlie Slatkin	5
Charlotte du Pin	6
Cheryl Kemp	1
Chihiro Minato	5
Chris Enos	20
Chris Johnson	15
Chris van Waggenheim	1
Christer Stromhšlm	1
Christian Boltanski	3
Christian Delbert	8
Christian Herdeg	2

Artist	Approximate Photo Quantity
Christian Vogt	41
Christiana Wirthwein	3
Christine Breslin	4
Christine Schnitzler Steinbach	1
Christine Straszweski	4
Christine Webster	2
Christoph Mulzer	8
Christopher Boas	4
Christopher Chinian	7
Christopher Doherty	6
Christopher James	20
Christopher Kai Olsen	2
Christopher Makos	1
Chuck Bradley	3
Chuck Close	14
Cindy Brown	1
Cindy Warwick	6
Ciuco Gutierrez	1
Clarence Kennedy	166
Claudia Smigrod	2
Claudio Feijo	
Claus Peter Becker	3
Colinda Taylor	2
Colleen Kenyon	3
Constantine Manos	6
Corinne Mercadier	5
Corneille	5
Cornelia Renson	3
Cornell Capa	1
Craig Barber	2
Craig Blacklock	32
Craig Dietz	20
Craig Osler	4
Curtis Williams	1
Cususala	1
Cynthia Nelissen-Nihart	2
Dagmar Hartig	1
Damien Hustinx	1
Dan Dennehy	3
Dan Powell	4
Dan Younger	2
Dana Salvo	29
Dana Simmons	3
Daniel Barraco	2
Daniel Boudinet	1
Daniel Kaufman	32
Danilo Sartoni	
Daillo Sattoill	30

Artist	Approximate Photo Quantity
Danko Pinkas	9
Danko Pinkas/Janko Stanovnik	5
Danny Lyon	3
Darcy Mara	1
David Aschkenas	3
David Bailey	2
David Bunn	1
David Em	46
David Franklin	9
David Freese	1
David Goldes	4
David Gurfinkel	2
David Haxton	8
David Heath	2
David Hockney	3
David Hoptman	2
David Lebe	3
David Levinthal	59
David Liebowitz	1
David M Hays	3
David Mahaffey	10
David Majea	4
David Masucci	1
David Mendelsohn	14
David Michael Kennedy	11
David Miller	3
David Moore	2
David Mussina	2
David Robinson	8
David Sokosh	13
David Stock	4
David van't Veen	86
David Zeiger	2
Davide Mosconi	18
Dawoud Bey	23
Debbie Fleming Caffery	1
Debora Vander Molen	4
Deborah Sternin-Damon	4
Deborah Turbeville	7
Debra Friedman	7
Debra Goldman	4
Debra Phillips	3
Declan Haun	3
Denis Brudna	2
Denis Haggerty	1
Denis Sivack	2
Denise Trouvain	3
Demse Houvain	3

Artist	Approximate Photo Quantity
Dennis Carpenter	2
Dennis Davis	1
Dennis Farber	16
Dennis Letbetter	2
Dennis Purcell/Jock Gill	2
Detlef Odenhausen	11
Diana Blok/Marlo Broekmans	34
Diana Blok	25
Diane Bush	2
Diane Farris	10
Diane Fenster	1
Diane Keaton	3
Diane Wignall	1
Dick McGraw	1
Dick Richard	2
Dick Richards	1
Dieter Gerlach	3
Dieter Koch	3
Dina Kawer	2
Dirk Wehmeyer	4
Domenico Chiriano	1
Dominique Anginot	9
Dominique Auerbacher	5
Dominique Isserman	1
Don Leavitt	2
Don Rodan	47
Don Weinstein	2
Don Worth	18
Donald Curran	28
Donald Dietz	22
Donald Lipski	3
Donald Woodman	44
Dong Mindell	2
Dora Handel	2
Dorothea Lange	6
Dorothy Bohm	4
Dorothy Norman	3
Douglas Holleley	38
Douglas Muir	1
Douglas Sandhage	18
Drew Sanborn	5
Drex Brooks	12
Duane Michals	16
Dusan Simanek	5
Dwight Primiano	2
Dylan Vitone	3
E. Jessie Shefrin	1

E.F. Kitchen 2 Earlie Hudnall 4 Eberhard Grames 22 Eberhard Rapp 1 Ed Judice 3 Ed Mitchell 6 Eddy an der Ven 4 Edgar Lissel 4 Edouard Boubat 2 Edward Gaffey 3 Edward Kaminski 1 Edward Ranney 7 Edward Weston 13 Edye Falencki 5 Eitichi Anzai 4 Eikoh Hosoe 2 Eileen Cowin 29 Elaine O'Neil 11 Elioto Porter 15 Elaine O'Neil 11 Elisabet Zeilon 3 Elisabeth Aunos 1 Elisabeth Wat 2 Elisabeth Horan 8 Elizabeth Wat 2 Ellen Land-Weber 25 Ellen Land-Weber 25 Elliot Schwarz 1 Ellot Schwarz 1 Elma	Artist	Approximate Photo Quantity
Eberhard Grames 22 Eberhard Rapp 1 Ed Judice 3 Ed Mitchell 6 Eddy van der Ven 4 Eddy and Edgar Lissel 4 Edouard Boubat 2 Edward Gaffey 3 Edward Kaminski 1 Edward Weston 13 Edye Falencki 5 Eiichi Anzai 4 Eikoh Hosoe 2 Eilene Cowin 29 Elaine D'Neil 11 Eliot Dorter 15 Elisabet Zeilon 3 Elisabet Sunday 7 Elisabeth Horan 8 Elizabeth Horan 8 Elizabeth Watt 2 Ellen Carey 39 Ellen Land-Weber 25 Elliot Erwitt 13 Elsa Dorfman 11 Emanuele Prandi 4 Emmet Gowin 4 Emmet Gowin 4 Emmet Gowin 4 Emm	E.F. Kitchen	
Eberhard Rapp 1 Ed Judice 3 Ed Mitchell 6 Eddy van der Ven 4 Edgar Lissel 4 Edouard Boubat 2 Edward Gaffey 3 Edward Kaminski 1 Edward Ranney 7 Edward Weston 13 Edye Falencki 5 Eiichi Anzai 4 Eikoh Hosoe 2 Eileen Cowin 29 Elaine Ling 2 Elaine O'Neil 11 Eliot Porter 15 Elisabett Zeilon 3 Elisabett Zeilon 3 Elisabeth Sunday 7 Elisabeth Wart 8 Elizabeth Horan 8 Elizabeth Watt 2 Ellen Carey 39 Ellen Land-Weber 25 Elliot Erwitt 13 Elliot Schildkrout 9 Elliot Schildkrout 9 Ellas Dorfman 11 <t< td=""><td>Earlie Hudnall</td><td>4</td></t<>	Earlie Hudnall	4
Ed Judice 3 Ed Mitchell 6 Eddy van der Ven 4 Edgar Lissel 4 Edouard Boubat 2 Edward Gaffey 3 Edward Kaminski 1 Edward Ranney 7 Edward Weston 13 Edye Falencki 5 Eiich Anzai 4 Eikoh Hosoe 2 Eilene Cowin 29 Elaine Ling 2 Elaine O'Neil 11 Eliot Porter 15 Elisabet Zeilon 3 Elisabeth Sunday 7 Elisabeth Sunday 7 Elisabeth Watt 2 Elizabeth Horan 8 Elizabeth Watt 2 Ellen Carey 39 Ellen Land-Weber 25 Elliot Erwitt 13 Elliot Schildkrout 9 Elliot Schwarz 1 Elsa Dorfman 11 Emmanuelle Haugel 2 E	Eberhard Grames	22
Ed Judice 3 Ed Mitchell 6 Eddy van der Ven 4 Edgar Lissel 4 Edouard Boubat 2 Edward Gaffey 3 Edward Kaminski 1 Edward Ranney 7 Edward Weston 13 Edye Falencki 5 Eiich Anzai 4 Eikoh Hosoe 2 Eilene Cowin 29 Elaine Ling 2 Elaine O'Neil 11 Eliot Porter 15 Elisabet Zeilon 3 Elisabeth Sunday 7 Elisabeth Sunday 7 Elisabeth Watt 2 Elizabeth Horan 8 Elizabeth Watt 2 Ellen Carey 39 Ellen Land-Weber 25 Elliot Erwitt 13 Elliot Schildkrout 9 Elliot Schwarz 1 Elsa Dorfman 11 Emmanuelle Haugel 2 E	Eberhard Rapp	1
Eddy van der Ven 4 Edgar Lissel 4 Edouard Boubat 2 Edward Gaffey 3 Edward Kaminski 1 Edward Weston 13 Edye Falencki 5 Eiichi Anzai 4 Eikoh Hosoe 2 Eileen Cowin 29 Elaine Ling 2 Elaine O'Neil 11 Eliot Porter 15 Elisabet Zeilon 3 Elisabeth Sunday 7 Elisabeth Horan 8 Elizabeth Horan 8 Elizabeth Watt 2 Ellen Carey 39 Ellen Carey 39 Ellen Land-Weber 25 Elliot Erwitt 13 Elliot Schildkrout 9 Ellot Schwarz 1 Elna Land-Weber 3 Esa Dorfman 11 Emmanuelle Prandi 4 Emily Medvec 3 Ermic Giovenzana 5		3
Edgar Lissel 4 Edouard Boubat 2 Edward Gaffey 3 Edward Kaminski 1 Edward Ranney 7 Edward Weston 13 Edye Falencki 5 Eiichi Anzai 4 Eikoh Hosoe 2 Eileen Cowin 29 Elaine Ling 2 Elaine O'Neil 11 Eliotabet Zeilon 3 Elisabet Zeilon 3 Elisabeth Sunday 7 Elisabeth Horos 1 Elizabeth Watt 2 Elizabeth Watt 2 Ellen Carey 39 Ellen Carey 39 Ellen Land-Weber 25 Elliot Erwitt 13 Elliot Schildkrout 9 Elliot Schwarz 1 Emalu Medvec 3 Emmanuelle Haugel 2 Emmanuelle Haugel 2 Emmanuelle Haugel 2 Eric Baden 8	Ed Mitchell	6
Edouard Boubat 2 Edward Gaffey 3 Edward Kaminski 1 Edward Ranney 7 Edward Weston 13 Edye Falencki 5 Eiichi Anzai 4 Eikoh Hosoe 2 Eileen Cowin 29 Elaine Ling 2 Elaine O'Neil 11 Eliot Porter 15 Elisabet Zeilon 3 Elisabeth Sunday 7 Elisabeth Sunday 7 Elisabeth Turk 6 Elizabeth Horan 8 Elizabeth Watt 2 Ellen Carey 39 Ellen Land-Weber 25 Elliot Erwitt 13 Elliot Schildkrout 9 Elliot Schildkrout 9 Elsa Dorfman 11 Emanuele Prandi 4 Emily Medvec 3 Emmet Gowin 4 Enric Baden 8 Eric Badell 6 E	Eddy van der Ven	4
Edouard Boubat 2 Edward Gaffey 3 Edward Kaminski 1 Edward Ranney 7 Edward Weston 13 Edye Falencki 5 Eiichi Anzai 4 Eikoh Hosoe 2 Eileen Cowin 29 Elaine Ling 2 Elaine O'Neil 11 Eliot Porter 15 Elisabet Zeilon 3 Elisabeth Sunday 7 Elisabeth Sunday 7 Elisabeth Turk 6 Elizabeth Horan 8 Elizabeth Watt 2 Ellen Carey 39 Ellen Land-Weber 25 Elliot Erwitt 13 Elliot Schildkrout 9 Elliot Schildkrout 9 Elsa Dorfman 11 Emanuele Prandi 4 Emily Medvec 3 Emmet Gowin 4 Enric Baden 8 Eric Badell 6 E	Edgar Lissel	4
Edward Kaminski 1 Edward Ranney 7 Edward Weston 13 Edye Falencki 5 Eichi Anzai 4 Eikoh Hosoe 2 Eileen Cowin 29 Elaine Ling 2 Elaine O'Neil 11 Elior Porter 15 Elisabet Zeilon 3 Elisabeth Aunos 1 Elisabeth Sunday 7 Elisabeth Horan 8 Elizabeth Horan 8 Elizabeth Watt 2 Ellen Carey 39 Ellen Carey 39 Ellot Erwitt 13 Elliot Schildkrout 9 Elliot Schildkrout 9 Elsa Dorfman 11 Emanuele Prandi 4 Emmet Gowin 4 Enrico Bossan 3 Enrico Giovenzana 5 Eric Badell 6 Eric Bedell 6 Eric Breitenbach 1 E		2
Edward Kaminski 1 Edward Ranney 7 Edward Weston 13 Edye Falencki 5 Eichi Anzai 4 Eikoh Hosoe 2 Eileen Cowin 29 Elaine Ling 2 Elaine O'Neil 11 Elior Porter 15 Elisabet Zeilon 3 Elisabeth Aunos 1 Elisabeth Sunday 7 Elisabeth Horan 8 Elizabeth Horan 8 Elizabeth Watt 2 Ellen Carey 39 Ellen Carey 39 Ellot Erwitt 13 Elliot Schildkrout 9 Elliot Schildkrout 9 Elsa Dorfman 11 Emanuele Prandi 4 Emmet Gowin 4 Enrico Bossan 3 Enrico Giovenzana 5 Eric Badell 6 Eric Bedell 6 Eric Breitenbach 1 E	Edward Gaffey	3
Edward Weston 13 Edye Falencki 5 Eiichi Anzai 4 Eikoh Hosoe 2 Eileen Cowin 29 Elaine Ling 2 Elaine O'Neil 11 Eliot Porter 15 Elisabet Zeilon 3 Elisabet Zeilon 3 Elisabeth Aunos 1 Elisabeth Sunday 7 Elisabeth Catalano 8 Elizabeth Horan 8 Elizabeth Turk 6 Elizabeth Watt 2 Ellen Carey 39 Ellen Land-Weber 25 Elliot Erwitt 13 Elliot Schildkrout 9 Elliot Schildkrout 9 Elliot Schwarz 1 Enmanuele Prandi 4 Emmet Gowin 4 Emmet Gowin 4 Enrico Bossan 3 Enrico Giovenzana 5 Eric Baden 8 Eric Breitenbach 1		1
Edward Weston 13 Edye Falencki 5 Eiichi Anzai 4 Eikoh Hosoe 2 Eileen Cowin 29 Elaine Ling 2 Elaine O'Neil 11 Eliot Porter 15 Elisabet Zeilon 3 Elisabeth Aunos 1 Elisabeth Sunday 7 Elisabeth Catalano 8 Elizabeth Horan 8 Elizabeth Horan 8 Elizabeth Watt 2 Ellen Carey 39 Ellen Land-Weber 25 Elliot Erwitt 13 Elliot Schildkrout 9 Elliot Schildkrout 9 Elliot Schwarz 1 Emanuele Prandi 4 Emmly Medvec 3 Emmet Gowin 4 Enrico Bossan 3 Enrico Giovenzana 5 Eric Baden 8 Eric Baden 6 Eric Blau 15 <td< td=""><td>Edward Ranney</td><td>7</td></td<>	Edward Ranney	7
Eilchi Anzai 4 Eikoh Hosoe 2 Eileen Cowin 29 Elaine Ling 2 Elaine O'Neil 11 Eliot Porter 15 Elisabet Zeilon 3 Elisabet Zeilon 3 Elisabeth Aunos 1 Elisabeth Sunday 7 Elisabeth Catalano 8 Elizabeth Horan 8 Elizabeth Watt 2 Ellen Carey 39 Ellen Land-Weber 25 Elliot Erwitt 13 Elliot Schildkrout 9 Elliot Schildkrout 9 Elliot Schildkrout 9 Elliot Schildkrout 9 Emmanuele Prandi 4 Emmanuele Prandi 4 Emily Medvec 3 Emmet Gowin 4 Enrico Bossan 3 Enrico Giovenzana 5 Eric Baden 8 Eric Bedell 6 Eric Breitenbach 1	-	13
Eilchi Anzai 4 Eikoh Hosoe 2 Eileen Cowin 29 Elaine Ling 2 Elaine O'Neil 11 Eliot Porter 15 Elisabet Zeilon 3 Elisabet Zeilon 3 Elisabeth Aunos 1 Elisabeth Sunday 7 Elisabeth Catalano 8 Elizabeth Horan 8 Elizabeth Watt 2 Ellen Carey 39 Ellen Land-Weber 25 Elliot Erwitt 13 Elliot Schildkrout 9 Elliot Schildkrout 9 Elliot Schildkrout 9 Elliot Schildkrout 9 Emmanuele Prandi 4 Emmanuele Prandi 4 Emily Medvec 3 Emmet Gowin 4 Enrico Bossan 3 Enrico Giovenzana 5 Eric Baden 8 Eric Bedell 6 Eric Breitenbach 1		
Eikoh Hosoe 2 Eileen Cowin 29 Elaine Ling 2 Elaine O'Neil 11 Eliot Porter 15 Elisabet Zeilon 3 Elisabeth Zeilon 3 Elisabeth Sunday 7 Elisabeth Sunday 7 Elisabeth Catalano 8 Elizabeth Horan 8 Elizabeth Turk 6 Elizabeth Watt 2 Ellen Carey 39 Ellen Land-Weber 25 Elliot Erwitt 13 Elliot Schildkrout 9 Elliot Schildkrout 9 Elliot Schwarz 1 Emanuele Prandi 4 Emily Medvec 3 Emmet Gowin 4 Enrico Bossan 3 Enrico Giovenzana 5 Eric Baden 8 Eric Bedell 6 Eric Breitenbach 1 Eric Cuvillier 3 Eric Lewandowski 6 <t< td=""><td></td><td></td></t<>		
Eileen Cowin 29 Elaine Ling 2 Elaine O'Neil 11 Eliot Porter 15 Elisabet Zeilon 3 Elisabeth Aunos 1 Elisabeth Sunday 7 Elisabeth Sunday 7 Elisabeth Horan 8 Elizabeth Horan 8 Elizabeth Watt 2 Ellen Carey 39 Ellen Land-Weber 25 Elliot Erwitt 13 Elliot Schildkrout 9 Elliot Schwarz 1 Elsa Dorfman 11 Emanuele Prandi 4 Emily Medvec 3 Emmet Gowin 4 Enrico Bossan 3 Enrico Giovenzana 5 Eric Baden 8 Eric Bedell 6 Eric Breitenbach 1 Eric Cuvillier 3 Eric Lewandowski 6 Eric Schurer 1		
Elaine O'Neil 11 Eliot Porter 15 Elisabet Zeilon 3 Elisabeth Aunos 1 Elisabeth Sunday 7 Elisabeth Sunday 7 Elisabeth Catalano 8 Elizabeth Horan 8 Elizabeth Turk 6 Elizabeth Watt 2 Ellen Carey 39 Ellen Land-Weber 25 Elliot Erwitt 13 Elliot Schildkrout 9 Elliot Schwarz 1 Elsa Dorfman 11 Emanuele Prandi 4 Emily Medvec 3 Emmet Gowin 4 Enrico Bossan 3 Enrico Giovenzana 5 Eric Baden 8 Eric Bedell 6 Eric Blau 15 Eric Breitenbach 1 Eric Cuvillier 3 Eric Lewandowski 6 Eric Schurer 1		29
Elaine O'Neil 11 Eliot Porter 15 Elisabet Zeilon 3 Elisabeth Aunos 1 Elisabeth Sunday 7 Elisabeth Sunday 7 Elisabeth Catalano 8 Elizabeth Horan 8 Elizabeth Turk 6 Elizabeth Watt 2 Ellen Carey 39 Ellen Land-Weber 25 Elliot Erwitt 13 Elliot Schildkrout 9 Elliot Schwarz 1 Elsa Dorfman 11 Emanuele Prandi 4 Emily Medvec 3 Emmet Gowin 4 Enrico Bossan 3 Enrico Giovenzana 5 Eric Baden 8 Eric Bedell 6 Eric Blau 15 Eric Breitenbach 1 Eric Cuvillier 3 Eric Lewandowski 6 Eric Schurer 1	Elaine Ling	2
Elisabeth Aunos 1 Elisabeth Sunday 7 Elisabetta Catalano 8 Elizabeth Horan 8 Elizabeth Turk 6 Elizabeth Watt 2 Ellen Carey 39 Ellen Land-Weber 25 Elliot Erwitt 13 Elliot Schildkrout 9 Elliot Schildkrout 9 Elliot Schwarz 1 Elsa Dorfman 11 Emanuele Prandi 4 Emily Medvec 3 Emmet Gowin 4 Enrico Bossan 3 Enrico Giovenzana 5 Eric Baden 8 Eric Bedell 6 Eric Blau 15 Eric Breitenbach 1 Eric Cuvillier 3 Eric Lewandowski 6 Eric Schurer 1		11
Elisabeth Aunos 1 Elisabeth Sunday 7 Elisabetta Catalano 8 Elizabeth Horan 8 Elizabeth Turk 6 Elizabeth Watt 2 Ellen Carey 39 Ellen Land-Weber 25 Elliot Erwitt 13 Elliot Schildkrout 9 Elliot Schildkrout 9 Elliot Schwarz 1 Elsa Dorfman 11 Emanuele Prandi 4 Emily Medvec 3 Emmet Gowin 4 Enrico Bossan 3 Enrico Giovenzana 5 Eric Baden 8 Eric Bedell 6 Eric Blau 15 Eric Breitenbach 1 Eric Cuvillier 3 Eric Lewandowski 6 Eric Schurer 1		15
Elisabeth Aunos 1 Elisabeth Sunday 7 Elisabetta Catalano 8 Elizabeth Horan 8 Elizabeth Turk 6 Elizabeth Watt 2 Ellen Carey 39 Ellen Land-Weber 25 Elliot Erwitt 13 Elliot Schildkrout 9 Elliot Schildkrout 9 Elliot Schwarz 1 Elsa Dorfman 11 Emanuele Prandi 4 Emily Medvec 3 Emmanuelle Haugel 2 Emmet Gowin 4 Enrico Bossan 3 Enrico Giovenzana 5 Eric Baden 8 Eric Bedell 6 Eric Blau 15 Eric Breitenbach 1 Eric Cuvillier 3 Eric Lewandowski 6 Eric Schurer 1		3
Elisabeth Sunday Elisabetta Catalano Elizabeth Horan Elizabeth Turk Elizabeth Watt Ellen Carey Ellen Land-Weber Elliot Erwitt Elliot Schildkrout Elliot Schwarz Elliot Schwarz Elsa Dorfman Emanuele Prandi Emily Medvec Emmanuelle Haugel Emmet Gowin Enrico Bossan Erric Baden Eric Baden Eric Bedell Eric Breitenbach Eric Cuvillier Eric Lewandowski Eric Schurer 7 8 8 8 8 8 8 8 8 8 8 8 8		
Elisabetta Catalano Elizabeth Horan Elizabeth Turk 6 Elizabeth Watt 2 Ellen Carey 39 Ellen Land-Weber 25 Elliot Erwitt 13 Elliot Schildkrout 9 Elliot Schwarz 1 Elsa Dorfman 11 Emanuele Prandi 4 Emily Medvec 3 Emmanuelle Haugel 2 Emmet Gowin 4 Enrico Bossan 3 Enrico Giovenzana 5 Eric Baden 8 Eric Bedell 6 Eric Blau 15 Eric Breitenbach 1 Eric Cuvillier 3 Eric Cuvillier 3 Eric Schurer		
Elizabeth Horan Elizabeth Turk Elizabeth Watt Ellen Carey Ellen Land-Weber Elliot Erwitt Elliot Schildkrout Elliot Schwarz Elsa Dorfman Emanuele Prandi Emily Medvec Emmet Gowin Enrico Bossan Erric Baden Eric Baden Eric Bedell Eric Breitenbach Eric Cuvillier Eric Cuvillier Eric Schurer 1 6 6 6 Elizabeth Turk 6 6 Elizabeth Turk 6 6 Elizabeth Turk 6 6 Elizabeth Turk 6 6 Elizabeth Watt 2 End 4 End 5 Elizabeth Watt 2 End 5 Elien Land-Weber 25 Elliot Erwitt 13 Eliot Schwarz 11 End End End End End End End	-	
Elizabeth Turk Elizabeth Watt 2 Ellen Carey 39 Ellen Land-Weber 25 Elliot Erwitt 13 Elliot Schildkrout 9 Elliot Schwarz 1 Elsa Dorfman 11 Emanuele Prandi 4 Emily Medvec 3 Emmanuelle Haugel 2 Emmet Gowin 4 Enrico Bossan 3 Enrico Giovenzana 5 Eric Baden 8 Eric Bedell 6 Eric Blau 15 Eric Breitenbach 1 Eric Cuvillier 3 Eric Schurer 1		
Ellen Carey 39 Ellen Land-Weber 25 Elliot Erwitt 13 Elliot Schildkrout 9 Elliot Schwarz 1 Elsa Dorfman 11 Emanuele Prandi 4 Emily Medvec 3 Emmanuelle Haugel 2 Emmet Gowin 4 Enrico Bossan 3 Enrico Giovenzana 5 Eric Baden 8 Eric Bedell 6 Eric Blau 15 Eric Breitenbach 1 Eric Cuvillier 3 Eric Lewandowski 6 Eric Schurer 1	Elizabeth Turk	
Ellen Carey 39 Ellen Land-Weber 25 Elliot Erwitt 13 Elliot Schildkrout 9 Elliot Schwarz 1 Elsa Dorfman 11 Emanuele Prandi 4 Emily Medvec 3 Emmanuelle Haugel 2 Emmet Gowin 4 Enrico Bossan 3 Enrico Giovenzana 5 Eric Baden 8 Eric Bedell 6 Eric Blau 15 Eric Breitenbach 1 Eric Cuvillier 3 Eric Lewandowski 6 Eric Schurer 1	Elizabeth Watt	2
Ellen Land-Weber 25 Elliot Erwitt 13 Elliot Schildkrout 9 Elliot Schwarz 1 Elsa Dorfman 11 Emanuele Prandi 4 Emily Medvec 3 Emmanuelle Haugel 2 Emmet Gowin 4 Enrico Bossan 3 Erric Giovenzana 5 Eric Baden 8 Eric Blau 15 Eric Breitenbach 1 Eric Cuvillier 3 Eric Lewandowski 6 Eric Schurer 1		39
Elliot Schidkrout 9 Elliot Schwarz 1 Elsa Dorfman 11 Emanuele Prandi 4 Emily Medvec 3 Emmanuelle Haugel 2 Emmet Gowin 4 Enrico Bossan 3 Enrico Giovenzana 5 Eric Baden 8 Eric Bedell 6 Eric Blau 15 Eric Breitenbach 1 Eric Cuvillier 3 Eric Lewandowski 6 Eric Schurer 1		25
Elliot Schwarz 1 Elsa Dorfman 11 Emanuele Prandi 4 Emily Medvec 3 Emmanuelle Haugel 2 Emmet Gowin 4 Enrico Bossan 3 Enrico Giovenzana 5 Eric Baden 8 Eric Bedell 6 Eric Blau 15 Eric Breitenbach 1 Eric Cuvillier 3 Eric Lewandowski 6 Eric Schurer 1	Elliot Erwitt	13
Elsa Dorfman 11 Emanuele Prandi 4 Emily Medvec 3 Emmanuelle Haugel 2 Emmet Gowin 4 Enrico Bossan 3 Enrico Giovenzana 5 Eric Baden 8 Eric Bedell 6 Eric Blau 15 Eric Breitenbach 1 Eric Cuvillier 3 Eric Lewandowski 6 Eric Schurer 1		9
Emanuele Prandi 4 Emily Medvec 3 Emmanuelle Haugel 2 Emmet Gowin 4 Enrico Bossan 3 Enrico Giovenzana 5 Eric Baden 8 Eric Bedell 6 Eric Blau 15 Eric Breitenbach 1 Eric Cuvillier 3 Eric Lewandowski 6 Eric Schurer 1	Elliot Schwarz	1
Emanuele Prandi 4 Emily Medvec 3 Emmanuelle Haugel 2 Emmet Gowin 4 Enrico Bossan 3 Enrico Giovenzana 5 Eric Baden 8 Eric Bedell 6 Eric Blau 15 Eric Breitenbach 1 Eric Cuvillier 3 Eric Lewandowski 6 Eric Schurer 1	Elsa Dorfman	11
Emmanuelle Haugel 2 Emmet Gowin 4 Enrico Bossan 3 Enrico Giovenzana 5 Eric Baden 8 Eric Bedell 6 Eric Blau 15 Eric Breitenbach 1 Eric Cuvillier 3 Eric Lewandowski 6 Eric Schurer 1	Emanuele Prandi	4
Emmanuelle Haugel 2 Emmet Gowin 4 Enrico Bossan 3 Enrico Giovenzana 5 Eric Baden 8 Eric Bedell 6 Eric Blau 15 Eric Breitenbach 1 Eric Cuvillier 3 Eric Lewandowski 6 Eric Schurer 1	Emily Medvec	3
Emmet Gowin 4 Enrico Bossan 3 Enrico Giovenzana 5 Eric Baden 8 Eric Bedell 6 Eric Blau 15 Eric Breitenbach 1 Eric Cuvillier 3 Eric Lewandowski 6 Eric Schurer 1		2
Enrico Giovenzana 5 Eric Baden 8 Eric Bedell 6 Eric Blau 15 Eric Breitenbach 1 Eric Cuvillier 3 Eric Lewandowski 6 Eric Schurer 1		4
Enrico Giovenzana 5 Eric Baden 8 Eric Bedell 6 Eric Blau 15 Eric Breitenbach 1 Eric Cuvillier 3 Eric Lewandowski 6 Eric Schurer 1		3
Eric Baden8Eric Bedell6Eric Blau15Eric Breitenbach1Eric Cuvillier3Eric Lewandowski6Eric Schurer1		
Eric Bedell6Eric Blau15Eric Breitenbach1Eric Cuvillier3Eric Lewandowski6Eric Schurer1		
Eric Blau15Eric Breitenbach1Eric Cuvillier3Eric Lewandowski6Eric Schurer1		
Eric Breitenbach1Eric Cuvillier3Eric Lewandowski6Eric Schurer1		
Eric Cuvillier3Eric Lewandowski6Eric Schurer1		
Eric Lewandowski 6 Eric Schurer 1		3
Eric Schurer 1		
	Erica Adams	1

Artist	Approximate Photo Quantity
Eric-Anthony Johnson	1
Erich Hartmann	17
Erich Spahn	2
Erik Lauritzen	2
Erika Blumenfeld	4
Ernest D. Scott	5
Ernesto Levy	2
Esad Cicic	1
Eugene Richards	5
Eugenia de Olazabal	14
Eve Sonneman	11
Evergon	73
Evon Streetman	11
Ewald Graber	7
F. Adler	5
F. L‡szl— Mik—	5
Fausto Gerevini	1
Fay Breed	3
Fazal Sheikh	27
Federica Bertino	2
Felicia Megginson	2
Felix Cordero	2
Ferdinando Dolfo	5
Ferenc Suto	1
Fernando Garzoni	7
Fiona Hall	41
Flavio Bizzarri	3
Florence Humbert	4
Floris NeusŸss	14
Franc Palaia	10
Francene Keery	2
Francesco Carbone	8
Francesco Maselli	2
Francine Zaslow	13
Francis Chabrolle	1
Francisco Hidalgo	2
Franco Fontana	46
Francois Hers	2
Francois Lamy	1
Frank Di Perna	5
Frank Gillette	17
Frank Gohlke	12
Frank Herrera	11
Frank Hunter	13
Frank Jackson	8
Franklin Aalders	1
Frans Rombout	2

Artist	Approximate Photo Quantity
Franx Monnier	2
Fred Cray	2
Frederic Karikese	4
Frederic Marsal	2
Frederick Brink	6
Friederich Zubler	3
Friedrich Hackstein	1
Fritz Goro	21
Fulvio Fulchiati	8
Gabriele Basilico	10
Gabrielle Keller	33
Gahg Ning	1
Gail Bryan	10
Gail Dianna Schleifer	1
Gail S. Rebhan	4
Gail Thacker	5
Gary Denham	8
Gary Pelkey	9
Gary Stubleck	1
Gavin Michaels	1
Gay Marshall	2
Gayle Portnow	1
Geanna Merola	1
George Peet	7
George Antos	5
George Janecek	6
George Krause	2
George Manupelli	1
George Parkin/Rose Farrell	8
George Rehrey/Kimiko Gunji	2
George Rybka	2
Georges Rousse	11
Georges Tourdjman	15
Georgianne Stinnett	2
Gerald Bergstein	3
Gerald Cyrus	3
Gerald Minkoff	17
Gerald Schnitzer	3
Gerald Slota	1
Gerard Petremand	1
Gerhard Schubert	6
Gerhard Vormwald	11
Geri Della Rocca de Candal	1
Gerlovina, Berghash, Gerlovin	8
Ghassan Kitmitto	2
Gian Marco Agazzi	1
Gianpaolo Barbieri	2

Artist	Approximate Photo Quantity
Giasco Bertoli	3
Gijs Royaards	1
Gilles Perrin	6
Giorgio Ghisolfi	2
Giovanni Gastel	1
Gisele Freund	2
Gladys	14
Glen McClure	4
Glenn Bydwell	4
Glenn Howlett	1
Gloria DeFilipps Brush	13
Godfried Zollner	27
Gonzalo Mezza	2
Gordon Cooper	2
Gottfried Helnwein	10
Gottfried JŠger	4
Graham Clark	1
Gunter Sachs	5
Guy Bourdin	1
Gwenn Thomas	1
Gyorgy Kepes	13
GŽrard LŸthi	3
H.J. Schmidt	8
Hakim Raquib	2
HallŽ Kiuston	4
Hank Herrera	1
Hanna Wajshoning	1
Hannah Collins	9
Hans Gedda	2
Hans Hammarskiold/Tio	2
Hans Hansen	7
Hans Hoefer	10
Hans Namuth	8
Hans Rudolf Rohrer	4
Hans Schmidt	3
Hans Wendler	1
Harold Edgerton	29
Harold Jones	3
Harry De Zitter	20
Harry Callahan	5
Harry Millyard	1
Harvey Stein	16
Helen Chadwick	8
Helen Rousakis	4
Helmut Newton	8
Helmuth Matzka	3
Henk Meulman	1

Artist	Approximate Photo Quantity
Henry Foley	7
Henry Horenstein	5
Henry Schwartz	3
Herbert Dšring-Spengler	11
HewŽ le Masson	3
Hil Scott	1
Hilton Braithwaite/Deborah Willis	12
Hiro	1
Hiromitsu Morimoto	1
Hiroshi Yamazaki	4
Hisun Wong	3
Holger TrŸlzsch	13
Holges Stumpf	6
Holland H Williams	3
Horm Botman	2
Horst Munzig	13
Horst Wackerbarth	8
Hugh Tatlock	5
HŽI\$ ne GuŽtary	31
I.G. Holmes	1
Ian Baxter	15
Ian Cowmeadow	13
Ilene Lewis	1
Ilona Lieberman	4
Ilona Manner	1
Imogen Cunningham	14
Inge Morath	17
Inge Reethof	53
Ingo Taubhorn	33
Ingrid Assman	3
Irene Pletka	14
Iris Spellings	17
Irma Eibich	5
Irutchka	5
Isabelle Waternaux	2
J. Tamada	
	1 12
J. Wayne Olson	
J.F.Kuchen, Marc Broillett	3
J.P. Fernandez	1
J.P. Gilson	1
Jaap van der Plas	6
Jack Booth	4
Jack Butler	12
Jack Caspary	1
Jack Perno	4
Jack Welpott	14
Jacques Halber	1

Jacques Pugin	Artist	Approximate Photo Quantity
Jacques PŽrŽ 2 Jaime Permuth 2 James Bones 7 James D. Colby 1 James Haberman 16 James Johnston 2 James Nitsch 2 James Radke 5 James Wood 1 Jan van der Horn 4 Jan Camp 1 Jan Faul 2 Jan Groover 4 Jan Hnizdo 3 Jan Nemec 3 Jan Nemec 3 Jan Paul Pietrzak 3 Jan Saudek 7 Jan Saudek 7 Jan Saudek 7 Jan Van Steenwijk 76 Jane Lidz 5 Jane Regan 1 Jane Regan 1 Jane W. Hudson 7 Janeil Engelstad 1 Janic Bell 2 Janic Pritchard 6 Janic Bell 2 Janic Pritchard 6 <tr< td=""><td>Jacques Pugin</td><td>10</td></tr<>	Jacques Pugin	10
Jade Albert 23 Jaime Permuth 1 James Bones 5 James D. Colby 1 James Haberman 16 James Haberman 16 James Nitsch 2 James Nitsch 2 James Radke 5 James Wood 1 Jan Camp 1 Jan Camp 2 Jan Faul 2 Jan Groover 4 Jan Hnizdo 35 Jan Nemec 31 Jan Nemec 31 Jan Paul Pietrzak 3 Jan Paul Pietrzak 3 Jan Saudek 7 Jan Saudek 7 Jan Saudek 7 Jane L. Calvin 14 Jane Regan 1 Jane Regan 1 Jane W. Hudson 1 Janeil Engelstad 1 Janic Bell 2 Janic Pritchard 6 Janic Pritchard 6		2
James Bones 7 James D. Colby 1 James Haberman 16 James Johnston 3 James Nitsch 2 James Radke 5 James Wood 1 Jan van der Horn 4 Jan Camp 1 Jan Camp 2 Jan Groover 4 Jan Hnizdo 35 Jan Nemec 31 Jan Nemec 31 Jan Otsen 2 Jan Paul Pietrzak 3 Jan Saudek 7 Jan Saudek 7 Jan van Steenwijk 76 Jane L. Calvin 14 Jane Regan 1 Jane Regan 1 Jane Tuckerman 4 Jane W. Hudson 7 Janeil Engelstad 1 Janice Bell 2 Janice Bell 2 Janice Seniak 3 Janusz Lesniak 3 Janusz Lesniak 3 <td></td> <td>23</td>		23
James D. Colby 1 James Haberman 16 James Johnston 3 James Nitsch 2 James Radke 5 James Wood 1 Jan van der Horn 4 Jan Camp 1 Jan Faul 4 Jan Groover 4 Jan Hnizdo 35 Jan Nemec 31 Jan Nemec 31 Jan Paul Pietrzak 32 Jan Paul Pietrzak 32 Jan Saudek 7 Jan van Steenwijk 76 Jane L. Calvin 14 Jane Regan 1 Jane Regan 1 Jane Tuckerman 44 Jane W. Hudson 3 Janeil Engelstad 1 Janic Bell 2 Janic Bell 2 Janic Bell 2 Janic Bell 3 Janic Bell 3 Janic Bell 3 Janic Bell 3	Jaime Permuth	1
James Haberman 16 James Johnston 3 James Nitsch 2 James Radke 5 James Wood 1 Jan van der Horn 2 Jan Camp 1 Jan Faul 4 Jan Groover 4 Jan Hnizdo 35 Jan Nemec 31 Jan Nemec 31 Jan Otsen 2 Jan Paul Pietrzak 3 Jan Saudek 7 Jan Saudek 7 Jan van Steenwijk 76 Jane L. Calvin 14 Jane Regan 1 Jane Regan 1 Jane Tuckerman 44 Jane W. Hudson 3 Janeil Engelstad 1 Janic Bell 2 Janic Bell 2 Janic Bell 2 Janic Bell 3 Janic Bell 3 Janic Bell 3 Janic Bell 3	James Bones	7
James Johnston 3 James Nitsch 2 James Radke 3 James Wood 1 Jan van der Horn 4 Jan Camp 1 Jan Faul 4 Jan Groover 2 Jan Hnizdo 35 Jan Nemec 31 Jan Otsen 2 Jan Paul Pietrzak 3 Jan Saudek 7 Jan Saudek 7 Jan van Steenwijk 76 Jane L. Calvin 14 Jane Regan 1 Jane Regan 1 Jane W. Hudson 7 Janeil Engelstad 1 Janeil Engelstad 1 Janie Pritchard 6 Janis Croft 1 Janis Croft 1 Jann Kwasneski 3 Janusz Lesniak 1 Jaroslav Benes 2 Jaroslav Razjik 15 Jason Simon 2 Javier Diaz Saiz 2 </td <td>James D. Colby</td> <td>1</td>	James D. Colby	1
James Nitsch James Radke James Wood Jan van der Horn Jan Camp Jan Faul Jan Groover Jan Hnizdo Jan Nemec Jan Paul Pietrzak Jan S‡gl Jan Saudek Jan van Steenwijk Jane L. Calvin Jane Lidz Jane Regan Jane Tuckerman Jane W. Hudson Janeil Engelstad Janelle MortŽ Janet Pritchard Janice Bell Janis Croft Jann Kwasneski Janyce Erlich-Moss Jaroslav Benes Jaroslav Razjik Jason Simon Javier Diaz Saiz	James Haberman	16
James Radke James Wood Jan van der Horn Jan Camp Jan Faul Jan Groover Jan Hnizdo Jan Nemec Jan Paul Pietrzak Jan S**gl Jan Saudek Jan van Steenwijk Jane L. Calvin Jane Lidz Jane Regan Jane Tuckerman Jane W. Hudson Janeil Engelstad Janeil Engelstad Janice Bell Janice Bell Janice Bell Janice Serich-Moss Jaroslav Benes Jaroslav Benes Jaroslav Razjik Jason Simon Javier Diaz Saiz Zane Rejan Janes Steenwijk Janes Steenwi	James Johnston	3
James Wood 1 Jan van der Horn 2 Jan Camp 1 Jan Faul 2 Jan Groover 2 Jan Nemec 31 Jan Nemec 31 Jan Otsen 2 Jan Paul Pietrzak 3 Jan Saudek 7 Jan Saudek 7 Jan Saudek 7 Jan van Steenwijk 7 Jane Lidz 5 Jane Regan 1 Jane Regan 1 Jane Tuckerman 4 Janeil Engelstad 1 Janeil Engelstad 1 Janiec Bell 2 Janice Bell 2 Janice Sorft 1 Jann Kwasneski 3 Janyce Erlich-Moss 3 Jaroslav Benes 2 Jaroslav Razjik 15 Jason Simon 2 Javier Diaz Saiz 2	James Nitsch	2
Jan van der Horn 2 Jan Camp 1 Jan Faul 2 Jan Groover 2 Jan Nemec 31 Jan Otsen 2 Jan Paul Pietrzak 3 Jan Saudek 5 Jan Saudek 7 Jan van Steenwijk 76 Jane L. Calvin 14 Jane Regan 1 Jane Regan 1 Jane W. Hudson 7 Janeil Engelstad 1 Janeil Pritchard 6 Janice Bell 2 Janice Scroft 1 Jan Kwasneski 3 Janusz Lesniak 1 Janyce Erlich-Moss 3 Jaroslav Benes 2 Jaroslav Razjik 15 Jason Simon 2 Javier Diaz Saiz 2	James Radke	5
Jan Camp 1 Jan Faul 2 Jan Groover 2 Jan Hnizdo 35 Jan Nemec 31 Jan Otsen 2 Jan Paul Pietrzak 3 Jan Saudek 7 Jan Saudek 7 Jan van Steenwijk 76 Jane L. Calvin 14 Jane Lidz 5 Jane Regan 1 Jane Tuckerman 44 Janeil Engelstad 1 Janeil Engelstad 1 Janeil Pritchard 6 Janis Croft 1 Jann Kwasneski 3 Janusz Lesniak 1 Janyce Erlich-Moss 3 Jaroslav Benes 2 Jaroslav Razjik 15 Jason Simon 2 Javier Diaz Saiz 2	James Wood	1
Jan Faul 2 Jan Groover 2 Jan Hnizdo 35 Jan Nemec 31 Jan Otsen 2 Jan Paul Pietrzak 3 Jan Staudek 3 Jan Saudek 7 Jan van Steenwijk 76 Jane L. Calvin 14 Jane Regan 1 Jane Regan 1 Jane Tuckerman 44 Jane W. Hudson 7 Janeil Engelstad 1 Janeile MortŽ 6 Janet Pritchard 6 Janis Croft 1 Janis Croft 1 Jann Kwasneski 3 Janyce Erlich-Moss 3 Jaroslav Benes 2 Jaroslav Razjik 15 Jason Simon 2 Javier Diaz Saiz 2	Jan van der Horn	4
Jan Faul 2 Jan Groover 2 Jan Hnizdo 35 Jan Nemec 31 Jan Otsen 2 Jan Paul Pietrzak 3 Jan Staudek 3 Jan Saudek 7 Jan van Steenwijk 76 Jane L. Calvin 14 Jane Regan 1 Jane Regan 1 Jane Tuckerman 44 Jane W. Hudson 7 Janeil Engelstad 1 Janeile MortŽ 6 Janet Pritchard 6 Janis Croft 1 Janis Croft 1 Jann Kwasneski 3 Janyce Erlich-Moss 3 Jaroslav Benes 2 Jaroslav Razjik 15 Jason Simon 2 Javier Diaz Saiz 2	Jan Camp	1
Jan Hnizdo 35 Jan Nemec 31 Jan Otsen 2 Jan Paul Pietrzak 3 Jan S‡gl 15 Jan Saudek 7 Jan van Steenwijk 76 Jane L. Calvin 14 Jane Regan 1 Jane Regan 1 Jane W. Hudson 7 Janeil Engelstad 1 Janelle MortŽ 6 Janet Pritchard 6 Janis Croft 1 Jann Kwasneski 3 Janusz Lesniak 1 Janyce Erlich-Moss 3 Jaroslav Benes 2 Jaroslav Razjik 15 Jason Simon 2 Javier Diaz Saiz 2		4
Jan Nemec 31 Jan Otsen 2 Jan Paul Pietrzak 3 Jan S‡gl 15 Jan Saudek 7 Jan van Steenwijk 76 Jane L. Calvin 14 Jane Lidz 5 Jane Regan 1 Jane Tuckerman 44 Jane W. Hudson 7 Janeil Engelstad 1 Janelle MortŽ 6 Janet Pritchard 6 Janis Croft 1 Janis Croft 1 Jann Kwasneski 3 Janyce Erlich-Moss 3 Jaroslav Benes 2 Jaroslav Razjik 15 Jason Simon 2 Javier Diaz Saiz 2	Jan Groover	4
Jan Otsen Jan Paul Pietrzak Jan S‡gl Jan Saudek Jan van Steenwijk Jane L. Calvin Jane Lidz Jane Regan Jane Tuckerman Jane W. Hudson Janeil Engelstad Janelle MortŽ Janet Pritchard Janice Bell Janis Croft Jann Kwasneski Janusz Lesniak Janyce Erlich-Moss Jaroslav Benes Jaroslav Razjik Jaschi Klein Jason Simon Jane Diez Saiz	Jan Hnizdo	35
Jan Paul Pietrzak Jan S‡gl Jan Saudek Jan van Steenwijk Jane L. Calvin Jane Lidz Jane Regan Jane Tuckerman Jane W. Hudson Janeil Engelstad Janelle MortŽ Janet Pritchard Janice Bell Janis Croft Jann Kwasneski Janyce Erlich-Moss Jaroslav Benes Jaroslav Razjik Jaschi Klein Jason Simon Javier Diaz Saiz	Jan Nemec	31
Jan S‡gl Jan Saudek Jan van Steenwijk Jane L. Calvin Jane Lidz Jane Regan Jane Tuckerman Jane W. Hudson Janeil Engelstad Janelle MortŽ Janet Pritchard Janice Bell Janis Croft Jann Kwasneski Janusz Lesniak Janyce Erlich-Moss Jaroslav Benes Jaroslav Razjik Jason Simon Javier Diaz Saiz	Jan Otsen	2
Jan Saudek Jan van Steenwijk Jane L. Calvin Jane L. Calvin Jane Regan Jane Regan Jane Tuckerman Jane W. Hudson Janeil Engelstad Janeile MortŽ Janet Pritchard Janice Bell Janis Croft Jann Kwasneski Janusz Lesniak Janyce Erlich-Moss Jaroslav Benes Jaroslav Razjik Jaschi Klein Jason Simon Javier Diaz Saiz	Jan Paul Pietrzak	3
Jan Saudek Jan van Steenwijk Jane L. Calvin Jane L. Calvin Jane Regan Jane Regan Jane Tuckerman Jane W. Hudson Janeil Engelstad Janeile MortŽ Janet Pritchard Janice Bell Janis Croft Jann Kwasneski Janusz Lesniak Janyce Erlich-Moss Jaroslav Benes Jaroslav Razjik Jaschi Klein Jason Simon Javier Diaz Saiz	Jan S‡gl	15
Jane L. Calvin 14 Jane Lidz 5 Jane Regan 1 Jane Tuckerman 44 Jane W. Hudson 7 Janeil Engelstad 1 Janelle MortŽ 6 Janice Bell 2 Janice Bell 2 Janis Croft 1 Jann Kwasneski 3 Janyce Erlich-Moss 3 Jaroslav Benes 2 Jaroslav Razjik 15 Jason Simon 2 Javier Diaz Saiz 2		7
Jane L. Calvin 14 Jane Lidz 5 Jane Regan 1 Jane Tuckerman 44 Jane W. Hudson 7 Janeil Engelstad 1 Janelle MortŽ 6 Janice Bell 2 Janice Bell 2 Janis Croft 1 Jann Kwasneski 3 Janyce Erlich-Moss 3 Jaroslav Benes 2 Jaroslav Razjik 15 Jason Simon 2 Javier Diaz Saiz 2	Jan van Steenwijk	76
Jane Regan Jane Tuckerman Jane W. Hudson Janeil Engelstad Janelle MortŽ Janet Pritchard Janice Bell Janis Croft Jann Kwasneski Janusz Lesniak Janyce Erlich-Moss Jaroslav Benes Jaroslav Razjik Jason Simon Javier Diaz Saiz		14
Jane Tuckerman Jane W. Hudson Janeil Engelstad Janelle MortŽ Janet Pritchard Janice Bell Janis Croft Jann Kwasneski Janusz Lesniak Janyce Erlich-Moss Jaroslav Benes Jaroslav Razjik Jaschi Klein Jason Simon Javier Diaz Saiz	Jane Lidz	5
Jane Tuckerman Jane W. Hudson Janeil Engelstad Janelle MortŽ Janet Pritchard Janice Bell Janis Croft Jann Kwasneski Janusz Lesniak Janyce Erlich-Moss Jaroslav Benes Jaroslav Razjik Jaschi Klein Jason Simon Javier Diaz Saiz	Jane Regan	1
Janeil Engelstad Janelle MortŽ Janet Pritchard Gell Janice Bell Janis Croft Jann Kwasneski Janusz Lesniak Janyce Erlich-Moss Jaroslav Benes Jaroslav Razjik Jaschi Klein Jason Simon Javier Diaz Saiz	_	44
Janelle MortŽ 6 Janet Pritchard 6 Janice Bell 2 Janis Croft 1 Jann Kwasneski 3 Janusz Lesniak 1 Janyce Erlich-Moss 3 Jaroslav Benes 2 Jaroslav Razjik 15 Jaschi Klein 8 Jason Simon 2 Javier Diaz Saiz 2	Jane W. Hudson	7
Janelle MortŽ 6 Janet Pritchard 6 Janice Bell 2 Janis Croft 1 Jann Kwasneski 3 Janusz Lesniak 1 Janyce Erlich-Moss 3 Jaroslav Benes 2 Jaroslav Razjik 15 Jaschi Klein 8 Jason Simon 2 Javier Diaz Saiz 2	Janeil Engelstad	1
Janice Bell Janis Croft Jann Kwasneski Janusz Lesniak Janyce Erlich-Moss Jaroslav Benes Jaroslav Razjik Jaschi Klein Jason Simon Javier Diaz Saiz		6
Janis Croft1Jann Kwasneski3Janusz Lesniak1Janyce Erlich-Moss3Jaroslav Benes2Jaroslav Razjik15Jaschi Klein8Jason Simon2Javier Diaz Saiz2	Janet Pritchard	6
Jann Kwasneski3Janusz Lesniak1Janyce Erlich-Moss3Jaroslav Benes2Jaroslav Razjik15Jaschi Klein8Jason Simon2Javier Diaz Saiz2	Janice Bell	2
Janusz Lesniak1Janyce Erlich-Moss3Jaroslav Benes2Jaroslav Razjik15Jaschi Klein8Jason Simon2Javier Diaz Saiz2	Janis Croft	1
Janyce Erlich-Moss3Jaroslav Benes2Jaroslav Razjik15Jaschi Klein8Jason Simon2Javier Diaz Saiz2	Jann Kwasneski	3
Jaroslav Benes2Jaroslav Razjik15Jaschi Klein8Jason Simon2Javier Diaz Saiz2	Janusz Lesniak	1
Jaroslav Razjik15Jaschi Klein8Jason Simon2Javier Diaz Saiz2	Janyce Erlich-Moss	3
Jaschi Klein8Jason Simon2Javier Diaz Saiz2	Jaroslav Benes	2
Jaschi Klein8Jason Simon2Javier Diaz Saiz2	Jaroslav Razjik	15
Jason Simon2Javier Diaz Saiz2		8
Javier Diaz Saiz		2
1 : 37 111	Javier Diaz Saiz	2
Javier Vailhornrat	Javier Vallhornrat	3
		23
		2
		5
		2
		1
		8

Artist	Approximate Photo Quantity
Jean Dieuzaide	19
Jean Lariviere	1
Jean Luc Mettoudi	1
Jean Meziere	8
Jean Mogerly	2
Jean Paul Debattice	1
Jean Pierre Sudre	1
Jean Pierre Vorlet	1
Jean Renard	8
Jean Ruiter	8
Jean-Claud DeWolf	8
Jean-Claude Pr\$ tre	11
Jean-Francois Bauret	14
Jean-Louis Losi	4
Jeanloup Sieff	10
Jean-Luc Koenig	5
Jeff Dunas	2
Jeff Millikan	8
Jeffrey Blackman	4
Jeffrey Coolidge	1
Jeffrey Cosley	12
Jeffrey Hudson	8
Jeffrey Kay	4
Jeffrey Silverthorne	43
Jennie Hayes	1
Jenny Lynn	2
Jerry Burchard	9
Jerry Burchfield	6
Jerry Uelsmann	18
Jerry Yalsman	2
Jesseca Ferguson	9
Jet Propulsion Lab	10
Jim Alinder	3
Jim Bengston	29
Jim Dine	4
Jim Enyeart	2
Jim Finlayson	2
Jim Goldberg	19
Jim Hughes	1
Jim Jipson	1
Jim Lee	3
Jim McHugh	10
Jim Stone	51
Jo Brunenberg	4
Jo Whaley	38
Joachim Bruno	1
Joachim WŸrfel	4

Artist	Approximate Photo Quantity
Joan Albert	1
Joan Fontcuberta	23
Joan Harrison	4
Joan Lyons	5
Joan Wiener	2
JoAnn Verburg	1
Jock Gill	43
Joe Oppedisano	1
Joel Janowitz	2
Joel Levick	6
Joel Meyerowitz	5
Joel Peter Witkin	1
John Atchley	10
John Batho	2
John Benson	58
John Brook	1
John Chervinsky	9
John Coplans	2
John D. Craig	27
John Divola	30
John F. Cooper	1
John Flattau	1
John Gillan	2
John Gintoff	21
John Gutowski	6
John Haines	1
John Hilgert	3
John Hill	2
John Kimmich-Javier	4
John LaFerlita	1
John Morrison	1
John Murphy	8
John Pfahl	7
John Reuter	100
John Schulze	4
John Schwartz	23
John Sexton	44
John Thornton	9
John Urban	14
John Wise	3
John Wolbarst	5
John Wood	13
Jolie Stahl	2
Jon Gordon	6
Jon Lewis	15
Jonas Westerman	3
Jonathan Dudley Ellis	1

Artist	Approximate Photo Quantity
Jonathan Goldman	1
Jonathan Sharlin	5
Jonathon Moller	1
Joni Carter	5
Joni Sternbach	3
Joop der Weduwen	1
Joost Guntenaar	4
Jorge Saenz	10
Jose Picayo	4
Josef Sudek	1
Joseph A. Gioia	6
Joseph De Maio	10
Joseph Evans	1
Joseph Saltzer	7
Joseph Scheer	1
Joseph Vitone	4
Joseph Ziolkowski	2
Josepha Haveman	9
Josephine Starrs/Leon Cmielewski	5
Joyce Culver	2
Joyce Neimanas	2
Joyce Ravid	1
Joyce Salloum	11
Joyce Tenneson	47
JSC	2
Judith Black	50
Judith Eglington	2
Judith Golden	61
Judith Thorpe	2
Judy Dater	15
Judy Dole	1
Judy Natal	6
Julia Hoersch	10
Julia Parker	7
Julio Mitchel	3
Julius Vitali	7
Jungo Shuden	4
Jurgen Tapprich	1
Jurriaan Eindhoven	2
JŸrg Andermatt	3
JŸrgen Wegner	1
K. L. Thurlbeck	17
Karel Appel	1
Karen Hymer-Thompson	2
Karen Kommer	8
Karen Pearce	5
Karen Solstad	3
Trainin Doibhid	3

Artist	Approximate Photo Quantity
Karen Tremaine	1
Karin ÒANDORAÓ Berndt	2
Karin Rosenthal	7
Karin Szekessy	10
Karina Barg	2
Karina Geideck	4
Karl Baden	4
Karl Heinz Soentgen	2
Kate Carter	1
Katherine Alling	7
Kathie Florsheim	30
Kathleen Campbell	3
Kathleen Kenyon	2
Kathleen Thormod Carr	3
Katinka Mann	4
Kaz Tsuchikawa	7
Kazumi Kurigami	4
Ke Francis	1
Kees Roosenboom	6
Kees Tillema	1
Keichi Tahara	3
Keith Collie	2
Kelly Wise	23
Ken Buck	3
Ken Kuenster	20
Ken Light	1
Ken Matsubara	8
Ken Straiton	4
Ken Williams	33
Kenario	1
Kenda North	30
Kenneth Laing Herdy	3
Kenneth Rush	2
Kenneth Shorr	8
Kenneth Sundh	1
Kersten Preussler	1
Kevin Bubriski	15
Kevin Mazur	27
Kevin Smith	1
Kim Mosley	2
Kim Stringfellow	6
Kishin Shinoyama	2
Klaus Schnitzer	31
Klaus Dieter Herrmann	9
Klaus Frahm	29
Klaus-Peter Nordmann	3
Kodo Tanaka	2

Artist	Approximate Photo Quantity
Koichi Nakazawa	1
Koldo Chamorro	10
Kristen Struebing-Beazley	4
Kristina Kachelle	1
Krzysztof Pruszkowski	3
Kunihiro Shinohara	4
Kurt Demmler	13
Kurt Markus	41
Lajos Keresztes	15
Lance Carlson	7
Larry Clark	1
Larry Davis	3
Larry Lawfer	10
Larry Sultan	5
Laura Blacklow	6
Laura Gilpin	4
Laura Hughes	2
Laura McPhee/Virginia Beahan	4
Lauren Shaw	1
Laurence Gartel	1
Laurie Simmons	4
Laurie Dietz	4
Laurie Seamans	10
Lawrence Huff	8
Lawrie Brown	34
Leandro Erlich	6
Lee Kraft	21
Leigh Ann Langwell	2
Len Gittleman	3
Leni Schwendinger	4
Lennart Nilsson	2
Leo Hilber	4
Leoni Reisberg	12
Les Krims	7
Leszek Brogowski	1
Lew Wilson	3
Lewis Hine	12
Lewis R. Toby	19
Lia Rothstein	1
Linda Robbennolt	26
Linda Benedict-Jones	1
Linda Broadfoot	4
Linda Burgess	8
Linda Connor	10
Linda Gunderson	3

Artist	Approximate Photo Quantity
Linda Lindroth	18
Linda Mahoney	11
Linda Rosenfield	1
Linda Rutenberg	3
Linda Troeller	13
Linda White	1
Lisa Gipton	2
Lisa SchŠublin	1
Lisa Stein	1
Lisa Tarchak	1
Lisette Model	2
Lissie Habie	2
Lloyd E. Moore	5
Lorie Novak	23
Lorna Simpson	13
Lorraine Dar Conte	3
Lorraine Gilbert	1
Lotte Jacobi	2
Lou Barranti	2
Lou Jones	7
Louise Shenstone	3
Lourdes Almeida	1
Luca Patella	2
Lucas Samaras	22
Lucco Pizzorno	1
Luciano Franchi de Alfaro III	88
Luciano Castelli	6
Luciano Montemurro	2
Luciano Ricci	5
Lucien Clergue	25
Lucille Feremans	2
Lucy Johnson	4
Luigi Ghirri	35
Luigi Vigliotti	1
Luke Pellatt	1
Luther Smith	36
Lyle Ashton Harris	8
Lyle Gomes	3
Lynn Butler	11
Lynn Johnson	4
Lynn Lennon	3
Lynn Weyland	3
M Naragawa	1
M Rumler	1
M.K. Berndt	1
M.M. Lum	1
Maarten Scheks	1

Artist	Approximate Photo Quantity
Maggie Kessel	1
Mags Harries	5
Malcolm S. Kirk	65
Maldwin Stride	1
Manfred Heiting	18
Man-Shin Yonfan	2
Manuel Alvarez Bravo	2
Marc Donnadieu	3
Marc Peverelli	6
Marc Pokempner	4
Marc Vanappelghem	4
Marcel Von Mentlen	3
Marco DÕAnna	2
Marco Misani	3
Marco Rigamonti	2
Marek Gardulski	11
Margaret Bourke-White	3
Margaretta Mitchell	8
Margarita Garcia Bunuel	1
Margery B. Franklin	1
Margot Pils	4
Maria Magdalena Campos-Pons	26
Maria Vodarich	4
Mariagrazia Federico	15
Marie Cosindas	4
	6
Marina Abramovic & Ulay	
Mario Beaudet	12
Mario De Biasi	12
Marius Boender	2
Marjorie Bal	2
Mark Berghash	2
Mark Chamberlain/Jerry Burchfield	6
Mark Cohen	10
Mark Cooper	28
Mark D. Thellmann	1
Mark Hooper	3
Mark Klett	66
Mark Lipson	5
Mark Morrisroe	4
Mark Orlove	5
Mark Power	11
Mark Ruwedel	13
Mark Schreyer	10
Markus Kirchgessner	2
Markus Roessle	4
Marlo Broekmans	12
Marsha Burns	36

Artist	Approximate Photo Quantity
Marshall	2
Martha Pearson	30
Martha Casanave	20
Martha Clark	1
Martien van Beeck	3
Martin Benjamin	3
Martin Cooper	6
Martin Pudenz	25
Martin Randourath	1
Martin Stupich	10
Marty Fumo	14
Marvin Heiferman	1
Mary A. Gawle	1
Mary Ann Nilsson	7
Mary Babb	1
Mary Ellen Carroll	6
Mary Ellen Mark	35
Mary Ross	2
Mary Sayer Hammond	4
Maryjean Viano Crowe	1
Marylu Raushenbush	1
Masahisa Fukase	2
Masami Mori	16
Masatoshi Naitoh	4
Mathijs van Spankeren	6
Mats G. H. Nordstrom	9
Maurice Heller	3
Maurice Smith	6
Maurizio Montanari	2
Max Kellenberger	18
Megan Jenkinson	3
Mel Goldman	2
Melanie Walker	18
Melina Deya	10
Melissa Shook	6
Melissa Tomich	3
Meroe Marston Morse	2
Micaela Garzoni	25
Michael D. Silver	
	10
Michael Brootte	5
Michael Becotte	7
Michael Belenky	4
Michael Bishop	16
Michael C. Barzda	1
Michael Cleary	3
Michael Cook	1
Michael DiBiase	15

Michael Doukas 1 Michael Eastman 2 Michael Geiger 6 Michael Hudock 7 Michael Hudock 7 Michael Kostink 7 Michael Kostink 7 Michael Kostink 7 Michael Labadie 3 Michael Philip Manheim 1 Michael McKelvey 5 Michael Philip Manheim 1 Michael Ryan 4 Michael Ryan 4<	Artist	Approximate Photo Quantity
Michael Forg 2 Michael Geiger 2 Michael Going 2 Michael Hammer-Rovner 6 Michael Hudock 7 Michael Kenna 1 Michael Koster 4 Michael Kostink 7 Michael Kostink 7 Michael Labadie 3 Michael Labides 1 Michael Lapides 1 Michael McKelvey 5 Michael McKelvey 5 Michael Northrup 2 Michael Philip Manheim 1 Michael Pierce 102 Michael Ryan 4 Michael Spano 6 Michael Ryan 4 Michael Spano 6 Michael Redon 1 Michele Baelus 2 Michele Baelus 2 Michel Redon 2 Michele Macrakis 2 2 4 Michele Bacuchard 1 Migele Battelle 2 <t< td=""><td>Michael Doukas</td><td>1</td></t<>	Michael Doukas	1
Michael Geiger 2 Michael Going 2 Michael Hammer-Rovner 6 Michael Hudock 7 Michael Kenna 1 Michael Koster 4 Michael Kostink 7 Michael Labadie 3 Michael Labides 1 Michael Lapides 1 Michael Ackelvey 5 Michael Northrup 2 Michael Northrup 2 Michael Philip Manheim 1 Michael Pierce 102 Michael Ryan 4 Michael Spano 6 Michael W. Davidson 1 Michel Baelus 2 Michel Baelus 2 Michel Baelus 2 Michel Redon 2 Michel Baetue 2 <	Michael Eastman	2
Michael Hammer-Rovner 6 Michael Hudock 7 Michael Kenna 1 Michael Koster 4 Michael Kostink 7 Michael Labadie 3 Michael Lapides 1 Michael McKelvey 5 Michael Northrup 2 Michael Philip Manheim 1 Michael Pierce 102 Michael Pierce 102 Michael Ryan 4 Michael Spano 6 Michael W. Davidson 1 Michel Baelus 2 Michel Medinger 24 Michel Redon 2 Michele Bouchard 1 Michele Bouchard 1 Michele Macrakis 2 Michiko Kon 4 Midge Battelle 2 Miguel Gandert 4 Miklo Matsuo 1 Milla Mareckova 1 Milla Mareckova 1 Millon Haberstadt 3 Millon Haberstadt	Michael Fong	2
Michael Hudock 7 Michael Kenna 1 Michael Koster 4 Michael Labadie 3 Michael Lapides 1 Michael Lapides 1 Michael Northrup 2 Michael Philip Manheim 1 Michael Pirece 102 Michael Ryan 4 Michael Spano 6 Michael W. Davidson 1 Michel Redon 2 Michel Redon 2 Michel Bouchard 1 Michele Macrakis 2 Michiko Kon 4 Midge Battelle 2 Miko Matsuo 1 Milko LeGrady 4 Mila Mareckova 1 Milton Greene 34 Milton Halberstadt 3 Milton Halberstadt 1 Mirray Bis 4 Morica Nestler von Rosen 11 Moriel Erman-Mandel 2 Murrel Olesen 5 Murray Van Dyke 3 </td <td>Michael Geiger</td> <td>2</td>	Michael Geiger	2
Michael Hudock 7 Michael Kenna 1 Michael Kolster 4 Michael Labadie 3 Michael Labadies 1 Michael McKelvey 5 Michael McKelvey 5 Michael Northrup 2 Michael Philip Manheim 1 Michael Pierce 102 Michael Ryan 4 Michael Spano 6 Michael W. Davidson 1 Michel Baelus 2 Michel Baelus 2 Michel Redon 2 Michel Redon 2 Michele Bouchard 1 Michele Macrakis 2 Michiko Kon 4 Midge Battelle 2 Miguel Gandert 4 Miklo Matsuo 1 Miklo Matsuo 1 Miklo Matsuo 1 Miller Levy 2 Milton Greene 34 Milton Halberstadt 1 Mimmo Laera 4 <t< td=""><td>Michael Going</td><td>2</td></t<>	Michael Going	2
Michael Kolster 4 Michael Kostink 7 Michael Labadie 3 Michael Lapides 1 Michael McKelvey 5 Michael Northrup 2 Michael Philip Manheim 1 Michael Pierce 102 Michael Ryan 4 Michael Spano 6 Michael W. Davidson 1 Michel Baelus 2 Michel Medinger 24 Michel Bedon 2 Michele Bouchard 1 Michele Macrakis 2 Michiko Kon 4 Midge Battelle 2 Mikio Matsuo 1 Miklo Matsuo 1 Miklo Matsuo 1 Miller Levy 2 Milton Greene 34 Milton Haberstadt 3 Milton Haberstadt 1 Mimmo Laera 4 Minor White 34 Mirella Eckhardt 2 Morica Nestler von Rosen 11	Michael Hammer-Rovner	6
Michael Kostink 7 Michael Labadie 3 Michael Lapides 1 Michael McKelvey 5 Michael Northrup 2 Michael Philip Manheim 1 Michael Pierce 102 Michael Ryan 4 Michael Spano 6 Michael W. Davidson 1 Michel Baelus 2 Michel Redon 2 Michel Bouchard 1 Michele Bouchard 1 Michele Macrakis 2 Michiko Kon 4 Midge Battelle 2 Miguel Gandert 4 Miklos LeGrady 4 Milton Matsuo 1 Milton Greene 34 Milton Haberstadt 3 Milton Halberstadt 1 Mimor White 34 Mirrella Eckhardt 2 Morica Nestler von Rosen 11 Morigan Thomas 5 Morley Baer 10 Murriel Ehrman-Mandel	Michael Hudock	7
Michael Labadie 3 Michael Lapides 1 Michael McKelvey 5 Michael Northrup 2 Michael Philip Manheim 1 Michael Philip Manheim 1 Michael Price 102 Michael Ryan 4 Michael Spano 6 Michael W. Davidson 1 Michael W. Davidson 1 Michel Baelus 2 Michel Medinger 24 Michel Redon 2 Michele Bouchard 1 Michele Macrakis 2 Michiko Kon 4 Midge Battelle 2 Miguel Gandert 4 Miklos LeGrady 4 Mila Mareckova 1 Milla Mareckova 1 Millon Greene 34 Milton Haberstadt 3 Milton Haberstadt 1 Mirano White 34 Mirano Lecra 4 Mirano Nestler von Rosen 11 Monique Jacot	Michael Kenna	1
Michael Lapides 1 Michael McKelvey 5 Michael Northrup 2 Michael Philip Manheim 1 Michael Pierce 102 Michael Ryan 4 Michael Spano 6 Michael W. Davidson 1 Michael Baelus 2 Michel Baelus 2 Michel Redon 2 Michele Bouchard 1 Michele Bouchard 1 Michele Macrakis 2 Michiko Kon 4 Midge Battelle 2 Miguel Gandert 4 Miko Matsuo 1 Miklo Mareckova 1 Milla Mareckova 1 Millon Greene 34 Milton Greene 34 Milton Halberstadt 1 Mirmon Laera 4 Mirol Backhardt 2 Miral Reckhardt 2 Morial Pierre 3 Morial Pierre 3 Morial Pierre 3 <td>Michael Kolster</td> <td>4</td>	Michael Kolster	4
Michael Lapides 1 Michael McKelvey 5 Michael Northrup 2 Michael Philip Manheim 1 Michael Pierce 102 Michael Ryan 4 Michael Ryan 4 Michael Spano 6 Michael W. Davidson 1 Michel Baelus 2 Michel Medinger 24 Michel Redon 2 Michel Bouchard 1 Michele Bouchard 1 Michele Macrakis 2 Michiko Kon 4 Midge Battelle 2 Miguel Gandert 4 Miklos LeGrady 4 Miklos LeGrady 4 Mila Mareckova 1 Miller Levy 2 Milton Haberstadt 3 Milton Haberstadt 3 Mirror Haberstadt 1 Mirror White 34 Mirror White 34 Mirror White 34 Monica Nestler von Rosen 11	Michael Kostink	7
Michael McKelvey 5 Michael Northrup 2 Michael Philip Manheim 1 Michael Pierce 102 Michael Ryan 4 Michael Spano 6 Michael W. Davidson 1 Michel Baelus 2 Michel Medinger 24 Michel Redon 2 Michele Bouchard 1 Michele Bouchard 1 Michele Macrakis 2 Michiko Kon 4 Midge Battelle 2 Miguel Gandert 4 Mikio Matsuo 1 Miklos LeGrady 4 Miller Levy 2 Miller Levy 2 Milton Greene 34 Milton Halberstadt 3 Milton Halberstadt 1 Mimmo Laera 4 Mimmo Laera 4 Miryako Ishiuchi 5 Monica Nestler von Rosen 11 Monique Jacot 10 Morjan Thomas 5	Michael Labadie	3
Michael Northrup 2 Michael Philip Manheim 1 Michael Pierce 102 Michael Ryan 4 Michael Spano 6 Michael W. Davidson 1 Michel Baelus 2 Michel Redon 2 Michel Redon 2 Michele Bouchard 1 Michele Macrakis 2 Michiko Kon 4 Midge Battelle 2 Miguel Gandert 4 Miklos LeGrady 4 Millos LeGrady 4 Miller Levy 2 Millon Greene 34 Milton Greene 34 Milton Halberstadt 3 Milton Halberstadt 1 Mimmo Laera 4 Minor White 34 Mirola Restler von Rosen 11 Monica Nestler von Rosen 11 Monique Jacot 10 Morgan Thomas 5 Morley Baer 2 Murriel Dlesen 5 </td <td>Michael Lapides</td> <td>1</td>	Michael Lapides	1
Michael Northrup 2 Michael Philip Manheim 1 Michael Pierce 102 Michael Ryan 4 Michael Spano 6 Michael W. Davidson 1 Michel Baelus 2 Michel Redon 2 Michel Redon 2 Michele Bouchard 1 Michele Macrakis 2 Michiko Kon 4 Midge Battelle 2 Miguel Gandert 4 Miklos LeGrady 4 Millos LeGrady 4 Miller Levy 2 Millon Greene 34 Milton Greene 34 Milton Halberstadt 3 Milton Halberstadt 1 Mimmo Laera 4 Minor White 34 Mirola Restler von Rosen 11 Monica Nestler von Rosen 11 Monique Jacot 10 Morgan Thomas 5 Morley Baer 2 Murriel Dlesen 5 </td <td>_</td> <td>5</td>	_	5
Michael Philip Manheim 1 Michael Pierce 102 Michael Ryan 4 Michael Spano 6 Michael W. Davidson 1 Michel Baelus 2 Michel Baelus 2 Michel Medinger 24 Michel Redon 2 Michele Bouchard 1 Michele Macrakis 2 Michiko Kon 4 Midge Battelle 2 Miguel Gandert 4 Mikio Matsuo 1 Miklos LeGrady 4 Mila Mareckova 1 Miller Levy 2 Milton Greene 34 Milton Halberstadt 3 Milton Halberstadt 1 Mimmo Laera 4 Minor White 34 Mirola Eckhardt 2 Miyako Ishiuchi 5 Monica Nestler von Rosen 11 Moriey Baer 2 Muriel Ehrman-Mandel 2 Murriel Olesen 5		
Michael Pierce 102 Michael Ryan 4 Michael Spano 6 Michael W. Davidson 1 Michel Baelus 2 Michel Medinger 24 Michel Redon 2 Michele Bouchard 1 Michele Macrakis 2 Michiko Kon 4 Midge Battelle 2 Miguel Gandert 4 Mikio Matsuo 1 Miklos LeGrady 4 Mila Mareckova 1 Miller Levy 2 Milton Greene 34 Milton Haberstadt 3 Milton Halberstadt 1 Mimmo Laera 4 Minor White 34 Mirella Eckhardt 2 Monica Nestler von Rosen 11 Morique Jacot 10 Morgan Thomas 5 Morley Baer 2 Muriel Ehrman-Mandel 2 Murriel Olesen 5 Murray Van Dyke 3 <td></td> <td>1</td>		1
Michael Spano 6 Michael W. Davidson 1 Michel Baelus 2 Michel Medinger 24 Michel Redon 2 Michele Bouchard 1 Michele Macrakis 2 Michiko Kon 4 Midge Battelle 2 Miguel Gandert 4 Mikio Matsuo 1 Miklos LeGrady 4 Mila Mareckova 1 Miller Levy 2 Milton Greene 34 Milton Haberstadt 3 Milton Halberstadt 1 Mimmo Laera 4 Miror White 34 Mirella Eckhardt 2 Miyako Ishiuchi 5 Monica Nestler von Rosen 11 Morgan Thomas 5 Morley Baer 2 Murriel Ehrman-Mandel 2 Murray Van Dyke 3	Michael Pierce	102
Michael Spano 6 Michael W. Davidson 1 Michel Baelus 2 Michel Medinger 24 Michel Redon 2 Michele Bouchard 1 Michele Macrakis 2 Michiko Kon 4 Midge Battelle 2 Miguel Gandert 4 Mikio Matsuo 1 Miklos LeGrady 4 Mila Mareckova 1 Miller Levy 2 Milton Greene 34 Milton Haberstadt 3 Milton Halberstadt 1 Mimmo Laera 4 Miror White 34 Mirella Eckhardt 2 Miyako Ishiuchi 5 Monica Nestler von Rosen 11 Morgan Thomas 5 Morley Baer 2 Murriel Ehrman-Mandel 2 Murray Van Dyke 3	Michael Ryan	4
Michael W. Davidson 1 Michel Baelus 2 Michel Medinger 24 Michel Redon 2 Michele Bouchard 1 Michele Macrakis 2 Michiko Kon 4 Midge Battelle 2 Miguel Gandert 4 Mikio Matsuo 1 Miklos LeGrady 4 Mila Mareckova 1 Miller Levy 2 Milton Greene 34 Milton Halberstadt 3 Milton Halberstadt 1 Mimmo Laera 4 Miror White 34 Mirella Eckhardt 2 Miyako Ishiuchi 5 Monica Nestler von Rosen 11 Monique Jacot 10 Morgan Thomas 5 Morley Baer 2 Muriel Ehrman-Mandel 2 Murray Van Dyke 3		6
Michel Medinger 24 Michel Redon 2 Michele Bouchard 1 Michele Macrakis 2 Michiko Kon 4 Midge Battelle 2 Miguel Gandert 4 Mikio Matsuo 1 Miklos LeGrady 4 Mila Mareckova 1 Miller Levy 2 Milton Greene 34 Milton Halberstadt 3 Milton Halberstadt 1 Mimmo Laera 4 Minor White 34 Mirella Eckhardt 2 Miyako Ishiuchi 5 Monica Nestler von Rosen 11 Morigan Thomas 5 Morley Baer 2 Muriel Ehrman-Mandel 2 Muriel Olesen 5 Murray Van Dyke 3		1
Michele Bouchard 1 Michele Bouchard 1 Michele Macrakis 2 Michiko Kon 4 Midge Battelle 2 Miguel Gandert 4 Mikio Matsuo 1 Miklos LeGrady 4 Mila Mareckova 1 Miller Levy 2 Milton Greene 34 Milton Haberstadt 3 Milton Halberstadt 1 Mimmo Laera 4 Miorella Eckhardt 2 Miyako Ishiuchi 5 Monica Nestler von Rosen 11 Monique Jacot 10 Morgan Thomas 5 Morley Baer 2 Muriel Ehrman-Mandel 2 Muriel Olesen 5 Murray Riss 4 Murray Van Dyke 3	Michel Baelus	2
Michele Bouchard 1 Michele Bouchard 1 Michele Macrakis 2 Michiko Kon 4 Midge Battelle 2 Miguel Gandert 4 Mikio Matsuo 1 Miklos LeGrady 4 Mila Mareckova 1 Miller Levy 2 Milton Greene 34 Milton Haberstadt 3 Milton Halberstadt 1 Mimmo Laera 4 Miorella Eckhardt 2 Miyako Ishiuchi 5 Monica Nestler von Rosen 11 Monique Jacot 10 Morgan Thomas 5 Morley Baer 2 Muriel Ehrman-Mandel 2 Muriel Olesen 5 Murray Riss 4 Murray Van Dyke 3	Michel Medinger	24
Michele Macrakis 2 Michiko Kon 4 Midge Battelle 2 Miguel Gandert 4 Mikio Matsuo 1 Miklos LeGrady 4 Mila Mareckova 1 Miller Levy 2 Milton Greene 34 Milton Haberstadt 3 Milton Halberstadt 1 Mimmo Laera 4 Minor White 34 Mirella Eckhardt 2 Miyako Ishiuchi 5 Monica Nestler von Rosen 11 Monique Jacot 10 Morgan Thomas 5 Morley Baer 2 Muriel Ehrman-Mandel 2 Muriel Olesen 5 Murray Riss 4 Murray Van Dyke 3		2
Michiko Kon 4 Midge Battelle 2 Miguel Gandert 4 Mikio Matsuo 1 Miklos LeGrady 4 Mila Mareckova 1 Miller Levy 2 Milton Greene 34 Milton Haberstadt 3 Milton Halberstadt 1 Mimmo Laera 4 Minor White 34 Mirella Eckhardt 2 Miyako Ishiuchi 5 Monica Nestler von Rosen 11 Monique Jacot 10 Morgan Thomas 5 Morley Baer 2 Muriel Ehrman-Mandel 2 Muriel Olesen 5 Murray Riss 4 Murray Van Dyke 3	Michele Bouchard	1
Midge Battelle 2 Miguel Gandert 4 Mikio Matsuo 1 Miklos LeGrady 4 Mila Mareckova 1 Miller Levy 2 Milton Greene 34 Milton Haberstadt 3 Milton Halberstadt 1 Mimmo Laera 4 Minor White 34 Mirella Eckhardt 2 Miyako Ishiuchi 5 Monica Nestler von Rosen 11 Monique Jacot 10 Morgan Thomas 5 Morley Baer 2 Muriel Ehrman-Mandel 2 Muriel Olesen 5 Murray Riss 4 Murray Van Dyke 3	Michele Macrakis	2
Miguel Gandert 4 Mikio Matsuo 1 Miklos LeGrady 4 Mila Mareckova 1 Miller Levy 2 Milton Greene 34 Milton Haberstadt 3 Milton Halberstadt 1 Mimmo Laera 4 Minor White 34 Mirella Eckhardt 2 Miyako Ishiuchi 5 Monica Nestler von Rosen 11 Monique Jacot 10 Morgan Thomas 5 Morley Baer 2 Muriel Ehrman-Mandel 2 Muriel Olesen 5 Murray Riss 4 Murray Van Dyke 3	Michiko Kon	4
Miguel Gandert 4 Mikio Matsuo 1 Miklos LeGrady 4 Mila Mareckova 1 Miller Levy 2 Milton Greene 34 Milton Haberstadt 3 Milton Halberstadt 1 Mimmo Laera 4 Minor White 34 Mirella Eckhardt 2 Miyako Ishiuchi 5 Monica Nestler von Rosen 11 Monique Jacot 10 Morgan Thomas 5 Morley Baer 2 Muriel Ehrman-Mandel 2 Muriel Olesen 5 Murray Riss 4 Murray Van Dyke 3	Midge Battelle	2
Mikio Matsuo 1 Miklos LeGrady 4 Mila Mareckova 1 Miller Levy 2 Milton Greene 34 Milton Haberstadt 1 Mimmo Laera 4 Minor White 34 Mirella Eckhardt 2 Miyako Ishiuchi 5 Monica Nestler von Rosen 11 Monique Jacot 10 Morgan Thomas 5 Morley Baer 2 Muriel Ehrman-Mandel 2 Muriel Olesen 5 Murray Riss 4 Murray Van Dyke 3		4
Mila Mareckova 1 Miller Levy 2 Milton Greene 34 Milton Haberstadt 3 Milton Halberstadt 1 Mimmo Laera 4 Minor White 34 Mirella Eckhardt 2 Miyako Ishiuchi 5 Monica Nestler von Rosen 11 Monique Jacot 10 Morgan Thomas 5 Morley Baer 2 Muriel Ehrman-Mandel 2 Muriel Olesen 5 Murray Riss 4 Murray Van Dyke 3		1
Mila Mareckova 1 Miller Levy 2 Milton Greene 34 Milton Haberstadt 3 Milton Halberstadt 1 Mimmo Laera 4 Minor White 34 Mirella Eckhardt 2 Miyako Ishiuchi 5 Monica Nestler von Rosen 11 Monique Jacot 10 Morgan Thomas 5 Morley Baer 2 Muriel Ehrman-Mandel 2 Muriel Olesen 5 Murray Riss 4 Murray Van Dyke 3	Miklos LeGrady	4
Milton Greene 34 Milton Haberstadt 1 Mimmo Laera 4 Minor White 34 Mirella Eckhardt 2 Miyako Ishiuchi 5 Monica Nestler von Rosen 11 Monique Jacot 10 Morgan Thomas 5 Morley Baer 2 Muriel Ehrman-Mandel 2 Muriel Olesen 5 Murray Riss 4 Murray Van Dyke 3		1
Milton Greene 34 Milton Haberstadt 1 Mimmo Laera 4 Minor White 34 Mirella Eckhardt 2 Miyako Ishiuchi 5 Monica Nestler von Rosen 11 Monique Jacot 10 Morgan Thomas 5 Morley Baer 2 Muriel Ehrman-Mandel 2 Muriel Olesen 5 Murray Riss 4 Murray Van Dyke 3	Miller Levy	2
Milton Halberstadt 1 Mimmo Laera 4 Minor White 34 Mirella Eckhardt 2 Miyako Ishiuchi 5 Monica Nestler von Rosen 11 Monique Jacot 10 Morgan Thomas 5 Morley Baer 2 Muriel Ehrman-Mandel 2 Muriel Olesen 5 Murray Riss 4 Murray Van Dyke 3		34
Milton Halberstadt 1 Mimmo Laera 4 Minor White 34 Mirella Eckhardt 2 Miyako Ishiuchi 5 Monica Nestler von Rosen 11 Monique Jacot 10 Morgan Thomas 5 Morley Baer 2 Muriel Ehrman-Mandel 2 Muriel Olesen 5 Murray Riss 4 Murray Van Dyke 3	Milton Haberstadt	3
Minor White 34 Mirella Eckhardt 2 Miyako Ishiuchi 5 Monica Nestler von Rosen 11 Monique Jacot 10 Morgan Thomas 5 Morley Baer 2 Muriel Ehrman-Mandel 2 Muriel Olesen 5 Murray Riss 4 Murray Van Dyke 3		
Mirella Eckhardt 2 Miyako Ishiuchi 5 Monica Nestler von Rosen 11 Monique Jacot 10 Morgan Thomas 5 Morley Baer 2 Muriel Ehrman-Mandel 2 Muriel Olesen 5 Murray Riss 4 Murray Van Dyke 3	Mimmo Laera	4
Mirella Eckhardt 2 Miyako Ishiuchi 5 Monica Nestler von Rosen 11 Monique Jacot 10 Morgan Thomas 5 Morley Baer 2 Muriel Ehrman-Mandel 2 Muriel Olesen 5 Murray Riss 4 Murray Van Dyke 3	Minor White	34
Miyako Ishiuchi 5 Monica Nestler von Rosen 11 Monique Jacot 10 Morgan Thomas 5 Morley Baer 2 Muriel Ehrman-Mandel 2 Muriel Olesen 5 Murray Riss 4 Murray Van Dyke 3		
Monica Nestler von Rosen 11 Monique Jacot 10 Morgan Thomas 5 Morley Baer 2 Muriel Ehrman-Mandel 2 Muriel Olesen 5 Murray Riss 4 Murray Van Dyke 3		
Monique Jacot 10 Morgan Thomas 5 Morley Baer 2 Muriel Ehrman-Mandel 2 Muriel Olesen 5 Murray Riss 4 Murray Van Dyke 3		
Morgan Thomas5Morley Baer2Muriel Ehrman-Mandel2Muriel Olesen5Murray Riss4Murray Van Dyke3	Monique Jacot	
Morley Baer 2 Muriel Ehrman-Mandel 2 Muriel Olesen 5 Murray Riss 4 Murray Van Dyke 3		
Muriel Ehrman-Mandel2Muriel Olesen5Murray Riss4Murray Van Dyke3		
Muriel Olesen5Murray Riss4Murray Van Dyke3	_	
Murray Riss4Murray Van Dyke3		
Murray Van Dyke 3		
, ,		

Artist	Approximate Photo Quantity
Nancy Burson	19
Nancy Engel	11
Nancy Goldring	16
Nancy Honey	1
Nancy M Stuart	1
Nanna Bisp BŸchert	1
NASA	34
Natale Zoppis	2
Natalie Sudman	6
Nathan Farb	6
Neal Rantoul	8
Neal Slavin	76
Neil Newton	1
Nicholas Callaway	2
Nicholas Dean	152
Nick Price	2
Nick Sinclair	6
Nico Hoebe	1
Nicolas Bossard	1
Nigel Poor	2
Niki Berg	3
Nina Tovish	2
Nino Mascardi	1
Nino Migliori	6
Nobuyuki Tabata	2
Non-Polaroid Publication	4
Norah Delaney	2
Norbert Guthier	4
Norman A. Aborjaily	1
Norman Dow	4
Norman Locks	3
Nubar Alexanian	14
Nuriel Guedalia	2
Odile Moulinas	4
Oliver Lwowski	6
Olivia Parker	137
Oliviero Toscani	7
Other	6
P. H. Oelman	12
Palko de Loor	1
Pamela Ann Housman	4
Pamela Bannos	7
Pamela Ellis Hawkes	7
Paola Garzoni	1
Paolo Di Bello	2
Paolo Gioli	17
Pat Friedman	6
rai ffleuman	6

Artist	Approximate Photo Quantity
Pat Kurs	2
Patricia Carroll	5
Patricia McPheron	2
Patricia Sener	3
Patrick Nagatani	98
Patrick Saad	2
Patrick McPhearson	1
Patti Bose	2
Paul de Nooijer	10
Paul A. Beauchemin	2
Paul Berger	4
Paul Camello	1
Paul Caponigro	671
Paul D. Andersen	13
Paul du Pont	2
Paul Huf	39
Paul Jeremias	1
Paul Laster	4
Paul Lipscombe/Laura Wellington	5
Paul Ouazan	6
Paul Strand	20
Paule Muxel/Bertrand DeSolliers	3
Paulo Gasparini	5
Pavel Banka	21
Paz Erraruiz	2
Pedro Uhart	7
Peer Bode	1
Peg Tracy	2
Peggy Hartzell	5
Pelle Cass	12
Pepi Merisio	19
Pere Formiguera	4
Pete Turner	44
Peter Beard	9
Peter Busselat	2
Peter Caroff	10
Peter Charuk	4
Peter De Lory	21
Peter Droege	21
Peter Engblom	1
Peter Feldstein	9
Peter Goin	6
Peter Gowland	13
Peter Hiesserer	13
	19
Peter Hunsberger Peter Johnson	
Peter Johnson	13

Artist	Approximate Photo Quantity
Peter Jones	9
Peter Laytin	24
Peter MacGill	2
Peter Misisco	4
Peter Molle	1
Peter Neumann-Witt	3
Peter Robert Thompson	4
Peter S‡ghy	1
Peter Salek	3
Peter Schlessinger	4
Peter Tischhauser	2
Peter Uschner	6
Peter Whyte	55
Phil Porcella	1
Phil Greene/Gerry Sharpe	15
Phil in Phlash	11
Phil Zimmerman	3
Philip Bekker	6
Philip Hyde	10
	10
Philip MacMillan-James	1
Philip Makanna	1
Philip Trager	5
Philippe Verm\$ s	12
Philippe Blache	3
Philippe Halsman	198
Philippe Jeantet	1
Phillip James	1
PHOTOGRAPHER UNKNOWN	47
PHOTOMICRO	102
Phyllis Galembo	17
Pier Paolo Bianda	2
Pierre de Fenoyl	3
Pierre Keller	3
Pierre Mercier	3
Pierre-Jean Amar	1
Pierre-Louis Martin	6
Pietro Privitera	11
Pinky Bass	5
Pirkle Jones	8
Poirier	12
Polaroid Archives	33
Polaroid Historical	3
Polly Brown	6
R. Brad Knipstein	1
Rainer Griese	23
	25
Ralph Gibson	

Artist	Approximate Photo Quantity
Randolph L. Rill	1
Randolph Langenbach	2
Randy Malkin	1
Ray Belcher	40
Ray McSavaney	23
Rebecca Sexton Larson	2
Regina Stella Fonseca de Moraes	1
Rei Taka	5
Reid Callanan	4
Reinhart Wolf	37
Rena Small	54
Rene Všlker	4
Reza J. Khatir	11
Richard Averitt	1
Richard Benson	14
Richard Chase Albertine	8
Richard Doherty	12
Richard Edelman	23
Richard Faller	13
Richard Garrod	5
Richard Hamilton	2
Richard J. Linke	15
Richard Kennedy	2
Richard L. (Mac) McCowan	1
Richard M. Feldman	4
	2
Richard Margolis Richard Pare	3
Richard Pearson	2
	3
Richard Procopio Richard Twarog	10
Rick Hock	
	8
Rien Bazen	
Rita DeWitt	17
Rita Dibert	13
Rita Kohmann	5
Rob Goldman	6
Robb Buitenman	8
Robert Wilson	5
Robert A. Ripps	1
Robert Carr	4
Robert Cumming	10
Robert Fichter	9
Robert Frank	6
Robert Glenn Ketchum	2
Robert Haiko	87
Robert Heinecken	47
Robert Jonas	25

Artist	Approximate Photo Quantity
Robert Karosis	6
Robert Kolbrener	20
Robert Lyons	2
Robert Mapplethorpe	12
Robert McClintock	2
Robert Miller	3
Robert Pollock	2
Robert Rauschenberg	10
Robert Schuler	5
Robert Siegelman	6
Robert Solywoda	4
Robert Tennenbaum	1
Roberta M. Graham	6
Roberto Di Vincenzo	6
Roberto Gatti	2
Roberto Salas	4
Robin Bowman	8
Robin Dru Germany	5
Robyn Stacey	8
Roel	2
Roel Jacobs	1
Roger Mertin	12
Roger Wagner	6
Roland MŸller	4
Rolf Bettner	4
Rolf Braun-Eis	1
Rolf Magener	2
Romano Segna	4
Ron Cooper	2
Ron Goudsmit	3
Ron LaFon	4
Ron Levine	2
Ron Oldfield	1
Ron Rosenstock	7
Ron Yablon	1
Ronald Larned	3
Rosalyn Gerstein	2
Rosamond Wolff Purcell	194
Rosanne Olson	1
Rosario Mazzeo	3
Rose Marasco	1
Rosemary Nero	4
Roswell Angier	8
Roy Hanta	1
Rudi Sebastian	3
Rudolf Bonvie	14
Russell Hart	5
11000011 11011	5

Artist	Approximate Photo Quantity
Rutger ten Broeke	21
Ruth-Marion Baruch	6
Ruud Bom	2
Ruud van der Noord	1
Ryan Weideman	1
Ryodino Debuchi	1
S. Hawley Appleton	2
Sachiko Kuru	21
Sahin Kaygun	13
Sally Mann	3
Salvatore Mancini	6
Sam Haskins	5
Sam Unger	5
Samuel Laundon	3
Sandi Fellman	122
Sandra Stark	3
Sandro Oramas	7
Sandy Johanson	4
Sarah Benham	1
Sarah Hood	4
Sarah Moon	8
Sarah Scrutton	1
Schroten Schreiber	2
Sean Kernan	8
Sepp von Mentlen	7
Sepp Thiele	3
Sergio Oriani	3
Sergio Tornaghi	2
Share Corsaut	1
Shari Nathanson Rosenbloom	1
Sharon Fox	2
Sharon Smith	25
Sheila Metzner	3
Sheila Rock	2
Shelby Lee Adams	14
Shellburne Thurber	11
Sheri Lynn Behr	6
Sheva Fruitman	2
Shyla Irving	2
Siegfried Halus	11
Silvano Maggi	11
Silvia Taccani	2
Silvie & Cherif Defraoui	6
Silvio Wolf	2
Skeet McAuley	9
Slobodan Dimitrov	11
Sonja Reiger	4

Artist	Approximate Photo Quantity
Spectra	10
Spel Man	2
Stan Gaz	3
Stanley Rowin	5
Starr Ockenga	61
Stefan Roloff	2
Stefani Thornton	2
Stephan Whealton	79
Stephan (SAKT) Troxler	14
Stephan Erfurt	9
Stephan Krahwinkel	1
Stephen Berens	3
Stephen Chalmers	2
Stephen Frailey	16
Stephen Gersh	53
Stephen L. Feldman	4
Stephen Lowry	1
Stephen M. Schaub	4
Stephen Petegorsky	11
Stephen R Milanowski	5
Stephen Roach	7
Stephen Shore	10
Stephen Tourlentes	4
Steve Cohen	2
Steve Collins	3
Steve Grohe	1
Steve Knapp	1
Steve Marsel	1
Steve Minkowitz	4
Steve Nozicka	4
Steve Speller	1
Steve Wilson	1
Steven Barrett	1
Steven Goldman	1
Steven Hirsch	13
Steven M. Morrison	5
StŽphane Coutelle	5
Suda House	4
Sue Abramson	4
Sue Arrowsmith	7
Suellen Symons	9
Susan Barron	2
Susan Corey	4
Susan Orr	2
Susan Rose Channing	1
Susan Ross	24
Susan Unterberg	2

Susane Pastor 3 Suzane Opton 3 SX(P)70 Gang 4 Sylvia Klein 4 Tadanori Yokoo 4 Tahir Un 3 Takanobu Hayashi 6 Talli Rosner-Kozuch 3 Tammara Kaida 5 Tammy Mercure 4 Tana Kaleya 22 Tani Capa 1 Ted Worrison 3 Ted Orland 7 Terri Lindbloom 1 Terry Walker 51 Thea Weiss Storz 1 Theo Windges 23 Theresa Airey 2 Thomas Brennan 2 Thomas E. Lindfors 5 Thomas F. Barrow 6 Thomas Harris 6 Thomas Harris 6 Thomas League 4 Thomas League 4 Thomas League 4 Tim Gallah 5 Tim Kathy Church 1 Tim Calahan <th>Artist</th> <th>Approximate Photo Quantity</th>	Artist	Approximate Photo Quantity
SX(P)70 Gang 4 Sylvia Klein 4 Tadanori Yokoo 4 Talir Un 3 Takanobu Hayashi 6 Talli Rosner-Kozuch 3 Tammara Kaida 5 Tammy Mercure 4 Tana Kaleya 22 Tani Capa 1 Ted Morrison 3 Ted Orland 7 Terri Lindbloom 1 Terry Walker 51 Thea Weiss Storz 1 Theo Windges 23 Theresa Airey 2 Thomas Brennan 2 Thomas F. Barrow 6 Thomas Harris 6 Thomas Huther 12 Thomas League 4 Thomas League 4 Thomas S. Young 13 Tim & Kathy Church 1 Tim Callahan 2 Tim Kopacz/Robert Hamil 3 Tim Porter 4 Tim Orty Moyers 8 Ti	Susane Pastor	
Sylvia Klein 4 Tadanori Yokoo 4 Tahir Un 3 Takanobu Hayashi 6 Talli Rosner-Kozuch 3 Tammara Kaida 5 Tammy Mercure 4 Tana Kaleya 22 Tani Capa 1 Ted Orland 7 Terri Lindbloom 1 Terry Walker 51 Thea Weiss Storz 1 Thee Windges 23 Theresa Airey 2 Thomas Henra 2 Thomas Bernnan 2 Thomas F. Lindfors 5 Thomas Harris 6 Thomas Huther 12 Thomas League 4 Thomas League 4 Thomas S. Young 13 Tim Gallahan 2 Tim Gidal 5 Tim Kopacz/Robert Hamil 3 Tim Opocz/Robert Hamil 3 Timothy Moyers 8 Timothy White 6 <	Suzanne Opton	3
Tadanori Yokoo 4 Tahir Un 3 Takali Rosner-Kozuch 3 Tammara Kaida 5 Tammy Mercure 4 Tana Kaleya 22 Tani Capa 1 Ted Morrison 3 Ted Orland 7 Terri Lindbloom 1 Terry Walker 51 Thea Weiss Storz 1 Theo Windges 23 Theresa Airey 2 Thomas Brennan 2 Thomas E. Lindfors 5 Thomas F. Barrow 6 Thomas Hurther 12 Thomas League 4 Thomas S. Young 13 Tim & Callahan 2 Tim Gidal 5 Tim Godal 3 Tim Kopacz/Robert Hamil 3 Tim Opoczer 8 Timothy Moyers 8 Timothy White 6 Tino Desconsolul Ticker 6 Todo Oldham 1 <td< td=""><td>SX(P)70 Gang</td><td>4</td></td<>	SX(P)70 Gang	4
Tahir Un 3 Takanobu Hayashi 6 Talli Rosner-Kozuch 3 Tammara Kaida 5 Tammy Mercure 4 Tana Kaleya 22 Tani Capa 1 Ted Morrison 3 Ted Orland 7 Terri Lindbloom 1 Terry Walker 51 Thea Weiss Storz 1 Theo Windges 23 Theresa Airey 2 Thomas Brennan 2 Thomas F. Barrow 6 Thomas Huther 12 Thomas Huther 12 Thomas League 4 Thomas S. Young 13 Tim & Kathy Church 1 Tim Gidal 5 Tim Gopacz/Robert Hamil 3 Tim Orter 4 Timothy Moyers 8 Timothy White 6 Tiziana De Silvestro Hirschi 3 Toba Tucker 6 Toby Old 2 <	Sylvia Klein	4
Talkanobu Hayashi 6 Talli Rosner-Kozuch 3 Tammara Kaida 5 Tammy Mercure 4 Tana Kaleya 22 Tani Capa 1 Ted Morrison 3 Ted Orland 7 Terri Lindbloom 1 Terry Walker 51 Thea Weiss Storz 1 Theo Windges 23 Theresa Airey 2 Thomas Brennan 2 Thomas F. Barrow 6 Thomas Harris 6 Thomas Huther 12 Thomas Huther 12 Thomas League 4 Thomas S. Young 13 Tim & Kathy Church 1 Tim Callahan 2 Tim Kopacz/Robert Hamil 3 Tim Kopacz/Robert Hamil 3 Tim Othy Moyers 8 Timothy White 6 Toba Tucker 6 Toby Old 2 Todd Oldham 1	Tadanori Yokoo	4
Talli Rosner-Kozuch 3 Tammara Kaida 5 Tammy Mercure 4 Tana Kaleya 22 Tani Capa 1 Ted Morrison 3 Ted Orland 7 Terri Lindbloom 1 Terry Walker 51 Thea Weiss Storz 1 Theo Windges 23 Theresa Airey 2 Thomas Brennan 2 Thomas F. Barrow 6 Thomas Harris 6 Thomas Huther 12 Thomas League 4 Thomas League 4 Thomas S. Young 13 Tim & Kathy Church 1 Tim Gidal 5 Tim Kopacz/Robert Hamil 3 Tim Orter 4 Timothy Greenfield-Sanders 19 Timothy White 6 Tima Petra 1 Tito Simboli 13 Tiziana De Silvestro Hirschi 3 Toba Tucker 6	Tahir Un	3
Tammara Kaida 5 Tammy Mercure 4 Tana Kaleya 22 Tani Capa 1 Ted Morrison 3 Ted Orland 7 Terri Lindbloom 1 Terry Walker 51 Thea Weiss Storz 1 Theo Windges 23 Theresa Airey 2 Thomas Brennan 2 Thomas E. Lindfors 5 Thomas F. Barrow 6 Thomas Harris 6 Thomas Huther 12 Thomas League 4 Thomas League 4 Thomas S. Young 13 Tim & Kathy Church 1 Tim Gidal 5 Tim Kopacz/Robert Hamil 3 Tim Orter 4 Timothy Moyers 8 Timothy White 6 Tima Petra 1 Tito Simboli 13 Tiziana De Silvestro Hirschi 3 Toba Tucker 6	Takanobu Hayashi	6
Tammy Mercure 4 Tana Kaleya 22 Tani Capa 1 Ted Morrison 3 Ted Orland 7 Terri Lindbloom 1 Terry Walker 51 Thea Weiss Storz 1 Theo Windges 23 Theresa Airey 2 Thomas Brennan 2 Thomas E. Lindfors 5 Thomas F. Barrow 6 Thomas Harris 6 Thomas Huther 12 Thomas League 4 Thomas S. Young 13 Tim & Kathy Church 1 Tim Callahan 2 Tim Gidal 5 Tim Kopacz/Robert Hamil 3 Tim Orter 4 Timothy Moyers 8 Timothy White 6 Timothy White 6 Tima Petra 1 Tito Simboli 13 Tiziana De Silvestro Hirschi 3 Todd Oldham 1	Talli Rosner-Kozuch	3
Tana Kaleya 22 Tani Capa 1 Ted Morrison 3 Ted Orland 7 Terri Lindbloom 1 Terry Walker 51 Thea Weiss Storz 1 Theo Windges 23 Theresa Airey 2 Thomas Brennan 2 Thomas E. Lindfors 5 Thomas F. Barrow 6 Thomas Harris 6 Thomas Huther 12 Thomas League 4 Thomas S. Young 13 Tim & Kathy Church 1 Tim Callahan 2 Tim Gidal 5 Tim Kopacz/Robert Hamil 3 Tim Orter 4 Timothy Moyers 8 Timothy White 6 Timothy White 6 Timothy White 6 Toba Tucker 6 Toba Ukler 5 Todd Oldham 1 Todd Oldham 1 Tom Drahos	Tammara Kaida	5
Tana Kaleya 22 Tani Capa 1 Ted Morrison 3 Ted Orland 7 Terri Lindbloom 1 Terry Walker 51 Thea Weiss Storz 1 Theo Windges 23 Theresa Airey 2 Thomas Brennan 2 Thomas E. Lindfors 5 Thomas F. Barrow 6 Thomas Harris 6 Thomas Huther 12 Thomas League 4 Thomas S. Young 13 Tim & Kathy Church 1 Tim Callahan 2 Tim Gidal 5 Tim Kopacz/Robert Hamil 3 Tim Orter 4 Timothy Moyers 8 Timothy White 6 Timothy White 6 Timothy White 6 Toba Tucker 6 Toba Ukler 5 Todd Oldham 1 Todd Oldham 1 Tom Drahos	Tammy Mercure	4
Ted Morrison 3 Ted Orland 7 Terri Lindbloom 1 Terry Walker 51 Thea Weiss Storz 1 Theo Windges 23 Theresa Airey 2 Thomas Brennan 2 Thomas E. Lindfors 5 Thomas F. Barrow 6 Thomas Harris 6 Thomas Huther 12 Thomas League 4 Thomas S. Young 13 Tim & Kathy Church 1 Tim Callahan 2 Tim Gidal 5 Tim Kopacz/Robert Hamil 3 Tim Kopacz/Robert Hamil 3 Tim Porter 4 Timothy Greenfield-Sanders 19 Timothy Moyers 8 Timothy White 6 Tinz Petra 1 Tito Simboli 13 Tiziana De Silvestro Hirschi 3 Tobd Oldham 1 Todd Oldham 1 Todd Oldham 1		22
Ted Morrison 3 Ted Orland 7 Terri Lindbloom 1 Terry Walker 51 Thea Weiss Storz 1 Theo Windges 23 Theresa Airey 2 Thomas Brennan 2 Thomas E. Lindfors 5 Thomas F. Barrow 6 Thomas Harris 6 Thomas Huther 12 Thomas League 4 Thomas S. Young 13 Tim & Kathy Church 1 Tim Callahan 2 Tim Gidal 5 Tim Kopacz/Robert Hamil 3 Tim Kopacz/Robert Hamil 3 Tim Porter 4 Timothy Greenfield-Sanders 19 Timothy Moyers 8 Timothy White 6 Tinz Petra 1 Tito Simboli 13 Tiziana De Silvestro Hirschi 3 Tobd Oldham 1 Todd Oldham 1 Todd Oldham 1	-	1
Terri Lindbloom 1 Terry Walker 51 Thea Weiss Storz 1 Theo Windges 23 Theresa Airey 2 Thomas Brennan 2 Thomas E. Lindfors 5 Thomas F. Barrow 6 Thomas Harris 6 Thomas Huther 12 Thomas League 4 Thomas S. Young 13 Tim & Kathy Church 1 Tim Gidal 5 Tim Kopacz/Robert Hamil 3 Tim Forter 4 Timothy Greenfield-Sanders 19 Timothy Moyers 8 Timothy White 6 Tina Petra 1 Tito Simboli 13 Tiziana De Silvestro Hirschi 3 Toba Tucker 6 Toby Old 2 Todd Oldham 1 Todd Walker 5 Tohru Nakayama 1 Tom Drahos 3 Tom Hammang Jr. 14	_	3
Terry Walker 51 Thea Weiss Storz 1 Theo Windges 23 Theresa Airey 2 Thomas Brennan 2 Thomas E. Lindfors 5 Thomas F. Barrow 6 Thomas Huther 12 Thomas League 4 Thomas S. Young 13 Tim & Kathy Church 1 Tim Gidal 5 Tim Gidal 5 Tim Kopacz/Robert Hamil 3 Tim Porter 4 Timothy Greenfield-Sanders 19 Timothy Moyers 8 Timothy White 6 Tina Petra 1 Tito Simboli 13 Tiziana De Silvestro Hirschi 3 Toba Tucker 6 Todd Oldham 1 Todd Walker 5 Tohru Nakayama 1 Tom Drahos 3 Tom Hammang Jr. 14 Tom Jones 2	Ted Orland	7
Thea Weiss Storz 1 Theo Windges 23 Theresa Airey 2 Thomas Brennan 2 Thomas E. Lindfors 5 Thomas F. Barrow 6 Thomas Huther 12 Thomas Huther 12 Thomas League 4 Thomas S. Young 13 Tim & Kathy Church 1 Tim Gidal 5 Tim Kopacz/Robert Hamil 3 Tim Forter 4 Timothy Greenfield-Sanders 19 Timothy Moyers 8 Timothy White 6 Tina Petra 1 Tito Simboli 13 Tiziana De Silvestro Hirschi 3 Toba Tucker 6 Toby Old 2 Todd Oldham 1 Todd Walker 5 Tohru Nakayama 1 Tom Drahos 3 Tom Hammang Jr. 14 Tom Jones 2	Terri Lindbloom	1
Thea Weiss Storz 1 Theo Windges 23 Theresa Airey 2 Thomas Brennan 2 Thomas E. Lindfors 5 Thomas F. Barrow 6 Thomas Huther 12 Thomas Huther 12 Thomas League 4 Thomas S. Young 13 Tim & Kathy Church 1 Tim Gidal 5 Tim Kopacz/Robert Hamil 3 Tim Forter 4 Timothy Greenfield-Sanders 19 Timothy Moyers 8 Timothy White 6 Tina Petra 1 Tito Simboli 13 Tiziana De Silvestro Hirschi 3 Toba Tucker 6 Toby Old 2 Todd Oldham 1 Todd Walker 5 Tohru Nakayama 1 Tom Drahos 3 Tom Hammang Jr. 14 Tom Jones 2	Terry Walker	51
Theresa Airey 2 Thomas Brennan 2 Thomas E. Lindfors 5 Thomas F. Barrow 6 Thomas Harris 6 Thomas Huther 12 Thomas League 4 Thomas S. Young 13 Tim & Kathy Church 1 Tim Callahan 2 Tim Gidal 5 Tim Kopacz/Robert Hamil 3 Tim Porter 4 Timothy Greenfield-Sanders 19 Timothy Moyers 8 Timothy White 6 Tina Petra 1 Tito Simboli 13 Tiziana De Silvestro Hirschi 3 Toba Tucker 6 Toby Old 2 Todd Oldham 1 Todd Walker 5 Tohru Nakayama 1 Tom Ang 2 Tom Drahos 3 Tom Hammang Jr. 14 Tom Jones 2		1
Theresa Airey 2 Thomas Brennan 2 Thomas E. Lindfors 5 Thomas F. Barrow 6 Thomas Harris 6 Thomas Huther 12 Thomas League 4 Thomas S. Young 13 Tim & Kathy Church 1 Tim Callahan 2 Tim Gidal 5 Tim Kopacz/Robert Hamil 3 Tim Porter 4 Timothy Greenfield-Sanders 19 Timothy Moyers 8 Timothy White 6 Tina Petra 1 Tito Simboli 13 Tiziana De Silvestro Hirschi 3 Toba Tucker 6 Toby Old 2 Todd Oldham 1 Todd Walker 5 Tohru Nakayama 1 Tom Ang 2 Tom Drahos 3 Tom Hammang Jr. 14 Tom Jones 2	Theo Windges	23
Thomas Brennan 2 Thomas E. Lindfors 5 Thomas F. Barrow 6 Thomas Harris 6 Thomas Huther 12 Thomas League 4 Thomas S. Young 13 Tim & Kathy Church 1 Tim Callahan 2 Tim Gidal 5 Tim Kopacz/Robert Hamil 3 Tim Porter 4 Timothy Greenfield-Sanders 19 Timothy Moyers 8 Timothy White 6 Tina Petra 1 Tito Simboli 13 Tiziana De Silvestro Hirschi 3 Toba Tucker 6 Toby Old 2 Todd Oldham 1 Todd Walker 5 Tohru Nakayama 1 Tom Ang 2 Tom Drahos 3 Tom Hammang Jr. 14 Tom Jones 2	_	2
Thomas F. Barrow 6 Thomas Huther 12 Thomas League 4 Thomas S. Young 13 Tim & Kathy Church 1 Tim Callahan 2 Tim Gidal 5 Tim Kopacz/Robert Hamil 3 Tim Porter 4 Timothy Greenfield-Sanders 19 Timothy Woyers 8 Timothy White 6 Tina Petra 1 Tito Simboli 13 Tiziana De Silvestro Hirschi 3 Toba Tucker 6 Toby Old 2 Todd Oldham 1 Todd Walker 5 Tohru Nakayama 1 Tom Ang 2 Tom Drahos 3 Tom Hammang Jr. 14 Tom Jones 2	-	2
Thomas Harris 6 Thomas League 4 Thomas S. Young 13 Tim & Kathy Church 1 Tim Callahan 2 Tim Gidal 5 Tim Kopacz/Robert Hamil 3 Tim Porter 4 Timothy Greenfield-Sanders 19 Timothy Moyers 8 Timothy White 6 Tina Petra 1 Tito Simboli 13 Tiziana De Silvestro Hirschi 3 Toba Tucker 6 Toby Old 2 Todd Oldham 1 Todd Walker 5 Tohru Nakayama 1 Tom Ang 2 Tom Drahos 3 Tom Hammang Jr. 14 Tom Jones 2	Thomas E. Lindfors	5
Thomas Huther 12 Thomas League 4 Thomas S. Young 13 Tim & Kathy Church 1 Tim Callahan 2 Tim Gidal 5 Tim Kopacz/Robert Hamil 3 Tim Porter 4 Timothy Greenfield-Sanders 19 Timothy Moyers 8 Timothy White 6 Tina Petra 1 Tito Simboli 13 Tiziana De Silvestro Hirschi 3 Toba Tucker 6 Toby Old 2 Todd Oldham 1 Todd Walker 5 Tohru Nakayama 1 Tom Ang 2 Tom Drahos 3 Tom Hammang Jr. 14 Tom Jones 2	Thomas F. Barrow	6
Thomas League 4 Thomas S. Young 13 Tim & Kathy Church 1 Tim Callahan 2 Tim Gidal 5 Tim Kopacz/Robert Hamil 3 Tim Porter 4 Timothy Greenfield-Sanders 19 Timothy Moyers 8 Timothy White 6 Tina Petra 1 Tito Simboli 13 Tiziana De Silvestro Hirschi 3 Toba Tucker 6 Toby Old 2 Todd Oldham 1 Todd Walker 5 Tohru Nakayama 1 Tom Ang 2 Tom Drahos 3 Tom Hammang Jr. 14 Tom Jones 2	Thomas Harris	6
Thomas S. Young 13 Tim & Kathy Church 1 Tim Callahan 2 Tim Gidal 5 Tim Kopacz/Robert Hamil 3 Tim Porter 4 Timothy Greenfield-Sanders 19 Timothy Moyers 8 Timothy White 6 Tina Petra 1 Tito Simboli 13 Tiziana De Silvestro Hirschi 3 Toba Tucker 6 Toby Old 2 Todd Oldham 1 Todd Walker 5 Tohru Nakayama 1 Tom Ang 2 Tom Drahos 3 Tom Hammang Jr. 14 Tom Jones 2	Thomas Huther	12
Thomas S. Young 13 Tim & Kathy Church 1 Tim Callahan 2 Tim Gidal 5 Tim Kopacz/Robert Hamil 3 Tim Porter 4 Timothy Greenfield-Sanders 19 Timothy Moyers 8 Timothy White 6 Tina Petra 1 Tito Simboli 13 Tiziana De Silvestro Hirschi 3 Toba Tucker 6 Toby Old 2 Todd Oldham 1 Todd Walker 5 Tohru Nakayama 1 Tom Ang 2 Tom Drahos 3 Tom Hammang Jr. 14 Tom Jones 2	Thomas League	4
Tim Callahan 2 Tim Gidal 5 Tim Kopacz/Robert Hamil 3 Tim Porter 4 Timothy Greenfield-Sanders 19 Timothy Moyers 8 Timothy White 6 Tina Petra 1 Tito Simboli 13 Tiziana De Silvestro Hirschi 3 Toba Tucker 6 Toby Old 2 Todd Oldham 1 Todd Walker 5 Tohru Nakayama 1 Tom Ang 2 Tom Drahos 3 Tom Hammang Jr. 14 Tom Jones 2		13
Tim Callahan 2 Tim Gidal 5 Tim Kopacz/Robert Hamil 3 Tim Porter 4 Timothy Greenfield-Sanders 19 Timothy Moyers 8 Timothy White 6 Tina Petra 1 Tito Simboli 13 Tiziana De Silvestro Hirschi 3 Toba Tucker 6 Toby Old 2 Todd Oldham 1 Todd Walker 5 Tohru Nakayama 1 Tom Ang 2 Tom Drahos 3 Tom Hammang Jr. 14 Tom Jones 2	Tim & Kathy Church	1
Tim Kopacz/Robert Hamil 3 Tim Porter 4 Timothy Greenfield-Sanders 19 Timothy Moyers 8 Timothy White 6 Tina Petra 1 Tito Simboli 13 Tiziana De Silvestro Hirschi 3 Toba Tucker 6 Toby Old 2 Todd Oldham 1 Todd Walker 5 Tohru Nakayama 1 Tom Ang 2 Tom Drahos 3 Tom Hammang Jr. 14 Tom Jones 2		2
Tim Porter 4 Timothy Greenfield-Sanders 19 Timothy Moyers 8 Timothy White 6 Tina Petra 1 Tito Simboli 13 Tiziana De Silvestro Hirschi 3 Toba Tucker 6 Toby Old 2 Todd Oldham 1 Todd Walker 5 Tohru Nakayama 1 Tom Ang 2 Tom Drahos 3 Tom Hammang Jr. 14 Tom Jones 2	Tim Gidal	5
Tim Porter 4 Timothy Greenfield-Sanders 19 Timothy Moyers 8 Timothy White 6 Tina Petra 1 Tito Simboli 13 Tiziana De Silvestro Hirschi 3 Toba Tucker 6 Toby Old 2 Todd Oldham 1 Todd Walker 5 Tohru Nakayama 1 Tom Ang 2 Tom Drahos 3 Tom Hammang Jr. 14 Tom Jones 2	Tim Kopacz/Robert Hamil	3
Timothy Moyers 8 Timothy White 6 Tina Petra 1 Tito Simboli 13 Tiziana De Silvestro Hirschi 3 Toba Tucker 6 Toby Old 2 Todd Oldham 1 Todd Walker 5 Tohru Nakayama 1 Tom Ang 2 Tom Drahos 3 Tom Hammang Jr. 14 Tom Jones 2	_	4
Timothy White 6 Tina Petra 1 Tito Simboli 13 Tiziana De Silvestro Hirschi 3 Toba Tucker 6 Toby Old 2 Todd Oldham 1 Todd Walker 5 Tohru Nakayama 1 Tom Ang 2 Tom Drahos 3 Tom Hammang Jr. 14 Tom Jones 2	Timothy Greenfield-Sanders	19
Timothy White 6 Tina Petra 1 Tito Simboli 13 Tiziana De Silvestro Hirschi 3 Toba Tucker 6 Toby Old 2 Todd Oldham 1 Todd Walker 5 Tohru Nakayama 1 Tom Ang 2 Tom Drahos 3 Tom Hammang Jr. 14 Tom Jones 2	Timothy Moyers	8
Tito Simboli 13 Tiziana De Silvestro Hirschi 3 Toba Tucker 6 Toby Old 2 Todd Oldham 1 Todd Walker 5 Tohru Nakayama 1 Tom Ang 2 Tom Drahos 3 Tom Hammang Jr. 14 Tom Jones 2		6
Tiziana De Silvestro Hirschi 3 Toba Tucker 6 Toby Old 2 Todd Oldham 1 Todd Walker 5 Tohru Nakayama 1 Tom Ang 2 Tom Drahos 3 Tom Hammang Jr. 14 Tom Jones 2	_	1
Toba Tucker 6 Toby Old 2 Todd Oldham 1 Todd Walker 5 Tohru Nakayama 1 Tom Ang 2 Tom Drahos 3 Tom Hammang Jr. 14 Tom Jones 2	Tito Simboli	13
Toby Old 2 Todd Oldham 1 Todd Walker 5 Tohru Nakayama 1 Tom Ang 2 Tom Drahos 3 Tom Hammang Jr. 14 Tom Jones 2	Tiziana De Silvestro Hirschi	3
Todd Oldham 1 Todd Walker 5 Tohru Nakayama 1 Tom Ang 2 Tom Drahos 3 Tom Hammang Jr. 14 Tom Jones 2	Toba Tucker	6
Todd Oldham 1 Todd Walker 5 Tohru Nakayama 1 Tom Ang 2 Tom Drahos 3 Tom Hammang Jr. 14 Tom Jones 2	Toby Old	2
Todd Walker 5 Tohru Nakayama 1 Tom Ang 2 Tom Drahos 3 Tom Hammang Jr. 14 Tom Jones 2	-	
Tom Ang2Tom Drahos3Tom Hammang Jr.14Tom Jones2		5
Tom Ang2Tom Drahos3Tom Hammang Jr.14Tom Jones2	Tohru Nakayama	1
Tom Drahos3Tom Hammang Jr.14Tom Jones2	-	2
Tom Hammang Jr. 14 Tom Jones 2		3
Tom Jones 2		14
	_	2

Artist	Approximate Photo Quantity
Tom Millea	10
Tom Norton	36
Tom Schuler	1
Tomasz Pobog-Mahiuowski	2
Tomiyasu Shiraiwa	13
Ton Huijbers	17
Toni Catany	9
Tony Gardner	3
Toshio Shibata	4
Toto Frima	63
Tracy Storer	2
Tricia Grantz	2
Tuna Ciner	1
Tuneo Enari	22
Tyrone Georgiou	24
Tyson Joye	6
Ulla Diedrichsen	3
Ulrich Mack	159
Ulrich Tillmann	9
Umberto Saliti	2
Umberto Stefanelli	4
Umihiko Konishi	3
Unglee	4
Urs Jauch	1
Urs Obrist	7
Uwe Gobel	1
Uwe Laysiepen	27
Valene Adamson	1
Valerie Burton	7
Valerie Shaff	1
Valerie Sonnenthal	3
Vance Allen	2
Vanni Monelli	2
Vaughn Sills	34
Vecto Glasses	1
Vera Sprunt	1
Verena von Gagern	14
Vic Pinto	3
Vicente Carlos Fadon	2
Vicki Lee Ragan	42
Vicky Alexander/Ellen Brooks	8
Victor Landweber	5
Victor Raphael	6
Victor Schrager	11
Viera Andersen	2

Artist	Approximate Photo Quantity
Vincent Borrelli	3
Vincent Cianni	15
Vincent Vallarino	10
Volk Mol	1
Volker Liesfeld	5
Volker Schrader	1
Volker Steinhau	7
Waclaw Nowak	8
Walker Evans	8
Wallace Wilson	13
Walter Chandoha	19
Walter Chappell	16
Warren Krupsaw	48
Warren Neidich	4
Wayne Eastep	2
Wayne Gravning	5
Wayne Lambert	5
Wayne Lazorik	1
Wayne Miller	1
Wendy Ewald	139
Wendy Holmes	3
Wendy Hughes Brown	2
Wendy Jans	1
Werner J. Hannappel	15
Werner Pawlok	25
Werner Schnelle	11
Weston Andrews	28
Weston Kemp	67
Wiemann, Wolfgang Bruggemann	2
Wilko van Oostrum	6
Will McBride	7
Willard Traub	2
Willard Van Dyke	19
Willem Dassen	9
Willem De Boer	4
William Coupon	1
William Ropp	6
William Albert Allard	108
William Binsen Jr.	6
William Buckley	33
William Christenberry	12
William Clift	59
William Crawford	2
William Dennison	2
William Drescher	1
William Garnett	5
William Kauffmann	2

Artist	Approximate Photo
	Quantity
William Larson	6
William Mercer	26
William Murphy	2
William Neill	3
William Paris	11
William Saunders	1
William Short	5
William Wegman	78
Wingate Paine	13
Wolfgang Dietze	9
Wolfgang Keuchl	1
Wolfgang Ludes	2
Wolfgang Mackrod	1
Wolfram Scheler	17
Wout Berger	34
Wutian WU	1
Xavier Vailhen	1
Yasu Suzuka	12
Yasuhiro Ishimoto	5
Yehuda Yordan	3
Yoko Yamamoto	2
Yoshinori Tomiyama	2
Yousuf Karsh	136
Yuko Tamaka	1
Yullian Lionel	1
Zarina Bhimji	8
Zeke Berman	1
Zeva Oelbaum	3
Zsolt-Peter Barta	2
Total	15,934